% EJC papers *must* begin with the following two lines. 
\documentclass[12pt]{article}
\usepackage{e-jc}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm}
\usepackage{amsmath}
\usepackage{amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue,bookmarks=false]{hyperref}

\usepackage{tikz}
\usepackage[vcentermath]{youngtab}
\usepackage{tikz-cd}
\usetikzlibrary{arrows}
\usepackage{ytableau}
\usetikzlibrary{calc}
\usetikzlibrary{patterns}
\usetikzlibrary{decorations.pathmorphing}
\usetikzlibrary{decorations}
\usetikzlibrary{decorations.markings}
\usetikzlibrary{decorations.pathreplacing}
\usetikzlibrary{positioning}
\usetikzlibrary{automata}
\usepackage{framed}
\usepackage{filecontents}


% use these commands for typesetting doi and arXiv references in the bibliography
%\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
%\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins

\tikzset{->-/.style={decoration={
  markings,
  mark=at position #1 with {\arrow{>}}},postaction={decorate}}}


\usetikzlibrary{fit}
\makeatletter
\tikzset{
  fitting node/.style={
    inner sep=0pt,
    fill=none,
    draw=none,
    reset transform,
    fit={(\pgf@pathminx,\pgf@pathminy) (\pgf@pathmaxx,\pgf@pathmaxy)}
  },
  reset transform/.code={\pgftransformreset}
}


%% theorem environments
\theoremstyle{plain}
\newtheorem{thm}{Theorem}
\newtheorem{lemma}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{proposition}[thm]{Proposition}
\newtheorem{fact}[thm]{Fact}
\newtheorem{obs}[thm]{Observation}
\newtheorem{claim}[thm]{Claim}

\theoremstyle{definition}
\newtheorem{defn}[thm]{Definition}
\newtheorem{example}[thm]{Example}
\newtheorem{conjecture}[thm]{Conjecture}
\newtheorem{open}[thm]{Open Problem}
\newtheorem{problem}[thm]{Problem}
\newtheorem{question}[thm]{Question}

\theoremstyle{remark}
\newtheorem{remark}[thm]{Remark}
\newtheorem{note}[thm]{Note}

%% user-defined macros
\newcommand{\g}{{\mathfrak{g}}}
\newcommand{\qslr}{{U_q(\widehat{\mathfrak{sl}}_{r+1})}}
\newcommand{\qaa}{{U_q(\widehat{\mathfrak{g}})}}
\newcommand{\h}{{\mathfrak{h}}}
\newcommand{\gl}{{\mathfrak{gl}}}
\newcommand{\n}{{\mathfrak{n}}}
\newcommand{\KR}{KR}
\newcommand{\supp}{{\rm{Supp}}}
\newcommand{\length}{{\rm{length}}}
\newcommand{\head}{{\rm{head}}}
\newcommand{\tail}{{\rm{tail}}}
\newcommand{\Bd}{{\rm{Bd}}}

\newcommand\X{0.5}
\newcommand\D{2}
\newcommand*{\QEDA}{\hfill\ensuremath{\blacksquare}}%
\newcommand*{\QEDB}{\hfill\ensuremath{\square}}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf $(q,t)$-characters of Kirillov-Reshetikhin modules \\ of type $A_r$ as quantum cluster variables}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address 

\author{Bolor Turmunkh\thanks{Partially supported by the NSF grant DMS-1404988.}\\
\small Department of Mathematics\\[-0.8ex]
\small University of Illinois - Urbana-Champaign\\[-0.8ex] 
\small Illinois, U.S.A.\\
\small\tt bturmunk@illinois.edu}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Jul 30, 2017}{Dec 13, 2017}\\
\small Mathematics Subject Classifications:  05E10}

\begin{document}
\begin{thebibliography}{10}

\bibitem{bazhanov1990restricted} V. V. Bazhanov and Nikolai Reshetikhin.  \newblock {R}estricted solid-on-solid models connected with simply laced algebras and conformal field theory.  \newblock {\em Journal of Physics A: Mathematical and General}, 23(9):1477, 1990.

\bibitem{berenstein2005quantum} Arkady Berenstein and Andrei Zelevinsky. \newblock {Q}uantum cluster algebras. \newblock {\em Advances in Mathematics}, 195(2):405--455, 2005.

\bibitem{chari2001fermionic} Vyjayanthi Chari. \newblock {O}n the fermionic formula and the {K}irillov-{R}eshetikhin conjecture.  \newblock {\em International Mathematics Research Notices}, 2001(12):629--654, 2001.

\bibitem{chari1995quantum} Vyjayanthi Chari and Andrew Pressley. \newblock {Q}uantum affine algebras and their representations.  \newblock {\em Representations of groups (Banff, AB, 1994)}, 16:59--78, 1995.

\bibitem{di2009positivity} Philippe Di Francesco and Rinat Kedem. \newblock {P}ositivity of the {T}-system cluster algebra. \newblock {\em The Electronic Journal of Combinatorics}, 16(1):R140, 2009.

\bibitem{di2011non} Philippe Di Francesco and Rinat Kedem. \newblock {N}on-commutative integrability, paths and quasi-determinants. \newblock {\em Advances in Mathematics}, 228(1):97--152, 2011.

\bibitem{di2012solution} Philippe Di Francesco and Rinat Kedem. \newblock {T}he solution of the quantum {$A_1$} {T}-system for arbitrary boundary. \newblock {\em Communications in Mathematical Physics}, 313(2):329--350, 2012.

\bibitem{di2013quantum} Philippe Di Francesco and Rinat Kedem. \newblock {Q}uantum cluster algebras and fusion products. \newblock {\em International Mathematics Research Notices}, 2014(10):2593--2642, 2013.

\bibitem{di2015difference} Philippe Di Francesco and Rinat Kedem. \newblock {D}ifference equations for graded characters from quantum cluster algebra. \newblock {\em arXiv preprint arXiv:1505.01657}, 2015.

\bibitem{di2017quantum} Philippe Di Francesco and Rinat Kedem. \newblock {Q}uantum {Q}-systems: from cluster algebras to quantum current algebras. \newblock {\em Letters in Mathematical Physics}, 107(2):301--341, 2017.

\bibitem{feigin1999generalized} Boris Feigin and Serge Loktev. \newblock {O}n generalized {K}ostka polynomials and the quantum {V}erlinde rule. \newblock {\em Translations of the American Mathematical Society-Series 2}, 194:61--80, 1999.

\bibitem{fomin2002cluster} Sergey Fomin and Andrei Zelevinsky. \newblock {C}luster algebras {I}: foundations. \newblock {\em Journal of the American Mathematical Society}, 15(2):497--529, 2002.

\bibitem{frenkel2001combinatorics} Edward Frenkel and Evgeny Mukhin. \newblock {C}ombinatorics of $q$-characters of finite-dimensional representations of quantum affine algebras. \newblock {\em Communications in Mathematical Physics}, 216(1):23--57, 2001.

\bibitem{frenkel1999q} Edward Frenkel and Nicolai Reshetikhin. \newblock The $q$-characters of representations of quantum affine algebras and deformations of $W$-algebras. \newblock {\em Contemporary Mathematics}, 248:163--206, 1999.

\bibitem{gekhtman2003cluster} Michael Gekhtman, Michael Shapiro and Alek Vainshtein. \newblock {C}luster algebras and {P}oisson geometry. \newblock {\em Mosc. Math. J}, 3(3):899--934, 2003.


\bibitem{hatayama1998remarks} Goro Hatayama, Atsuo Kuniba, Masato Okado, Taichiro Takagi, and Yasuhiko Yamada. \newblock {R}emarks on fermionic formula. \newblock {\em arXiv preprint math/9812022}, 1998.

\bibitem{hernandez2004algebraic} David Hernandez. \newblock {A}lgebraic approach to $q,t$-characters. \newblock {\em Advances in Mathematics}, 187(1):1--52, 2004.

\bibitem{hernandez2013cluster} David Hernandez and Bernard Leclerc. \newblock {A} cluster algebra approach to $q$-characters of {K}irillov-{R}eshetikhin modules. \newblock {\em arXiv preprint arXiv:1303.0744}, 2013.

\bibitem{kedem2008q} Rinat Kedem. \newblock {Q}-systems as cluster algebras. \newblock {\em Journal of Physics A: Mathematical and Theoretical}, 41(19):194011, 2008.

\bibitem{kirillov1990representations} Anatol N. Kirillov and Nicolai Reshetikhin. \newblock {R}epresentations of {Y}angians and multiplicities of occurrence of the irreducible components of the tensor product of representations of simple {L}ie algebras. \newblock {\em Journal of mathematical sciences}, 52(3):3156--3164, 1990.

\bibitem{knight1995spectra} Harold Knight. \newblock {S}pectra of tensor products of finite dimensional representations of {Y}angians. \newblock {\em Journal of algebra}, 174(1):187--196, 1995.

\bibitem{kuniba1994functional} Atsuo Kuniba, Tomoki Nakanishi, and Junji Suzuki. \newblock {F}unctional relations in solvable lattice models {I}: functional relations and representation theory. \newblock {\em International Journal of Modern Physics A}, 9(30):5215--5266, 1994.


\bibitem{nakajima2000t} Hiraku Nakajima.  \newblock $t$-analogue of the $q$-characters of finite dimensional representations of quantum affine algebras.  \newblock {\em Physics and combinatoric}, pages 196--219, 2000.

\bibitem{nakajima2002t} Hiraku Nakajima.  \newblock $t$-analogs of $q$-characters of quantum affine algebras of type ${A}_n $, ${D}_n$.  \newblock {\em arXiv preprint math/0204184}, 2004.

\bibitem{nakajima2003t} Hiraku Nakajima.  \newblock $t$-analogs of $q$-characters of {K}irillov-{R}eshetikhin modules of quantum affine algebras.  \newblock {\em Represent. Theory}, 7(2):3, 2003.

\bibitem{nakajima2004quiver} Hiraku Nakajima.  \newblock {Q}uiver varieties and $t$-analogs of $q$-characters of quantum affine algebras.  \newblock {\em Annals of mathematics}, pages 1057--1097, 2004.


\end{thebibliography}


\maketitle

% E-JC papers must include an abstract. The abstract should consist of a
% succinct statement of background followed by a listing of the
% principal new results that are to be found in the paper. The abstract
% should be informative, clear, and as complete as possible. Phrases
% like "we investigate..." or "we study..." should be kept to a minimum
% in favor of "we prove that..."  or "we show that...".  Do not
% include equation numbers, unexpanded citations (such as "[23]"), or
% any other references to things in the paper that are not defined in
% the abstract. The abstract will be distributed without the rest of the
% paper so it must be entirely self-contained.

\begin{abstract}
  Nakajima (2003) introduced a $t$-deformation of $q$-characters, $(q,t)$-characters for short, 
  and their twisted multiplication through the geometry of quiver varieties. The Nakajima 
  $(q,t)$-characters of Kirillov-Reshetikhin modules satisfy a $t$-deformed $T$-system. 
  The $T$-system is a discrete dynamical system that can be interpreted as a mutation 
  relation in a cluster algebra in two different ways, depending on the choice of direction 
  of evolution. In this paper, we show that the Nakajima $t$-deformed $T$-system of type 
  $A_r$ forms a quantum mutation relation in a quantization of exactly one of the cluster 
  algebra structures attached to the $T$-system. 
  % keywords are optional
  
  \bigskip\noindent \textbf{Keywords:} $T$-system, quantum cluster algebra, $q$-characters
\end{abstract}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\section{Introduction}

Let $\mathfrak{g}$ be a simple Lie algebra over $\mathbb{C}$ of rank $r$, and let $U_q(\widehat{\mathfrak{g}})$ be the corresponding untwisted quantum affine algebra. Let $I$ be the set $\left\{1,2,\ldots, r\right\}$, and let $q\in \mathbb{C}^*$ be not a root of unity. The category of finite-dimensional complex $U_q(\widehat{\mathfrak{g}})$-modules has been classified by Chari and Pressley \cite{chari1995quantum}. Simple objects in this category are parametrized by an $r$-tuple of polynomials of one variable with constant term $1$, called the Drinfeld polynomials \cite{chari1995quantum}. 

Given $i\in I$, $k\in \mathbb{Z}_{\geq 0}$, $j\in \mathbb{Z}$, let 
\begin{equation} \label{KRDrinfeld}
\mathbf{P}_{k,j}^{(i)} = \left(  (P_{k,j}^{(i)})_\alpha (u) \right)_{\alpha\in I} \,\, ,\text{ where }
 (P_{k,j}^{(i)})_\alpha(u) =\left\{ \begin{array}{cc} \prod_{s=1}^k (1-q^j q^{2s-2}u) &\text{if } \alpha=i\\
 1&\text{otherwise}\end{array}\right.  
\end{equation}

A finite-dimensional irreducible module with an $r$-tuple of Drinfeld polynomials given by $\mathbf{P}_{k,j}^{(i)}$ is called a \emph{Kirillov-Reshetikhin module} (KR-module) and denoted $W_{k,j}^{(i)}$. KR-modules were introduced in \cite{kirillov1990representations}, and then further studied by Kuniba, Nakanishi, and Suzuki in \cite{kuniba1994functional}, Hatayama, Kuniba, Okado, Takagi, and Yamada in \cite{hatayama1998remarks} and Chari in \cite{chari2001fermionic}. One of the main tools used to study finite-dimensional $U_q(\widehat{\mathfrak{g}})$-modules is their \emph{$q$-characters}, which are the affine analogs of $U_q(\g)$-characters. The theory of $q$-characters was introduced by Knight \cite{knight1995spectra} and Frenkel-Reshetikhin \cite{frenkel1999q} for the Yangians and the quantum affine algebras respectively. One of the key properties concerning the KR-modules is that their characters and $q$-characters satisfy the functional relations called the $Q$-system and the $T$-system respectively. Nakajima proved the latter result in \cite{nakajima2003t} for types ADE using the $t$-analog of $q$-characters, $(q,t)$-characters for short, defined geometrically through quiver varieties. The same result was proved by Hernandez in \cite{hernandez2004algebraic} for all types using a different $t$-deformation of $q$-characters defined algebraicly. 



The $T$-system of type $A_r$ is a recursion relation on commuting variables $\left\{T_{k}^{(i)}(a)\right\}$, for $a\in \mathbb{C}^*, k\in \mathbb{Z}_+$, and $i\in I$, defined as follows:
\begin{eqnarray}\label{eq:classicTsystemSpec}
T_{k}^{(i)}\left( a\right) T_{k}^{(i)}\left(aq^2 \right) = T_{k+1}^{(i)}(a)T_{k-1}^{(i)}(aq^2) + T_{k}^{(i+1)}(a)T_{k}^{(i-1)}(a)\,,
\end{eqnarray}
with the convention $T_{k}^{(0)}(a) = T_{k}^{(r+1)}(a)=1$. Without loss of generality, we can always assume $a\in \mathbb{C}^*$ is fixed. Then we need only keep track of the powers of $q$. Using a change of variables $T_k^{(i)}(aq^j) \rightarrow T_{k, k+j}^{(i)}$ and relabeling $l = k+j+1$, we arrive at another form of the $T$-system, also known as the \emph{octahedron recurrence}:
\begin{eqnarray} \label{eq:classicTsystem}
T_{k,l-1}^{(i)} T_{k,l+1}^{(i)} = T_{k+1,l}^{(i)} T_{k-1,l}^{(i)} + T_{k,l}^{(i-1)} T_{k,l}^{(i+1)} ,
\end{eqnarray}
for  $k\in \mathbb{Z}_+$, $i=1,2,\ldots, r$, and with the convention $T_{k,l}^{(0)} = T_{k,l}^{(r+1)} = 1$. The $T$-system of type $A_r$ in \eqref{eq:classicTsystemSpec} was originally discovered by Bazhanov-Reshetikhin in \cite{bazhanov1990restricted} as a functional relations among the transfer matrices of the generalized RSOS models, and later, generalized by Kuniba, Nakanishi, and Suzuki in \cite{kuniba1994functional} to all Dynkin types. The Nakajima $(q,t)$-characters of KR-modules satisfy a deformed $T$-system with a twisted multiplication on the variables $T_{k,l}^{(i)}$. The $t$-deformed $T$-system is a quadratic recursion relation on non-commutative variables that reduce to the classical $T$-system when $t=1$. 


\emph{Cluster algebras} are commutative algebras generated by the union of commutative variables, called \emph{cluster variables}. The generators are related by rational transformations called \emph{mutations}, which are determined by an \emph{exchange matrix} \cite{fomin2002cluster}. Since all cluster variables are related to one another via mutations, it suffices to state a single cluster, called the \emph{fundamental cluster}, along with the exchange matrix, in order to define a cluster algebra. When the exchange matrix is invertible, there exists a canonical Poisson structure on the cluster variables \cite{gekhtman2003cluster}. A quantization of this canonical Poisson structure was introduced by Berenstein-Zelevinsky in \cite{berenstein2005quantum}, and is called a \emph{quantum cluster algebra}. Quantum cluster algebras are non-commutative algebras, and as such, their generators are not required to have any commutation relation at all. However, the variables within the same cluster satisfy a $t$-commutation relation. 


Kedem \cite{kedem2008q} and Di Francesco-Kedem \cite{di2009positivity} realized $Q$ and $T$ systems as mutation relations in certain cluster algebras. The $Q$-system is a recursion relation on commuting variables $\left\{Q_{k}^{(i)}\right\}$ obtained from the $T$-system in the $l\rightarrow \infty$ limit. The cluster algebra formulation of the $Q$-system in \cite{kedem2008q} was used to obtain a unique quantization of the $Q$-system in \cite{di2012solution,di2011non}. The resulting quantum $Q$-system was shown to have deep connections with many areas such as the fusion product, defined in \cite{feigin1999generalized}, of KR-modules \cite{di2013quantum}, the quantum current subalgebra $U_q(\n_+[u,u^{-1}])$ in $U_q(\widehat{\mathfrak{sl}}_2)$ \cite{di2017quantum} , where $\n_+$ is the positive nilpotent subalgebra in $\mathfrak{sl}_2$, and a new set of $q$-difference operators \cite{di2015difference}, which are generalizations of the Macdonald raising operators in the limit $t\rightarrow \infty$, acting on the characters of KR-modules of type $A_r$. 


The $T$-system equation can be interpreted as mutation relations in an infinite rank cluster algebra if we declare the $T_{k,l}^{(i)}$ variables to be invertible  \cite{hernandez2013cluster,di2009positivity}. The $T$-system in \eqref{eq:classicTsystem} is written such that the direction of mutation is in the direction of the $l$-parameter. That is, one can obtain $T_{k,l+1}^{(i)}$ using variables with lower value of $l$ only: 
\begin{eqnarray*}
T_{k,l+1}^{(i)} =  (T_{k,l-1}^{(i)})^{-1} \left(T_{k+1,l}^{(i)}T_{k-1,l}^{(i)} + T_{k,l}^{(i-1)}T_{k,l}^{(i+1)} \right).
\end{eqnarray*}
It is possible to rewrite the $T$-system in \eqref{eq:classicTsystem} such that the direction of mutation is in the $k$-parameter (see \cite{di2012solution}). Because the $T$-system cluster algebra is an infinite rank cluster algebra, these two choices of directions of mutations define $2$ distinct cluster algebra structures. Moreover, since the exchange matrices in both cases are of infinite rank, there are no canonical Poisson structures to these cluster algebras associated with the $T$-system.  Di Francesco and Kedem considered a quantization of the $A_1$ $T$-system in \cite{di2012solution}. Their choice corresponds to the quantization of the $T$-system cluster algebra with direction of mutation in the $k$-parameter. 

In light of the above results, it is natural to ask if the Nakajima deformed $T$-system forms a quantum mutation in a quantum cluster algebra, and if yes, then which one. In this paper, we provide answers to both questions. In particular, we show that the Nakajima deformed $T$-system \emph{does} form a quantum mutation, but \emph{only} when the direction of mutation is in the $l$-parameter. 


\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

	\tikzstyle{black} = [circle, minimum width=5pt, fill, inner sep=0pt]
	\tikzstyle{white} = [circle, minimum width=5pt,draw,inner sep=0pt]

\node at (-0.5,0)[right] {$k$};
\node at (0,-1) [above] {$0$};
\node at (1,-1) [above] {$1$};
\node at (2,-1) [above] {$2$};
\node at (3,-1) [above] {$3$};
\node at (4,-1) [above] {$4$};

\draw (-.5,0) -- (-.2,-.5);

\node at (-0.5,0) [left,below] {$i$};
\node at (0,-1) [left] {$1$};
\node at (0,-2) [left] {$2$};
\node at (0,-3) [left] {$3$};
\node at (0,-4) [left] {$r$};

\node[black] (10) at (0,-1) {};
\node[white] (11) at (1,-1) {};
\node[black] (12) at (2,-1) {};
\node[white] (13) at (3,-1) {};
\node[black] (14) at (4,-1) {};
\node at (5,-1) {$\cdots$};
\node at (5,-2) {$\cdots$};
\node at (5,-3) {$\cdots$};
\node at (5,-4) {$\cdots$};
\node[white] (20) at (0,-2) {};
\node[black] (21) at (1,-2) {};
\node[white] (22) at (2,-2) {};
\node[black] (23) at (3,-2) {};
\node[white] (24) at (4,-2) {};

\node[black] (30) at (0,-3) {};
\node[white] (31) at (1,-3) {};
\node[black] (32) at (2,-3) {};
\node[white] (33) at (3,-3) {};
\node[black] (34) at (4,-3) {};
\node at (0,-3.4) {$\vdots$};
\node at (1,-3.4) {$\vdots$};
\node at (2,-3.4) {$\vdots$};
\node at (3,-3.4) {$\vdots$};
\node at (4,-3.4) {$\vdots$};

\node[white] (r0) at (0,-4) {};
\node[black] (r1) at (1,-4) {};
\node[white] (r2) at (2,-4) {};
\node[black] (r3) at (3,-4) {};
\node[white] (r4) at (4,-4) {};


\path[->,thick]

(10) edge (11)
(11) edge (21)
(21) edge (20)
(20) edge (10)
(20) edge (30)
(30) edge (31)
(31) edge (21)

(12) edge (11)
(21) edge (22)
(22) edge (12)
(22) edge (32)
(32) edge (31)

(12) edge (13)
(13) edge (23)
(23) edge (22)
(33) edge (23)
(32) edge (33)

(14) edge (13)
(23) edge (24)
(24) edge (14)
(24) edge (34)
(34) edge (33)

(r1) edge (r0)
(r1) edge (r2)
(r3) edge (r2)
(r3) edge (r4)

;

\end{tikzpicture}
\end{center}
\caption{The quiver $\Gamma_B$}
\label{fig:Tsystem}
\end{figure}



We now summarize the main findings of this paper. We first define the cluster algebra formulation of the $T$-system in a similar way to that in \cite{di2009positivity}, with the difference being that the direction of evolution is in the $l$-direction. The parameters $i,k,l$ in \eqref{eq:classicTsystem} correspond to $\alpha, k, j$ parameters in \cite{di2009positivity} respectively. Let $B$ be the signed adjacency matrix of the quiver $\Gamma_B$ in Figure~\ref{fig:Tsystem}, which is the exchange matrix of the cluster algebra we are constructing. 

\begin{defn} \label{notation:mod}
For any $n\in \mathbb{Z}_+$, we define $(n)_2 := n\,{\rm{mod}}2$. 
\end{defn}
To each vertex $(k,i)$ in $\Gamma_B$, we associate the fundamental cluster variable $T_{k,(i+k+1)_2}^{(i)}$, which is given by the $q$-character of the corresponding KR-module. The fundamental cluster is given by the set:
\begin{equation} \label{eq:FundClustT}
\mathcal{C} := \left\{T_{k,(i+k+1)_2}^{(i)} \,\, \big| \,\, i\in I, \,\, k\in \mathbb{Z}_+\right\}\,\, ,
\end{equation}
with the boundary conditions $T_{k,l}^{(i)} = 1$ if either $k=0$, $i=0$, or $i=r+1$. With the exchange matrix $B$ and the fundamental cluster $\mathcal{C}$, mutation relations are given precisely by the $T$-system equations. 

We now state the main result of this paper, which answers the question stated above:
\begin{thm}\label{thm:main}
Let $(B, \mathcal{C})$ be as before, only now we substitute $T_{k,l}^{(i)}$ with their non\-com\-mut\-ative versions given by $(q,t)$-characters of KR-modules. 

(I). The fundamental cluster variables $t$-com\-mute with respect to the Nakajima's twis\-ted multiplication (this condition ensures that it is possible to carry out the quantum mutation). Moreover, the commutation matrix $\Lambda = (\Lambda_{i,k,l}^{i',k',l'}),$ defined as:
\begin{equation} \label{LambdaMatrix}
T_{k,l}^{(i)} * T_{k',l'}^{(i')} = t^{\Lambda_{i,k,l}^{i',k',l'} } T_{k',l'}^{(i')} * T_{k,l}^{(i)} \text{ for all } T_{k',l'}^{(i')} , T_{k,l}^{(i)} \in \mathcal{C},
\end{equation}
is an integer-valued matrix.

(II). $(B,\Lambda)$ forms a compatible pair, i.e. 
\begin{equation} \label{Cond2}
\Lambda B = D,
\end{equation} 
where $D$ is a diagonal matrix with positive entries. 

(III). The quantum mutation is given by %Proposition 4.9 in BZ. 
\begin{eqnarray*} \label{Cond3}
T_{k,l-1}^{(i)} *T_{k,l+1}^{(i)} &=&  t^{\frac{1}{2} \Lambda_{i,k,l-1}^{i,k-1,l} + \frac{1}{2}\Lambda_{i,k,l-1}^{i,k+1,l} - \frac{1}{2} \Lambda_{i,k-1,l}^{i,k+1,l} } \,\,T_{k-1,l}^{(i)}*T_{k+1,l}^{(i)} \\&+& t^{\frac{1}{2} \Lambda_{i,k,l-1}^{i-1,k,l} + \frac{1}{2}\Lambda_{i,k,l-1}^{i+1,k,l} - \frac{1}{2} \Lambda_{i-1,k,l}^{i+1,k,l} } \,\,T_{k,l}^{(i-1)}*T_{k,l}^{(i+1)}.
\end{eqnarray*}
In other words, $(B,\mathcal{C}, \Lambda)$ defines a quantum cluster algebra. 
\end{thm}

There are $2$ main parts to this work. The bulk of the work is a combinatorial construction that proves $(q,t)$-characters in the fundamental cluster $t$-commute with one another under Nakajima's twisted multiplication. We use a slightly modified version of the tableaux-sum notation for $q$-characters introduced in \cite{nakajima2002t} and define the notion of a block-tableau, which plays an integral role in the proof. Once $t$-commutativity is established, the second half of this paper is concerned with the commutation coefficients of the fundamental cluster variables. In particular, we show that the commutation coefficients are compatible with the cluster algebra exchange matrix and the mutation relations in the language of Berenstein-Zelevinsky \cite{berenstein2005quantum}.  

The structure of the paper is as follows. In Chapter 2, we give the necessary background material. In Chapter 3, we introduce a modification of the tableaux-sum notation for the $q$-characters of Kirillov-Reshetikhin modules that appeared in \cite{nakajima2002t} and prove Theorem \ref{thm:main}, condition I. In Chapter 4, we show  conditions II and III in Theorem \ref{thm:main} are satisfied. In Chapter 5, we show that Nakajima's $t$-deformed $T$-system is not a quantum cluster algebra when the direction of mutation is in $k$ parameter using a simple counter example. And we conclude with Chapter 6.   


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\textbf{Acknowlodgements:} I would like to thank Rinat Kedem, Philippe Di Francesco, and Maarten Bergvelt for their valuable discussion, guidance, and editing. This work was partially funded by the NSF grant DMS-1404988.  


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Definitions} \label{defs}

\subsection{$(q,t)$-characters and the deformed $T$-system}
For the rest of this paper, $\g$ will always refer to the simple Lie algebra of type A and rank $r$, i.e. $\mathfrak{sl}_{r+1}$. Let $\left\{\alpha_i\right\}_{i \in I}$ and $\left\{ \omega_i\right\}_{i\in I}$ be the simple roots and the fundamental weights of $\g$. We denote $Q_+=\bigoplus_{i=1}^r \mathbb{Z}_{+} \alpha_i$ to be the positive root lattice and $P=\bigoplus_{i=1}^r \mathbb{Z} \omega_i$ to be the weight lattice. Let $\mathbf{R}$ and $\widehat{\mathbf{R}}$ be the Grothendieck rings of the categories of finite dimensional representations of $U_q(\g)$ and $U_q(\widehat{\g})$ respectively. %We will not distinguish between a module $V$ and its class in $\mathbf{R}$ or $\widehat{\mathbf{R}}$.

Given $\lambda\in P$ and the irreducible highest weight module $V_\lambda = \oplus_{\alpha \in Q_+} V_{\lambda - \alpha}$ in $\mathbf{R}$, the character of $V_\lambda$ is a formal sum $\chi(V_\lambda) = \sum_{\alpha \in Q_+} \dim(V_{\lambda - \alpha}) e^{\lambda - \alpha}$. If we write $y_i = e^{\omega_i}$, the character is an injective homomorphism of commutative rings:
\begin{eqnarray*}
\chi :\mathbf{R} \rightarrow \mathbb{Z}[y_i, y_i^{-1}]_{i\in I}.
\end{eqnarray*}
Let us also define monomials:
\begin{equation} \label{eq:finiteA}
a_i = y_i^2 y_{i-1}^{-1} y_{i+1}^{-1},
\end{equation}
with the convention $y_i = 1$ if $i\notin I$. Notice that $a_i$ is identified with $e^{\alpha_i}$. %Therefore, multiplying by $a_i^{-1}$ is equivalent to applying a lowering operator with weight $-\alpha_i$. 

The affine analog of the character map $\chi$ is the $q$-character map $\chi_q$, first introduced in \cite{knight1995spectra, frenkel1999q}, and it uniquely characterizes the isomorphism classes of irreducible modules in $\widehat{\mathbf{R}}$. The character of a finite dimensional $U_q(\g)$-module is the generating series of the weight space multiplicities. Similarly, the $q$-character of a finite dimensional $U_q(\widehat{\g})$-module is the generating series of the affine analog of the weight space multiplicities, called the $l$-weight spaces. The $q$-character map \cite{frenkel1999q} is an injective ring homomorphism $\chi_q$ that makes the following diagram commute:
\[
\begin{tikzcd}
\widehat{\mathbf{R}} \arrow{r}{\chi_q} \arrow[swap]{d}{\rm{res}} & \mathbb{Z}[Y_{i,j}^{\pm 1}]_{i\in I, j\in \mathbb{Z}} \arrow{d}{p} \\
\mathbf{R}  \arrow{r}{\chi} & \mathbb{Z}[y_i^{\pm 1}]_{i\in I}
\end{tikzcd}
\]
where $p(Y_{i,j})=y_i$ and $\rm{res}$ is the restriction map.  

Recall that the Drinfeld polynomial corresponding to $V\in \widehat{\mathbf{R}}$ is an $r$-tuple of polynomials of one variable. We identify every term of the form $(1-q^ju)$ appearing in the $i$th polynomial with $Y_{i,j}$. With this identification, the Drinfeld polynomial is mapped to a monomial of positive powers of $Y_{i,j}$'s only, i.e. a \emph{dominant monomial}, which corresponds to the $U_q(\g)$-highest weight vector in $V$. The Drinfeld polynomial corresponding to the KR-module $W_{k,j}^{(i)}$ defined in \eqref{KRDrinfeld} is identified with the dominant monomial $\mathbf{Y}_{k,j}^{(i)}$ as follows:
\begin{equation} \label{KRdom}
\mathbf{P}_{k,j}^{(i)} \mapsto \mathbf{Y}_{k,j}^{(i)} := Y_{i,j}Y_{i,j+2} \cdots Y_{i,j+2(k-2)}.
\end{equation}

Nakajima \cite{nakajima2000t,nakajima2002t,nakajima2004quiver} introduced a $t$-analog of the Grothendieck ring  $\widehat{\mathbf{R}}_t = \mathbb{Z}[t,t^{-1}]\otimes_{\mathbb{Z}} \widehat{\mathbf{R}}$ and the $t$-analog of $q$-characters through the geometry of Quiver varieties. The $(q,t)$-map is a $\mathbb{Z}[t,t^{-1}]$-linear injective map:
\begin{eqnarray*}
\chi_{q,t}: \widehat{\mathbf{R}}_t \rightarrow \mathbb{Z}[t,t^{-1}]\otimes \mathbb{Z}[Y_{i,j},Y_{i,j}^{-1}]_{i\in I, j\in \mathbb{Z}},
\end{eqnarray*}
with the property that $\chi_{q,t=1}=\chi_q$. Frenkel and Mukhin \cite{frenkel2001combinatorics} gave a combinatorial algorithm that computes the $q$-characters of a class of modules that includes the fundamental modules \cite{frenkel2001combinatorics} and the KR-modules \cite{nakajima2002t}. Nakajima gave a similar algorithm that computes the $(q,t)$-characters of the fundamental modules. Although $\chi_{q,t}$ is not a ring homomorphism, Nakajima introduced a twisted multiplication on both the source and the target of $\chi_{q,t}$ that makes it into a homomorphism of non-commutative rings. Thus, starting with the $(q,t)$-characters of the fundamental modules, one can obtain the $(q,t)$-character of any twisted product of fundamental modules. %. standard module, which is a tensor product of fundamental modules \cite{varagnolo2002standard}. 

The $(q,t)$-characters of KR-modules of type $A$ are identical to their $q$-characters \cite{nakajima2000t}. Therefore, we will omit the description of the algorithm for obtaining the $(q,t)$-characters of the fundamental modules and move straight to the description of the twisted multiplication. 

Let
\begin{eqnarray}\label{def:A}
A_{i,j} = Y_{i,j-1}Y_{i,j+1} Y_{i-1,j}^{-1}Y_{i+1,j}^{-1} \,\, ,
\end{eqnarray} 
with the convention that if $i \notin I$, then $Y_{i,j}=1$. This is the affine analog of the monomials in \eqref{eq:finiteA}. Given $V\in \widehat{\mathbf{R}}$, the $q$-character of $V$ \cite{frenkel2001combinatorics}, and in our case, the $(q,t)$-character of $V$ if $V$ is a KR-module, is of the form:
\begin{eqnarray*}
\chi_q(V) = m_+ + \sum_{M\in \mathcal{A}} m_+ M \,\, ,
\end{eqnarray*}
where $m_+$ is a dominant monomial and $\mathcal{A}$ is some subset of monomials in $\mathbb{Z}_+[A_{i,j}^{-1}]_{i\in I j\in \mathbb{Z}}$. 

\begin{defn} 
Let $\mathcal{M}$ be the set of monomials in $\mathbb{Z}[Y_{i,j}, Y_{i,j}^{-1}]_{i\in I, j\in \mathbb{Z}}$. Given $m,m'\in \mathcal{M}$, we say $m'$ is a descendant of $m$ if $m'$ is obtained by applying a set of $A_{i,j}^{-1}$ to $m$, i.e. $m'm^{-1}\in \mathbb{Z}[A_{i,j}^{-1}]_{i\in I, j\in \mathbb{Z}}$. 
\end{defn}
With this definition, given a $q$-character of an irreducible module, all its monomials are descendants of the dominant monomial. 
\begin{defn} \label{uv}
For $m_+,m\in \mathcal{M}$ such that $m_+$ dominant and $m$ descendant of $m_+$, define $u_{i,j}(m) \in \mathbb{Z}$ and $v_{i,j}(m,m_+)\in \mathbb{N}$ as follows: 
\begin{equation*}
m = \prod_{i,j} Y_{i,j}^{u_{i,j}(m)} =m_+ \prod_{i,j} A_{i,j}^{-v_{i,j}(m,m_+)}\,\, .
\end{equation*}
\end{defn}

The following definitions are due to Nakajima \cite{nakajima2003t}.

\begin{defn} \label{uinverse}
For $m\in \mathcal{M}$, let $\tilde{u}_{i,j}(m) \in \mathbb{R}$ ($i\in I, j\in \mathbb{Z}$) be a solution of the system
\begin{equation*} 
u_{i,j}(m) = \tilde{u}_{i,j-1}(m) + \tilde{u}_{i,j+1}(m) - \tilde{u}_{i-1,j}(m) - \tilde{u}_{i+1,j}(m) \,\, ,
\end{equation*}
such that $\tilde{u}_{i,j}(m)=0$ for $j$ sufficiently small. As usual, $\tilde{u}_{i,j}(m) = 0$ if $i\notin I$. 
\end{defn}
\begin{remark}
For any monomial $m\in \mathcal{M}$, it can be seen from Definition \ref{uv}, there are only finitely many non-zero $u_{i,j}(m)$. The condition $\tilde{u}_{i,j}(m)=0$ for $j$ sufficiently small ensures there is in fact a unique integral solution to the system, which can be verified through direct computation. However, there can be infinitely many non-zero $\tilde{u}_{i,j}(m)$. The system in Definition \ref{uinverse} looks cryptic. As we will never have to solve this system explicitly, the reader need only realize that there is a unique solution to the system at this point.  
\end{remark}

\begin{defn} \label{def:Nakajima}
Let $m_+^1,m_+^2 \in \mathcal{M}$ be dominant monomials and $m^1,m^2\in \mathcal{M}$ such that $m^i$ is a descendant of $m_+^i$. 
\begin{eqnarray*}
\epsilon(m_+^1, m_+^2) &:=& - \sum_{i,j} u_{i,j+1}(m_+^1) \tilde{u}_{i,j}(m_+^2) + \sum_{i,j} u_{i,j+1}(m_+^2) \tilde{u}_{i,j}(m_+^1) \,\, ,\\
d(m^1,m_+^1; m^2, m_+^2) &:=& \sum_{i,j} v_{i,j+1}(m^1,m_+^1)u_{i,j}(m^2) + u_{i,j+1}(m_+^1) v_{i,j}(m^2, m_+^2) \,\, ,\\
\gamma(m^1, m_+^1; m^2, m_+^2) &:=& d(m^1,m_+^1; m^2, m_+^2) - d(m^2,m_+^2; m^1, m_+^1)\,\, .
\end{eqnarray*}
\end{defn}
Notice that the above sums are all finite sums as there are only finitely many non-zero values of $u_{i,j}(m)$ and $v_{i,j}(m,m_+)$ for any monomials $m$ and $m_+$. Also, it follows immediately from the definition that both $\gamma$ and $\epsilon$ are anti-symmetric. 

\begin{remark} \label{rmk:simplegamma}
Given a monomial $m$ in a $(q,t)$-character, in our application of the $\gamma$ function, it will always be clear what the dominant monomial of $m$ is. Therefore, we simplify our notation and write
\begin{equation*}
\gamma(m^1,m^2) := \gamma(m^1,m_+^1; m^2, m^2_+)\,\,.
\end{equation*}
\end{remark}

\begin{defn} \label{twistedMult}
Let $V^1, V^2 \in \widehat{\mathbf{R}}$ and let $\chi_{q,t}^1$ and $\chi_{q,t}^2$ be their $(q,t)$-characters. Let $m_+^1, m_+^2$ be the dominant monomials and $m^1, m^2$ any monomials in $\chi_{q,t}^1$ and $\chi_{q,t}^2$ respectively. The twisted multiplication on monomials is defined as follows:
\begin{equation*} 
m^1 * m^2 := t^{\gamma(m^1,m^2) + \epsilon(m_+^1,m_+^2)} m^1m^2 \,\, ,
\end{equation*}
where $m^1m^2$ is the usual multiplication of monomials. Multiplication on $(q,t)$-characters is defined by linearly expanding the twisted multiplication on monomials. 
\end{defn}
We also define $*_\gamma$ multiplication as follows: 
\begin{equation} \label{gammaMult}
m^1*_\gamma m^2 := t^{-\epsilon(m_+^1, m_+^2)} m^1*m^2 = t^{\gamma(m^1,m^2)} m^1m^2 \,\, .
\end{equation}
We now state the result that this work is based on. 
\begin{thm}[Nakajima, \cite{nakajima2003t}] \label{thm:Nakajima}
Let $\chi_{k,j}^{(i)} := \chi_{q,t}(W_{k,j}^{(i)})$, where $W_{k,j}^{(i)}$ is the KR-module with dominant monomial $\mathbf{Y}_{k,j}^{(i)}$ (see \eqref{KRdom}). The following relations hold between the $(q,t)$-characters:
\begin{equation} \label{eq:tTsystem}
\chi_{k,j}^{(i)} *_\gamma \chi_{k,j+2}^{(i)} =\chi_{k+1,j}^{(i)} *_\gamma \chi_{k-1,j+2}^{(i)} + t^{-1} \chi_{k,j+1}^{(i-1)} *_\gamma \chi_{k,j+1}^{(i+1)} \,\, ,
\end{equation}
with the convention $\chi_{k,j}^{(i)} = 1$ if $k=0$ or $i\notin I$. 
\end{thm}
\begin{remark} \label{rmk:changeOfVar}
Equation \eqref{eq:tTsystem} can be transformed into a deformation of \eqref{eq:classicTsystem} by a change of variables $\chi_{k,j}^{(i)} \rightarrow T_{k, k+j}^{(i)}$ and relabeling $l = k+j+1$ with no changes to the coefficients. 
\end{remark}
The goal is to show that the deformed $T$-system in \eqref{eq:tTsystem} and the $(q,t)$-characters of KR-modules form a quantum cluster algebra. The exchange matrix is the adjacency matrix $B$ of the quiver $\Gamma_B$ in Figure~\ref{fig:Tsystem}. The fundamental cluster variables, given in \eqref{eq:FundClustT}, can be written in terms of $\chi_{k,j}^{(i)}$ variables as follows:
\begin{equation} \label{eq:FundClust}
\mathcal{C} = \left\{ \chi_{k,-k+(i+k+1)_2}^{(i)} \,\, \big| \,\,  i\in I, \,\, k \in \mathbb{Z}_+ \right\} \,.
\end{equation}
Notice that the fundamental cluster \eqref{eq:FundClust} is obtained from \eqref{eq:FundClustT} by the change of variables described in Remark \ref{rmk:changeOfVar}. 

\begin{remark}
Let $\chi_1$ and $\chi_2$ be $(q,t)$-characters with dominant monomials $m_+^1$ and $m_+^2$ respectively. Suppose $\chi_1$ and $\chi_2$ $t$-commute with respect to $*$. That is, there exists some $\alpha\in \mathbb{Z}$ such that
\begin{eqnarray}\label{CondIEquiv}
\chi_1*\chi_2 = t^{\alpha} \chi_2*\chi_1 \,\, ,
\end{eqnarray}
which is equivalent to
\begin{eqnarray} 
\chi_1*_\gamma \chi_2 = t^{\alpha - 2\epsilon(m_+^1,m_+^2)} \chi_2*_\gamma \chi_1\,\, .
\end{eqnarray}
Notice that $\gamma(m_+^1,m_+^2,) =0$ since $v_{i,j}(m_+,m_+)=0$ for all $i,j$ and any dominant monomial $m_+$.  That is, the dominant monomials in $ \chi_1 *_\gamma \chi_2$ and $\chi_2 *_\gamma \chi_1$ have coefficient $1$. Therefore, \eqref{CondIEquiv} holds if and only if $\alpha= 2\epsilon(m_+^1, m_+^2)$, and  $\chi_1 *_\gamma \chi_2 = \chi_2 *_\gamma \chi_1$. 
\end{remark}
  
\begin{cor}\label{gammaComm}
The fundamental cluster variables $t$-commute with respect to $*$, i.e. Condition I holds, if and only if they commute with respect to $*_\gamma$. Moreover, if the fundamental cluster variables $t$-commute, then the commutation matrix $\Lambda$, defined in \eqref{LambdaMatrix}, is given by 
\begin{eqnarray*}
\Lambda_{i,j,k}^{i',j',k'} = 2\epsilon( \mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k',j'}^{(i')} ) \,\, ,
\end{eqnarray*}
where $\mathbf{Y}_{k,j}^{(i)}$ is the dominant monomial of $\chi_{q,t}(W_{k,j}^{(i)})$ defined in \eqref{KRdom}. 
\end{cor}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Proof of Condition I}
The main tool we use in this section is the tableaux-sum expression for the $q$-characters of KR-modules, introduced by Nakajima in \cite{nakajima2002t}, and the mapping $m:T\mapsto m_T$ from the space of allowed tableaux, called KR-tableaux in the text and to be defined in Section \ref{tableaux-sum}, to the space of monomials $\mathcal{M}$. Let $\mathcal{B}_{k,j}^{(i)}$ be the set of all KR-tableaux parametrizing the monomials of $\chi_{q}(W_{k,j}^{(i)})$.  

The idea of the proof of Condition I is as follows. We define a division of the set $\mathcal{B}_{k,j}^{(i)} \times\mathcal{B}_{k',j'}^{(i')}$ into $3$ disjoint subsets $\mathcal{P}_0, \mathcal{P}_1, \mathcal{P}_{-1}$ and an automorphism $\sigma$ on this set such that 
\begin{itemize}
\item $\sigma$ fixes the elements of $\mathcal{P}_0$ and is an involution between $\mathcal{P}_1$ and $\mathcal{P}_{-1}$
\item $m_{(C,T)} = m_{\sigma(C,T)}$, where $m_{(C,T)} := m_Cm_T$. 
\item $\gamma(m_C,m_T) =: \gamma(C,T) = -\gamma(\sigma(C,T))$.  
\end{itemize}
The main result of this section is:
\begin{thm} \label{thm:main1}
Let $(C,T)\in \mathcal{P}_0$, i.e. $(C,T)$ is a fixed point of $\sigma$. Then $\gamma(C,T)=0$. 
\end{thm}

The proof of Theorem \ref{thm:main1} is given in Section \ref{proofOfMain1}. Let us show how Theorem \ref{thm:main1} implies Condition I. 

\begin{cor}
The fundamental cluster variables $t$-commute with respect to the twisted multiplication $*$ defined in \eqref{twistedMult}. 
\end{cor}
\begin{proof}
By Corollary \ref{gammaComm}, Condition I is equivalent to the statement that elements in $\mathcal{C}$ (see \eqref{eq:FundClust}) commute with respect to $*_\gamma$. That is, given any $\chi_{k,j}^{(i)},\chi_{k',j'}^{(i')}\in \mathcal{C}$, we want to show
\begin{equation} \label{Lambda}
\chi_{k,j}^{(i)}* _\gamma \chi_{k',j'}^{(i')} = \chi_{k',j'}^{(i')} *_\gamma \chi_{k,j}^{(i)}
\end{equation}
Let's write
\begin{eqnarray*}
\chi_{k,j}^{(i)} = \sum_{C\in \mathcal{B}_{k,j}^{(i)}} m_C \text{ and }\chi_{k',j'}^{(i')} = \sum_{T\in \mathcal{B}_{k',j'}^{(i')}} m_T \,\, ,
\end{eqnarray*}
where $\mathcal{B}_{k,j}^{(i)}$ and $\mathcal{B}_{k',j'}^{(i')}$ are the sets of allowed KR-tableaux. Denote $\mathcal{B} = \mathcal{B}_{k,j}^{(i)} \times \mathcal{B}_{k',j'}^{(i')}$. Then,
\begin{eqnarray*}
\chi_{k,j}^{(i)} *_\gamma\chi_{k',j'}^{(i')} &=& \sum_{(C,T) \in \mathcal{B}} t^{\gamma(C,T)} m_Cm_T \\
&=& \underbrace{     \sum_{(C,T) \in \mathcal{P}_0\cap \mathcal{B}}  t^{\gamma(C,T)} m_{(C,T)} }_\text{$\gamma(C,T)=0$ by Theorem \ref{thm:main1}} + \sum_{(C,T) \in \mathcal{P}_1 \cap \mathcal{B}}  \left(t^{\gamma(C,T)} m_{(C,T)} + t^{\gamma(\sigma(C,T))} m_{\sigma(C,T)}\right)  \\
&=& \sum_{(C,T)\in \mathcal{P}_0 \cap \mathcal{B} } m_{T}m_{C} +  \sum_{ (C,T)\in \mathcal{P}_1 \cap \mathcal{B}} \left( t^{-\gamma(T,C)}  m_{T}m_{C} + t^{ \gamma(T,C)} m_{T}m_{C} \right)\\
&=&\chi_{k',j'}^{(i')}*_\gamma \chi_{k,j}^{(i)}
\end{eqnarray*}
where we used $\gamma(\sigma(C,T)) = -\gamma(C,T) = \gamma(T,C)$ and $m_{(C,T)} = m_Cm_T = m_{(T,C)}$. 
\end{proof}

\subsection{Tableaux-sum notation} \label{tableaux-sum}
The tableaux-sum notation we are using is a slight adjustment of the notation introduced in \cite{nakajima2002t}. Let us begin with the tableaux-sum description of the ordinary character $\chi(V)$ for $V\in \mathbf{R}$ as a motivation to the later definitions of the tableaux-sum notation for $q$-characters.

Let $V_\lambda$ be the highest weight irreducible $U_q(\mathfrak{sl}_{r+1})$-module with highest weight $\lambda = \sum_{i=1}^r \lambda_i \omega_i$. Then there exists a basis of $V_\lambda$ parametrized by semi-standard Young tableaux of shape $\Lambda = (\Lambda_1, \ldots, \Lambda_r)$, where $\Lambda_j = \sum_{i=j}^r \lambda_i$. Let $S(\Lambda)$ be the set of all semi-standard Young tableaux on the letters $\left\{1,2, \ldots, r+1\right\}$ of shape $\Lambda$. We define a map
\begin{eqnarray} \label{map:mon}
\begin{array}{cccc}m: &S(\Lambda) &\rightarrow & \multicolumn{1}{l}{\mathbb{Z}[y_i^{\pm 1}]_{i\in I}}\\
&&&\\
&T&\mapsto& m_T = \prod_{i\in I} y_i^{ \#T(i) - \# T(i+1)} \end{array}
\end{eqnarray}
where $\#T(i)$ is the number of times $i$ appears in $T$. With this mapping, we obtain the tableaux-sum expression for the character of $V_\lambda$. More precisely, 
\begin{eqnarray*}
\chi(V_\lambda) = \sum_{T \in S(\Lambda)} m_T\,\, .
\end{eqnarray*}


\begin{remark} \label{rectangularT}
The $U_q(\g)$-highest weight of the KR-module $W_{k,j}^{(i)}$ is given by $k\omega_i$. Therefore, the tableaux that parametrize the character of KR-modules, considered as $U_q(\g)$-modules, are rectangular of length $i$ and width $k$.  
\end{remark}

\begin{example}\label{ex:finite}
Consider $\g = \mathfrak{sl}_4$ and $V=V_{\omega_3}$. Then, $\Lambda = (1,1,1)$. 
\ytableausetup{mathmode,boxsize=1em}

\vspace{0.1in}
\begin{center}
\begin{tikzpicture}

\node at (-2,2) {$\chi(V_{\omega_3})$};
\node at (-1,2) {$=$};

\node (1') at (0,2) {$y_3$};

\node (1) at (0,0) {
\begin{ytableau}
1\\
2\\
3
\end{ytableau}
};

\node (2') at (2,2) { $y_2y_3^{-1}$};

\node (2) at (2,0) {
\begin{ytableau}
1\\
2\\
4
\end{ytableau}
};


\node (3') at (4,2) { $y_1y_2^{-1}$};

\node (3) at (4,0) {
\begin{ytableau}
1\\
3\\
4
\end{ytableau}
};


\node (4') at (6,2) { $y_1^{-1}$};

\node (4) at (6,0) {
\begin{ytableau}
2\\
3\\
4
\end{ytableau}
};

\node at (1,2) {$+$};
\node at (3,2) {$+$};
\node at (5,2) {$+$};

\path[->,thick]

(1) edge node[above] {$a_3^{-1}$}  (2)
(2) edge node[above] {$a_2^{-1}$}  (3)
(3) edge node[above] {$a_1^{-1}$} (4)
(1') edge[dotted] (1)
(2') edge[dotted] (2)
(3') edge[dotted] (3)
(4') edge[dotted] (4)
;

\end{tikzpicture}
\end{center}
\vspace{0.1in}
Recall that the monomials $a_i$ in \eqref{eq:finiteA} are identified with $e^{\alpha_i}$. Therefore, multiplying a monomial by $a_i^{-1}$ is equivalent to applying a lowering operator with weight $-\alpha_i$. The action of $a_i^{-1}$ on the tableaux is given by changing a box with $i$ to $i+1$. \qed
\end{example}

We now describe the tableaux-sum notation for $q$-characters. Recall that the restriction of $U_q(\widehat{\g})$-modules to $U_q(\g)$ corresponds to the map $Y_{i,j} \rightarrow y_i$ on the monomials, and the finite weight of $Y_{i,j}$ and $y_i$ are the same, given by $\omega_i$. Therefore, the tableau representation of $Y_{i,j}$ is almost identical to that of $y_i$, except there are infinitely many $Y_{i,j}$'s corresponding to $j\in \mathbb{Z}$. This infinite property is represented by an extra vertical coordinate added to the usual tableau data. 

\begin{defn}
\begin{enumerate}
\item Let $T$ be a diagram consisting of a single column of length $i$ equip\-ped with an additional datum $j\in \mathbb{Z}$. To each box in $T$, we associate an \emph{index} as follows: the top box gets an index $\frac{1-i -j}{2}$ and the indices of the lower boxes increase by one starting from the top box. We call the set of indices of $T$ the \emph{support} of $T$, denoted $\rm{Supp}(T)$. Then,
\begin{eqnarray*}
\rm{Supp}(T) = \left\{ \frac{1-i-j}{2}, \ldots, \frac{1 + i - j}{2} \right\}
\end{eqnarray*}
The diagrams that we will consider in this paper will have $i$ and $j$ such that $(1-i-j)$ is always divisible by $2$. That is, $\supp(T)\subset \mathbb{Z}$. We say $T$ is a \emph{column diagram} of shape $(i,j)$.  
\item The \emph{head} of a column diagram $T$, denoted $\head(T)$, is the index of the first box.
\item The \emph{tail} of a column diagram $T$, denoted $\tail(T)$, is the index of the last box.
\item The \emph{length} of a column diagram $T$ is given by $i=\tail(T)-\head(T)+1$. For convenience, we denote it $\length(T)$. 
\item A \emph{column tableau} $T$ is a column diagram $T$ of some shape $(i,j)$ decorated with letters $\left\{1,2,\ldots, r+1\right\}$, i.e. we equip $T$ with an arbitrary map
 \begin{equation*}
 \supp(T) \rightarrow \left\{1,2,\ldots, r+1\right\} \,,
 \end{equation*}
 where $r$ is the rank of the Lie algebra $\g$. Equivalently, a column tableau is a column diagram with integers between $1$ and $r+1$ filled in each box.
 
The image of the map at $p\in \supp(T)$, i.e. the integer in the box with index $p$, is denoted $T[p]$ and is called the \textsl{value} of $T$ at $p$. If $p<\head(T)$, we set $T[p]=0$ and if $p>\tail(T)$, we set $T[p]=\infty$. With this redefinition, we can consider $T$ to be defined for all $p\in \mathbb{Z}$ and $\supp(T)$ is where the value of $T$ is nonzero and finite. 
\item A \emph{strip} in $T$ between $p_0$ and $p_1$, denoted $T[p_0,p_1]$, is the tableau given by the piece of $T$ between indices $p_0$ and $p_1$ with end points included. 
\item A \emph{general tableau} is obtained by stacking column tableaux horizontally with the indices of the boxes, defined in (1), aligned. 
\end{enumerate}
\end{defn}

Let $T=(T_1,\ldots, T_k)$ be a general tableau. Let $\supp(T) = \cup \supp(T_l)$. We identify $T$ with a monomial $m_T\in \mathcal{M}$ given by 
\begin{eqnarray} \label{map:qmon}
m_T= \prod_{p\in \rm{Supp}(T)} \prod_{i=1}^r Y_{i,i-2p-1}^{\#(T[p]=i) - \#(T[p+1]=i+1)} \,, 
\end{eqnarray}
where $\#(T[p]=i)$ is the number of times $i$ appears in $T$ at index $p$. This is the affine analog of the map \eqref{map:mon}. 
\begin{remark}
If we drop the indices of the tableaux and collapse all the columns until the heads of every column are on the same level, then we obtain Young tableaux, which gives the classical tableaux-sum notation for the highest weight modules. In particular, if we collapse the general tableau, then the map \eqref{map:qmon} reduces to the map \eqref{map:mon}. Diagramaticaly, we can add a third column to the commutative diagram 
\[
\begin{tikzcd}
\widehat{\mathbf{R}} \arrow{r}{\chi_q} \arrow[swap]{d}{\rm{res}} & \mathbb{Z}[Y_{i,j}^{\pm 1}]_{i\in I, j\in \mathbb{Z}} \arrow{d}{p}  &\left\{\text{General Tableaux}\right\} \arrow{l}[swap]{\eqref{map:qmon}} \arrow{d}{collapse}\\
\mathbf{R}  \arrow{r}{\chi} & \mathbb{Z}[y_i^{\pm 1}]_{i\in I}& \left\{\text{Tableaux}\right\} \arrow{l}[swap]{\eqref{map:mon}}
\end{tikzcd}
\]

\end{remark}

\begin{defn}
Let $T$ be a column tableau of shape $(i,j)$. We define $T_{dom}$ (dominant) to be the column tableau of the same underlying diagram given by
\begin{eqnarray*}
T_{dom}[\head(T)] = 1 &\text{ and }&T_{dom}[p+1] = T_{dom}[p]+1\,\,,\,\,\,\, \text{ for } p\in \supp(T)
\end{eqnarray*}
\end{defn}
\begin{remark}
Let $T$ be of shape $(i,j)$. Then the map \eqref{map:qmon} identifies $m_{T_{dom}}=Y_{i,j}$. In other words, $T_{dom}$ corresponds to the dominant monomial of all column tableaux $T$ of shape $(i,j)$. More generally, the tableaux corresponding to the descendants of any dominant monomial (not necessarily column) will have the same underlying diagram as their dominant monomial, i.e. the lengths and the indices of each column will be the same.  
\end{remark}

Let us illustrate all these definitions with an example.
\begin{example}
Let $r\geq 5$. Let $T=(T_1,T_2)$ be the general tableau consisting of columns of shapes $\left\{(3,2), (2,1)\right\}$. The first column diagram $T_1$ is of shape $(3,2)$, which means the length of $T_1$ is $3$ and $h(T_1)=\frac{1-i_1-j_1}{2}=-2$. Similarly, the second column diagram $T_2$ is of length $2$ and $h(T_2)=-1$. An example of such general tableau is given below, together with the corresponding collapsed tableau and its dominant tableau.  
\begin{center}
\begin{tikzpicture}
\node at (-3, -2) {General Tableaux};
\node at (0,-0.5) {diagram of $T$};
\node (1) at (0,-2) {$
\begin{ytableau}
\none[\scriptscriptstyle{-2}]&\\
\none[\scriptscriptstyle{-1}] & &\\
\none[\scriptscriptstyle{0}]& &\\
\end{ytableau} $};

\node at (3,-0.5) {tableau $T$};
\node (2) at (3,-2) {$
\begin{ytableau}
\none[\scriptscriptstyle{-2}]&1\\
\none[\scriptscriptstyle{-1}] &3 &4\\
\none[\scriptscriptstyle{0}]&4 &5\\
\end{ytableau} $};

\node at (6,-0.5) {$T_{dom}$};
\node (3) at (6,-2) {$
\begin{ytableau}
\none[\scriptscriptstyle{-2}]&1\\
\none[\scriptscriptstyle{-1}] &2 &1\\
\none[\scriptscriptstyle{0}]&3 &2\\
\end{ytableau} $};

\node at (9, -0.5) {$m_{T_{dom}}$};
\node at (9,-2) {$Y_{3,2}Y_{2,1}$};

\node at (-3, -4) {Collapsed Tableaux};

\node (4) at (0,-4) {$
\begin{ytableau}
\none[]&&\\
\none[] & &\\
\none[]& \\
\end{ytableau} $};

\node (2) at (3,-4) {$
\begin{ytableau}
\none&1&4\\
\none&3 &5\\
\none&4\\
\end{ytableau} $};

\node (2) at (6,-4) {$
\begin{ytableau}
\none&1&1\\
\none&2 &2\\
\none&3\\
\end{ytableau} $};

\node at (9,-4) {$y_3y_2$};

\end{tikzpicture}
\end{center}

\qed
\end{example}

\begin{example}
Consider $\g=\mathfrak{sl}_4$. Let $V$ be the fundamental $U_q(\widehat{\mathfrak{sl}_4})$-module with dominant monomial $Y_{3,0}$. The $U_q(\mathfrak{sl}_4)$-highest weight of this module is $\omega_3$. This is the affine analog of Example \ref{ex:finite}. We give the $q$-character of $V$ below:
\vspace{0.1in}
\begin{center}
\begin{tikzpicture}

\node at (-2,2) {$\chi_q(V)$};
\node at (-1,2) {$=$};

\node (1') at (0,2) {$Y_{3,0}$};

\node (1) at (0,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&1\\
\none[{\color{blue}\scriptstyle{0}}]&2\\
\none[{\color{blue}\scriptstyle{1}}]&{\color{red}3}
\end{ytableau}
};

\node (2') at (3,2) { $Y_{2,1}Y_{3,2}^{-1}$};

\node (2) at (3,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&1\\
\none[{\color{blue}\scriptstyle{0}}]&{\color{red}2}\\
\none[{\color{blue}\scriptstyle{1}}]&4
\end{ytableau}
};


\node (3') at (6,2) { $Y_{1,2}Y_{2,3}^{-1}$};

\node (3) at (6,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&{\color{red}1}\\
\none[{\color{blue}\scriptstyle{0}}]&3\\
\none[{\color{blue}\scriptstyle{1}}]&4
\end{ytableau}
};


\node (4') at (9,2) { $Y_{1,4}^{-1}$};

\node (4) at (9,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&2\\
\none[{\color{blue}\scriptstyle{0}}]&3\\
\none[{\color{blue}\scriptstyle{1}}]&4
\end{ytableau}
};

\node at (1.5,2) {$+$};
\node at (4.5,2) {$+$};
\node at (7.5,2) {$+$};

\path[->,thick]

(1) edge node[above] {$A_{{\color{red}3},{\color{red}3}-2{\color{blue}\cdot 1} }^{-1}$}  (2)
(2) edge node[above] {$A_{{\color{red}2}, {\color{red}2} - 2\cdot {\color{blue} 0}}^{-1}$}  (3)
(3) edge node[above] {$A_{{\color{red}1}, {\color{red} 1} - 2 \cdot ( {\color{blue} -1} )}^{-1}$} (4)
(1') edge[dotted] (1)
(2') edge[dotted] (2)
(3') edge[dotted] (3)
(4') edge[dotted] (4)
;

\end{tikzpicture}
\end{center}
\vspace{0.1in}

\qed

\end{example}
Let $T=(T_1,\ldots,T_k)$ be a general tableaux. The monomial $m_T$ in \eqref{map:qmon} can also be written as:
\begin{eqnarray} \label{eq:aAction}
m_T = m_{T_{dom}} \prod_{l=1}^k \prod_{p\in \supp(T_l)} \prod_{i=(T_l)_{dom}[p]}^{T_l[p]-1} A_{i,i-2p}^{-1}\,.
\end{eqnarray}

We have presented so far the modified crystal basis for $q$-character monomials in \cite{nakajima2002t}. From here on, we define new concepts. 

Consider KR-modules and the associated general tableaux. Consider the KR-module with dominant monomial $\mathbf{Y}_{k,j}^{(i)}=Y_{i,j}Y_{i,j+2}\cdots Y_{i,j+2k-2}$. The diagram associated with this dominant monomial and all its descendants consists of $k$ columns stacked as follows:

\begin{center}
\begin{tikzpicture}
\node at (0,0) {$T$};
\node at (1,0) {$=$};
\node at (3,0) {$
\begin{ytableau}
\none[\scriptstyle{T_1}]&\none[\scriptstyle{T_2}]&\none&\none&\none&\none[\scriptstyle{T_{k-1}}]&\none[\scriptstyle{T_k}]\\
\none	&\none&\none	&\none&\none&\none&\\
\none	&\none&\none 	&\none &\none&&\\
\none	&&\none&	\none	&\none&&\\
		&&\none&	\none[\cdots]&\none&\none[\vdots]&\none[\vdots]	\\
		&&\none&	\none &\none	&&\\
\none[\vdots]&\none[\vdots]&\none &\none& \none &\\
		& &\none&\none\\
		&\none\\
\end{ytableau}$};

\draw [
	thick,
	decoration = {
		brace,
		mirror,
		raise=0.5cm
		},
		decorate
	] (1.4,-1.7) -- (4.5,-1.7)
node [pos=0.5, anchor=north,yshift=-0.55cm] {$k$ columns};


\draw [
	thick,
	decoration = {
		brace,
		raise=0cm
		},
		decorate
	] (4.7,1.5) -- (4.7,-0.5)
node [pos=0.5, anchor=north,xshift=0.5cm,yshift=0.2cm] {$i$ };


\end{tikzpicture}
\end{center}
where each column $T_l$, $1\leq l\leq k$, is of length $i$ with appropriate indices determined by $j_l=j+2l-2$. We call a diagram of this shape the staircase diagram of shape $(i,j,k)$. Notice that the corresponding collapsed diagram is rectangular, as expected by Remark \ref{rectangularT}. 
\begin{defn} \label{defn:KRtableau}
Define $\mathcal{B}_{k,j}^{(i)}$ to be the set of general tableaux given by staircase diagrams $T=(T_1,\ldots, T_k)$ of shape $(i,j,k)$ along with decorations by $\left\{1,2,\ldots, r+1\right\}$ such that
\begin{enumerate}
\item the values of the columns strictly increase from top to bottom, i.e. $T_i[p] < T_i[p+1]$ for all $i,p$. (see Figure~\ref{fig:KRarrows}, left)
\item the values of the diagonals weakly increase from left to right, i.e. $T_i[p] \leq T_{i+1}[p-1]$ for all $i,p$. (see Figure~\ref{fig:KRarrows}, right)
\end{enumerate}


\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

\node at (0,0) {\begin{tikzpicture}
\node at (0.5*\X, \X*6) {$\scriptstyle{T_{1}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at ($(03.center)-(\X,0)$) {$\scriptscriptstyle p-1$};
\node at ($(02.center)-(\X,0)$) {$\scriptscriptstyle p$};
\node at ($(01.center)-(\X,0)$) {$\scriptscriptstyle p+1$};

\node at (1.5*\X, \X*6) {$\scriptstyle{T_2}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};
\node at ($(11.center)+(\X,0)$) {$ \scriptstyle \cdots$};
\node at ($(12.center)+(\X,0)$) {$ \scriptstyle \cdots$};
\node at ($(13.center)+(\X,0)$) {$ \scriptstyle \cdots$};
\node at ($(14.center)+(\X,0)$) {$ \scriptstyle \cdots$};

\draw[arrows={- latex'}] (03.center) -- (02.center);
\draw[arrows={- latex'}] (02.center) -- (01.center);
\draw[arrows={- latex'}] (01.center) -- (00.center);

\draw[arrows={- latex'}] (14.center) -- (13.center);
\draw[arrows={- latex'}] (13.center) -- (12.center);
\draw[arrows={- latex'}] (12.center) -- (11.center);
\end{tikzpicture}};

\node at (3,0) {and};

\node at (6,0) {\begin{tikzpicture}
\node at (0.5*\X, \X*6) {$\scriptstyle{T_{1}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at ($(03.center)-(\X,0)$) {$\scriptscriptstyle p-1$};
\node at ($(02.center)-(\X,0)$) {$\scriptscriptstyle p$};
\node at ($(01.center)-(\X,0)$) {$\scriptscriptstyle p+1$};

\node at (1.5*\X, \X*6) {$\scriptstyle{T_2}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};
\node at ($(11.center)+(\X,0)$) {$ \scriptstyle \cdots$};
\node at ($(12.center)+(\X,0)$) {$ \scriptstyle \cdots$};
\node at ($(13.center)+(\X,0)$) {$ \scriptstyle \cdots$};
\node at ($(14.center)+(\X,0)$) {$ \scriptstyle \cdots$};


\draw[arrows={- latex'},dotted] (03.center) -- (14.center);
\draw[arrows={- latex'},dotted] (02.center) -- (13.center);
\draw[arrows={- latex'},dotted] (01.center) -- (12.center);
\draw[arrows={- latex'},dotted] (00.center) -- (11.center);

\end{tikzpicture}};


\end{tikzpicture}
\end{center}
\caption{Solid and dotted arrows indicate strict and weak inequalities in the direction of the arrows respectively, i.e. $a\rightarrow b$ is equivalent to $a<b$, and $a\dashrightarrow b$ is equivalent to $a\leq b$.}
\label{fig:KRarrows}
\end{figure}

We call a tableau that belongs to $\mathcal{B}_{k,j}^{(i)}$ for some $(i,j,k)$ a KR-tableau. 
\end{defn}

\begin{thm}
The $q$-character of the KR-module $W_{k,j}^{(i)}$ is parametrized by the KR-tableaux in $\mathcal{B}_{k,j}^{(i)}$. That is, we have:
\begin{eqnarray*}
\chi_q(W_{k,j}^{(i)}) = \sum_{T\in \mathcal{B}_{k,j}^{(i)}} m_T\,\, ,
\end{eqnarray*}
where $m_T$ is the monomial associated to the tableau $T$ through the map \eqref{map:qmon}. 
\end{thm}
\begin{proof}
Notice that if we collapse a KR-tableau, we obtain a rectangular semi-standard Young tableau. Since ${\rm{res}}(W_{k,j}^{(i)}) = V_{k\omega_i}$, and $\chi(V_{k\omega_i})$ is given by rectangular semi-standard Young tableaux, the result follows. 
\end{proof}

\subsection{Fundamental cluster diagrams} \label{fundClusterDiagrams}
Recall the fundamental cluster $\mathcal{C}$ defined in \eqref{eq:FundClust}. The column diagrams in $\mathcal{C}$ are shown in Figure~\ref{fig:fundTab}(a). Given $\chi_{k,j(k,i)}^{(i)}\in \mathcal{C}$, the corresponding staircase diagram of shape $(i,j,k)$ is constructed as follows: start with the column diagram $\chi_{1, j(1,i)}^{(i)}$ (Figure~\ref{fig:fundTab}(a)), call it the central column and start adding columns of equal length alternatingly to both sides of the central tableau. When $i$ is odd, we start adding on the left and when $i$ is even, we start adding on the right (see Figure~\ref{fig:fundTab}(b) for an example). 

\begin{figure}[ht]
\begin{tikzpicture}
\node at (0,-3) {\begin{minipage}{0.35\textwidth}(a) Diagrams corresponding to the fun\-da\-men\-tal modules ($k=1$) in $\mathcal{C}$\end{minipage}};
\node at (0,0) {$
\begin{ytableau}
\none[i=]&\none& \none[1]&\none&\none[2]&\none&\none[3]&\none&\none[4]&\none&\none[5]&\none&\none[6]&\none&\none[7]&\none&\none[\cdots]\\
\none&\none&		\none	&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none\\
\none[\scriptscriptstyle-3]&\none&	\none	&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\\
\none[\scriptscriptstyle-2]&\none&	\none	&\none&\none&\none&\none&\none&\none&\none&		&\none&		&\none&\\
\none[\scriptscriptstyle-1]&\none&	\none	&\none&\none&\none&	    &\none&	 &\none&		&\none&		&\none&\\
\none[\scriptscriptstyle0]&\none&			&\none&		&\none&	    &\none&	 &\none&		&\none&		&\none&\\
\none[\scriptscriptstyle1]&\none&	\none	&\none&		&\none&	    &\none&	 &\none&		&\none&		&\none&\\
\none[\scriptscriptstyle2]&\none&	\none	&\none&\none&\none&\none&\none&	 &\none&		&\none&		&\none&\\
\none[\scriptscriptstyle3]&\none&	\none	&\none&\none&\none&\none&\none&\none&\none&\none&\none&		&\none&\\
\end{ytableau}
$};
\node at (9,-3) {\begin{minipage}{0.35\textwidth}(b) $i=3$ and $k=1,2,3,4$. The central and the last column that is added are colored. 
\end{minipage}};
\node at (9,0) {$\begin{ytableau}
\none[k=1]&\none&\none&\none&\none[k=2]&\none&\none&\none&\none[k=3]&\none&\none&\none&\none[k=4]&\none\\
\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none\\
\none&\none&\none&\none&\none&\none&\none&\none&*(blue!50)&\none&\none&\none&\none&\\
*(yellow)&\none&\none&\none&*(yellow)&\none&\none&*(yellow)&*(blue!50)&\none&\none&\none&*(yellow)&\\
*(yellow)&\none&\none&*(blue!50)&*(yellow)&\none&&*(yellow)&*(blue!50)&\none&\none&&*(yellow)&\\
*(yellow)&\none&\none&*(blue!50)&*(yellow)&\none&&*(yellow)&\none&\none&*(blue!50)&&*(yellow)&\none\\
\none&\none&\none&*(blue!50)&\none&\none&&\none&\none&\none&*(blue!50)&&\none&\none\\
\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&*(blue!50)&\none&\none&\none\\
\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none&\none
\end{ytableau}$};
\end{tikzpicture}
\caption{Diagrams corresponding to the fundamental cluster variables}
\label{fig:fundTab}
\end{figure}

\subsection{Twisted multiplication of tableaux} \label{multTableaux}
We will now describe the twisted multiplication $*_\gamma$ from \eqref{gammaMult} on the tableaux. Let $C,T$ be KR-tableaux. In order to simplify our notation of Definition \ref{def:Nakajima}, we denote
\begin{eqnarray}  \label{simpleuv}
\begin{array}{ccc}
u_{i,j}(C):=u_{i,j}(m_C)\,; & v_{i,j}(C):=v_{i,j}(m_C, m_{C_{dom}})\,; &\gamma(C,T):=\gamma(m_C, m_T)  \,.
 \end{array}
\end{eqnarray}

\begin{remark} \label{rmk:gammaAdditive}
In order to compute $\gamma(C,T)$ for any general KR-tableaux $C$ and $T$, it suffices to compute the value of $\gamma$ between column tableaux only. Indeed, suppose $T=(T_l)$. Then,
\begin{eqnarray*}
m_T &=& \prod_{l} \left(\prod_{i,j} Y_{i,j}^{u_{i,j}(T_l)}\right) = \prod_{i,j} Y_{i,j}^{\sum_{l} u_{i,j}(T_l)} \,\, ,\\
m_T &=& \prod_{l} \left( m_{{T_l}_{dom}} \prod_{i,j} A_{i,j}^{-v_{i,j}(T_l)}  \right) = m_{T_{dom}} \prod_{i,j} A_{i,j}^{-\sum_{l} v_{i,j}(T_l)}\,\, .
\end{eqnarray*}
That is, both $u_{i,j}$ and $v_{i,j}$ are additive. Therefore, for any general tableau $C=(C_k)$, since $\gamma(C,T)$ is a linear expression in $u_{i,j}$ and $v_{i,j}$, we have
\begin{eqnarray*}
\gamma(C,T) = \sum_{k,l} \gamma(C_k,T_l) \,.
\end{eqnarray*}
\end{remark}

\begin{defn}
Let $(C,T)$ be a pair of column KR-tableaux. A \textsl{block} in $(C,T)$ at index $p$ is a pair of boxes given by $(C[p], T[p])$. We say there is an $L^\pm$-block, an $N^\pm$-block, or a $U$-block at $p$ in $(C,T)$ if the following inequality conditions hold:
\begin{center}
\begin{tikzpicture}
\node at (0,0) {
\begin{tikzpicture}
\node at (0.5*\X,2.5*\X+.5) {$C$};
\node at (1.5*\X,2.5*\X+.5) {$T$};
\node at (-.5*\X,1.5*\X) {$\scriptscriptstyle p-1$};
\node at (-.5*\X,.5*\X) {$\scriptscriptstyle p$};

\draw (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw[dotted] (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw[dotted] (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\draw[arrows={- latex'},thick] (01.center) -- (10.center);
\draw[arrows={- latex'},thick] (10.center) -- (00.center);

\node at (1*\X,-1*\X) { $L^+$-block};
\end{tikzpicture}
};

\node at (3,0) {
\begin{tikzpicture}
\node at (0.5*\X,2.5*\X+.5) {$C$};
\node at (1.5*\X,2.5*\X+.5) {$T$};
\node at (-.5*\X,1.5*\X) {$\scriptscriptstyle p-1$};
\node at (-.5*\X,.5*\X) {$\scriptscriptstyle p$};

\draw (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw[dotted] (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw[dotted] (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\draw[arrows={latex'-},thick] (00.center) -- (11.center);
\draw[arrows={latex' -},thick] (10.center) -- (00.center);

\node at (1*\X,-1*\X) { $L^-$-block};
\end{tikzpicture}
};

\node at (6,0) {
\begin{tikzpicture}
\node at (0.5*\X,2.5*\X+.5) {$C$};
\node at (1.5*\X,2.5*\X+.5) {$T$};
\node at (-.5*\X,1.5*\X) {$\scriptscriptstyle p-1$};
\node at (-.5*\X,.5*\X) {$\scriptscriptstyle p$};

\draw (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw[dotted] (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw[dotted] (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\draw[arrows={- latex'},thick,dotted] (00.center) -- (11.center);

\node at (1*\X,-1*\X) { $N^+$-block};
\end{tikzpicture}
};

\node at (9,0) {
\begin{tikzpicture}
\node at (0.5*\X,2.5*\X+.5) {$C$};
\node at (1.5*\X,2.5*\X+.5) {$T$};
\node at (-.5*\X,1.5*\X) {$\scriptscriptstyle p-1$};
\node at (-.5*\X,.5*\X) {$\scriptscriptstyle p$};

\draw (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw[dotted] (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw[dotted] (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\draw[arrows={- latex'},thick,dotted] (10.center) -- (01.center);

\node at (1*\X,-1*\X) { $N^-$-block};
\end{tikzpicture}
};
\node at (12,0) {
\begin{tikzpicture}
\node at (0.5*\X,2.5*\X+.5) {$C$};
\node at (1.5*\X,2.5*\X+.5) {$T$};
\node at (-.5*\X,1.5*\X) {$\scriptscriptstyle p-1$};
\node at (-.5*\X,.5*\X) {$\scriptscriptstyle p$};

\draw (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw[dotted] (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw[dotted] (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\draw[arrows={latex' - latex'},thick]  (00.center) -- (10.center);

\node at (1*\X,-1*\X) { $U$-block};
\end{tikzpicture}
};
\end{tikzpicture}
\end{center}
The arrow notation is the same as in Figure~\ref{fig:KRarrows} and two-sided arrow is equivalent to equality. 
\end{defn}
It is important to stress that a block consists of $2$ boxes at index $p$, even though there may be an inequality requirement coming from boxes with index $p-1$, such as the case of $L$ and $N$ blocks.  
\begin{defn}
If there is either an $L$-block or a $U$-block at $p$ in $(C,T)$, for convenience we say there is an $LU$-block at $p$. Any other combinations are allowed (e.g. $L^+UN^-$-block).  
\end{defn}

\begin{remark} \label{rmk:obs1}
As we are dealing with KR-tableaux, inequality conditions in Figure~\ref{fig:KRarrows} are always present. Due to this restriction, any block in a pair of KR-tableaux $(C,T)$ is precisely one of the 5 types: $L^\pm, N^\pm$ or $U$. Moreover, the strictly increasing columns condition puts additional restrictions on the order with which the blocks can appear. For example, (a) an $L^\pm$-block is never followed by an $N^\pm$-block; (b) an $N^\pm$-block is never followed by an $N^\mp$-block, and (c) a $U$-block is never followed by an $N$-block. The reason can be seen by simply composing the arrows. For example,
\begin{center}
\begin{tikzpicture}

\node (1) at (0,0) {
\begin{tikzpicture}
\node at (-1,3) {(a)};
\node at (0.5*\X, \X*5) {$\scriptstyle{C}$};

\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw[dotted] (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (\X*1.5, \X*5) {$\scriptstyle{T}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\node at ($(12)+(0.5,0)$) {$\scriptstyle{L^-}$};
\draw[dotted] (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\node at ($(11)+(0.5,0)$) {$\scriptstyle{N^-}$};

\draw[arrows={- latex'},red] (13.center) -- (02.center);
\draw[arrows={- latex'},red] (02.center) -- (12.center);
\draw[arrows={- latex'},blue,dotted] (11.center) -- (02.center);

\node at (\X*1-1 + 1.5*\D, 2) {$\scriptstyle \text{composing arrows}$};
\draw[arrows={- latex'},thick] (\X*1-.8 + \D,1.5) -- (\X*1+2*\D-1,1.5);

\node at (0.5*\X+3*\D-2, \X*5) {$\scriptstyle{C}$};

\draw (0+3*\D-2, \X*1) rectangle (\X*1+3*\D-2, \X*2) node[fitting node] (c01) {};
\draw (0+3*\D-2, \X*2) rectangle (\X*1+3*\D-2, \X*3) node[fitting node] (c02) {};
\draw[dotted] (0+3*\D-2, \X*3) rectangle (\X*1+3*\D-2, \X* 4) node[fitting node] (c03) {};

\node at (\X*1.5+3*\D-2, \X*5) {$\scriptstyle{T}$};
\draw (\X*1+3*\D-2,\X*1) rectangle (\X*2+3*\D-2,\X*2) node[fitting node] (t11) {};
\draw (\X*1+3*\D-2,\X*2) rectangle (\X*2+3*\D-2,\X*3) node[fitting node] (t12) {};
\draw[dotted] (\X*1+3*\D-2,\X*3) rectangle (\X*2+3*\D-2,\X*4) node[fitting node] (t13) {};
\node at ($(t11)+(1,0)$) {$\scriptstyle{\text{Contrad.}}$}; 

\draw[arrows={- latex'}] (t11.center) -- (t12.center);
\end{tikzpicture}
};


\node (2) at (7.5,0) {

\begin{tikzpicture}
\node at (-1,3) {(b)};

\node at (0.5*\X, \X*5) {$\scriptstyle{C}$};

\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw[dotted] (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (\X*1.5, \X*5) {$\scriptstyle{T}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\node at ($(12)+(0.5,0)$) {$\scriptstyle{N^+}$};
\draw[dotted] (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\node at ($(11)+(0.5,0)$) {$\scriptstyle{N^-}$};

\draw[arrows={- latex'},dotted,red] (02.center) -- (13.center);
\draw[arrows={- latex'},dotted,blue] (11.center) -- (02.center);

\node at (\X*1-1 + 1.5*\D, 2) {$\scriptstyle \text{composing arrows}$};
\draw[arrows={- latex'},thick] (\X*1-.8 + \D,1.5) -- (\X*1+2*\D-1,1.5);

\node at (0.5*\X+3*\D-2, \X*5) {$\scriptstyle{C}$};

\draw (0+3*\D-2, \X*1) rectangle (\X*1+3*\D-2, \X*2) node[fitting node] (c01) {};
\draw (0+3*\D-2, \X*2) rectangle (\X*1+3*\D-2, \X*3) node[fitting node] (c02) {};
\draw[dotted] (0+3*\D-2, \X*3) rectangle (\X*1+3*\D-2, \X* 4) node[fitting node] (c03) {};

\node at (\X*1.5+3*\D-2, \X*5) {$\scriptstyle{T}$};
\draw (\X*1+3*\D-2,\X*1) rectangle (\X*2+3*\D-2,\X*2) node[fitting node] (t11) {};
\draw (\X*1+3*\D-2,\X*2) rectangle (\X*2+3*\D-2,\X*3) node[fitting node] (t12) {};
\draw[dotted] (\X*1+3*\D-2,\X*3) rectangle (\X*2+3*\D-2,\X*4) node[fitting node] (t13) {};
\node at ($(t11)+(1,0)$) {$\scriptstyle{\text{Contrad.}}$}; 

\draw[arrows={- latex'},dotted] (t11.center) -- (t13.center);

\end{tikzpicture}
};
\end{tikzpicture}
\end{center}
\end{remark}

\begin{defn}
Let $(C,T)$ be a pair of column KR-tableaux. We define for each $p\in \mathbb{Z}$ functions $L_p$ and $N_p$ as follows:
\begin{eqnarray*}
L_p(C,T) &=& \left\{ \begin{array}{cl} 
1 & \text{if } C[p-1] < T[p] < C[p]\\
-1 & \text{if } T[p-1] < C[p] < T[p]\\
0 & \text{otherwise} 
\end{array}\right. \\
N_p(C,T) &=&
\left\{\begin{array}{cl}
1 & \text{if }C[p] \leq T[p-1]\\
-1 & \text{if } T[p] \leq C[p-1]\\
0 &\text{otherwise }
\end{array}\right.
\end{eqnarray*}
In other words, $L_p(C,T)$ is $\pm1$ if there is an $L^\pm$-block at $p$ and $0$ otherwise, and similarly for $N_p(C,T)$. When there is no confusion as to which KR-tableaux we are referring to, we will suppress the dependance on $C$ and $T$ and simply write $L_p$ and $N_p$. 
\end{defn}

\begin{defn}
Let $(C,T)$ be a pair of column KR-tableaux with $\supp(C)\cap \supp(T) = \left\{ h, \ldots, t\right\}$. The block-tableau of $(C,T)$, denoted $B_{CT}$, is a single column diagram with $\supp(B_{CT}) = \left\{ h, \ldots, t, t+1\right\}$ decorated with letters $\left\{L^+,L^-,N^+,N^-,U\right\}$, such that the value of $B_{CT}$ at index $p$ is given by the corresponding block in $(C,T)$ at $p$. 
\end{defn}

\begin{example}
Block-tableau is a convenient way to describe all the blocks of $(C,T)$ at once. Saying there is an $L^+$-block at $p$ in $(C,T)$ is equivalent to $B_{CT}[p]=L^+$.
\begin{center}
\begin{tikzpicture}
\node at (0,0) {$
\ytableausetup
{mathmode, boxsize=1em}
\begin{ytableau}
\none[\scriptstyle index]&\none&\none[C] 	&\none	&\none[T]\\
\none&\none&\none	&\none	&\none\\
\none&\none&1		&\none	&\none[\scriptstyle 0]\\
\none[\scriptstyle 0]&\none&3		&\none	&2\\
\none[\scriptstyle 1]&\none&4		&\none	&3\\
\none[\scriptstyle 2]&\none&7		&\none	&9\\
\none[\scriptstyle 3]&\none&8		&\none &\none[\scriptstyle \infty]	\\
\end{ytableau}$};

\draw[->] (2,0)--(4.5,0);

\node at (6,0) {$
\begin{ytableau}
\none[\scriptstyle index]&\none&\none[B_{CT}] 		\\
\none&\none&\none	\\
\none&\none&\none		\\
\none[\scriptstyle 0]&\none&\scriptscriptstyle L^+		\\
\none[\scriptstyle 1]&\none&\scriptscriptstyle N^-		\\
\none[\scriptstyle 2]&\none&\scriptscriptstyle L^-		\\
\none[\scriptstyle 3]&\none&\scriptscriptstyle N^+			
\end{ytableau}$};
\end{tikzpicture}
\end{center}

\end{example}

\begin{defn} \label{def:pairs}
Let $(C,T)$ be a pair of column tableaux. We say $(C,T)$ is a \emph{fundamental} pair if $\head(C) \geq \head(T)$ and $\tail(C)\leq \tail(T)$. We say $(C,T)$ is \emph{anti-fundamental} if $(T,C)$ is fundamental. We say $(C,T)$ is a \emph{regular} pair if $\head(C) > \head(T)$ and $\tail(C) > \tail(T)$. We say $(C,T)$ is \emph{anti-regular} if $(T,C)$ is regular (see Figure~\ref{types}). 
\end{defn}

\begin{figure}[ht] 
\begin{center}
\begin{tikzpicture}
\node at (0,0.2) {Fundamental};
\node at (0,-2) {$
\begin{ytableau}
\none[\scriptstyle index]			&\none		&\none[C]		&\none&\none[T]	\\
\none					&\none		&\none[\scriptstyle 0]			&\none&\none[\scriptstyle 0]		\\
\none[\scriptstyle h-1]		&\none		&\none[\scriptstyle 0]	&\none&			\\
\none[\scriptstyle h]			&\none		&*(yellow)			&\none&*(yellow)			\\
\none					&\none		&			&\none&			\\
\none[\scriptstyle t]			&\none		&			&\none&			\\
\none[\scriptstyle t+1]		&\none		&\none[\scriptstyle \infty]		&\none&*(yellow)			\\
\none[\scriptstyle t+2]		&\none		&\none[\scriptstyle \infty]		&\none&	\none[\scriptstyle \infty]		\\
\end{ytableau}
$};

\node at (3.5,-0.3) {$B_{CT}$};
\node at (3.5,-2) {$
\begin{ytableau}
\none\\
\none\\
*(yellow)\\
\\
\\
*(yellow)\\
\end{ytableau}
$};

\node at (7,0.2) {Regular};

\node at (7,-2) {$
\begin{ytableau}
\none[C]			&\none&\none[T]			&\none&\none[\scriptstyle index]							\\
\none[\scriptstyle 0]	&\none&\none[\scriptstyle 0]	&		\none&\none			\\
\none[\scriptstyle 0]	&\none&					&		\none&\none[\scriptstyle h-1]		\\
*(yellow)			&\none&*(yellow)			&			\none&\none[\scriptstyle h]		\\
				&\none&					&	\none&\none						\\
				&\none&					&			\none&\none[\scriptstyle t]		\\
*(yellow)			&\none&\none[\scriptstyle \infty]	&		\none&\none[\scriptstyle t+1]			\\
\none[\scriptstyle \infty]			&\none&\none[\scriptstyle \infty]	&	\none&\none[\scriptstyle t+2]		\\
\end{ytableau}
$};
\end{tikzpicture}
\end{center}
\caption{Types of pairs of column diagrams}
\label{types}
\end{figure}


The reason for the name fundamental pair is because of the diagrams in $\mathcal{C}$ corresponding to the fundamental modules, i.e KR-modules with $k=1$. Every pair of fundamental modules in $\mathcal{C}$ forms a fundamental or an anti-fundamental pair as in Definition \ref{def:pairs} (see Figure~\ref{fig:fundTab}(a)). 

\begin{lemma} \label{rmk:obs2}

Let $(C,T)$ be a pair of column KR-tableaux and suppose $(C,T)$ is either fundamental or regular. Denote $h=\head(C,T)$ and $t=\tail(C,T)$. Then,

(a) $B_{CT}[h]$ is never $N^-$ since $C[h-1] = 0 \ngeq T[p]$.

(b) If $(C,T)$ is fundamental, $B_{CT}[t+1]$ is never $L^+$ or $N^+$ since $T[t]<T[t+1] \leq \infty = C[t]$.

(c) If $(C,T)$ is regular, $B_{CT}[t+1]$ is never $L^-$ or $N^-$ since $C[t]<C[t+1] < \infty = T[t]$.
\end{lemma}
\begin{proof}
The statements are clear from the following pictures:
\begin{center}
\begin{tikzpicture}
\node at (0,0) {\begin{tikzpicture}

\node at (0+\D+0.5*\X-5, \X*5-0.5) {(a)};
\node at (0+\D+0.5*\X-4, \X*5-0.5) {$\scriptstyle{C}$};
\node at (0+\D+\X*1.5-4, \X*5-0.5) {$\scriptstyle{T}$};

\draw[white] (0+\D-4, \X*2) rectangle (\X*1+\D-4, \X*3) node[fitting node] (c02) {};
\draw (0+\D-4, \X*1) rectangle (\X*1+\D-4, \X*2) node[fitting node] (c01) {};
\node at ($(c01.center)-(\X,0)$) {$\scriptscriptstyle h$};


\draw[dotted] (\X*1+\D-4,\X*2) rectangle (\X*2+\D-4,\X*3) node[fitting node] (t12) {};
\draw (\X*1+\D-4,\X*1) rectangle (\X*2+\D-4,\X*2) node[fitting node] (t11) {};

\draw[arrows={- latex'},red] (c02.center) -- (t11.center);
\node[fill=white] at (c02.center) {$\scriptstyle{0}$};
\node at ($(t11)+(1,0)$) {$\scriptstyle{\text{not }N^-}$};
\end{tikzpicture}
};

\node at (5.5,0) {\begin{tikzpicture}
\node at (0+\D+0.5*\X-1, \X*5-0.5) {(b)};
\node at (0+\D+0.5*\X, \X*5-0.5) {$\scriptstyle{C}$};
\node at (0+\D+\X*1.5, \X*5-0.5) {$\scriptstyle{T}$};

\draw[white] (0+\D, \X*1) rectangle (\X*1+\D, \X*2) node[fitting node] (c01) {};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (c02) {};
\node at ($(c02.center)-(\X,0)$) {$\scriptscriptstyle t$};

\draw (\X*1+\D,\X*2) rectangle (\X*2+\D,\X*3) node[fitting node] (t12) {};
\draw[dotted] (\X*1+\D,\X*1) rectangle (\X*2+\D,\X*2) node[fitting node] (t11) {};

\draw[arrows={- latex'},red,dotted] (t11.center) -- ($(t11.west)-(.1,0)$);
\draw[arrows={-latex'},blue]  (t12.center) -- (t11.center);
\node at (c01.center) {$\scriptstyle{\infty}$};
\node at ($(t11)+(1.3,0)$) {$\scriptstyle{\text{not }L^+N^+}$};
\end{tikzpicture}
};

\node at (11,0) {\begin{tikzpicture}
\node at (0+\D+0.5*\X+3, \X*5-0.5) {(c)};
\node at (0+\D+0.5*\X+4, \X*5-0.5) {$\scriptstyle{C}$};
\node at (0+\D+\X*1.5+4, \X*5-0.5) {$\scriptstyle{T}$};

\draw[white] (\X*1+\D+4,\X*1) rectangle (\X*2+\D+4,\X*2) node[fitting node] (11) {};
\draw (\X*1+\D+4,\X*2) rectangle (\X*2+\D+4,\X*3) node[fitting node] (12) {};

\draw (0+\D+4, \X*1) rectangle (\X*1+\D+4, \X*2) node[fitting node] (01) {};
\draw (0+\D+4, \X*2) rectangle (\X*1+\D+4, \X*3) node[fitting node] (02) {};
\node at ($(02.center)-(\X,0)$) {$\scriptscriptstyle t$};


\draw[arrows={- latex'},red] (01.center) -- ($(11.west)+(0.1,0)$);
draw[arrows={-latex'},blue] (12.center) -- (11.center);
\node  at (11.center) {$\scriptstyle{\infty}$};
\node at ($(11)+(1.5,0)$) {$\scriptstyle{\text{not }L^-N^-}$};
\end{tikzpicture}
};

\end{tikzpicture}
\end{center}
\end{proof}

\begin{remark}
It suffices to compute $\gamma(C,T)$ for $(C,T)$ either fundamental or regular. If $(C,T)$ is of neither type, then $(T,C)$ is either fundamental or regular and $\gamma(C,T)=-\gamma(T,C)$. 
\end{remark}

The value of $\gamma(C,T)$ depends on the occurrence of $L$ and $N$ blocks in $(C,T)$ encoded in the functions $L_p$ and $N_p$. More precisely,

\begin{thm}\label{gamma}
Let $(C,T)$ be a pair of column KR-tableaux and let $\supp(C)\cap \supp(T) := \left\{ h, \ldots, t\right\}$ (see Figure~\ref{types}). Then
\begin{eqnarray*}
\gamma(C,T) = \sum_{p=h}^{t} L_p (C,T) + \left\{ \begin{array}{cc} 
N_{t+1}(C,T) &\text{ if } (C,T) \text{ is (anti-)fundamental}\\
L_{t+1}(C,T) &\text{ otherwise}
\end{array}\right.  \,\, .
\end{eqnarray*}
\end{thm}

\begin{remark}
In other words, $\gamma(C,T)$ counts the number of $L^+$-blocks minus the number of $L^-$-blocks in $(C,T)$ with  one more contribution from the tail block. 
\end{remark}

\begin{proof} 
Let's decompose $L_p$ into its positive $L_p^+$ and negative $L_p^-$ parts such that 
\begin{eqnarray*}
L_p = L_p^+ - L_p^- 
\end{eqnarray*}
Recall that $\gamma(C,T) = d(C,T) - d(T,C)$. The calculation of $d(C,T)$ is borrowed from \cite{nakajima2002t}. For the reader's convenience, we reproduce it here. 

We want to compute (see Definition \ref{def:Nakajima}, Remark \ref{rmk:simplegamma}, and \eqref{simpleuv} )
\begin{equation} \label{d}
d(C,T)=\sum_{i,p} v_{i,i-2p}(C)u_{i,i-2p-1}(T) + \sum_{i,p}u_{i,i-2p}(C_{dom})v_{i,i-2p-1}(T) \,\, .
\end{equation}
Let $\left< condition \right>$ be $1$ if the $condition$ is true and $0$ otherwise. By \eqref{eq:aAction} and Definition \ref{uv}, we know
\begin{equation} \label{1}
v_{i,i-2p}(C) = \left<C_{dom}[p] \leq i \leq C[p]-1\right> \,\, .
\end{equation}
By \eqref{map:qmon} and Definition \ref{uv}, we have
\begin{equation} \label{2}
u_{i,i-2p-1}(T) = \left<T[p]=i\right> - \left<T[p+1]=i+1\right>\,\, .
\end{equation}
Putting \eqref{1} and \eqref{2} together,
\begin{eqnarray*}
&&\sum_i v_{i,i-2p}(C) u_{i,i-2p-1}(T) \\
&&\hspace{0.2in} = \sum_i \left<C_{dom}[p] \leq i \leq C[p]-1\right>  \left(  \left<T[p]=i\right> - \left<T[p+1]=i+1\right> \right) \\
&&\hspace{0.2in} = \left<C_{dom}[p] \leq T[p] < C[p]\right> - \left<C_{dom}[p + 1] \leq T[p+1] \leq C[p]\right> \,\, ,
\end{eqnarray*}
where we used $C_{dom}[p+1] = C_{dom}[p]+1$. We want to sum up the above value for all $p$. By shifting the summation index $p$, we consider the sums of the form
\begin{eqnarray*}
- \left< C_{dom}[p] \leq T[p] \leq C[p-1] \right> &+& \left< C_{dom}[p] \leq T[p] < C[p]\right> \\
&& \hspace{0.2in} = \left< C[p-1]<T[p]<C[p]\right> = L_p^+ \,\, .
\end{eqnarray*}
Now, summing over all $p$, we obtain:
\begin{equation} \label{A}
\sum_{i,p} v_{i,i-2p}(C)u_{i,i-2p-1}(T) = \sum_{p=h}^t L_p^+ - \left< C_{dom}[t] < T[t+1] \leq C[t]\right> \,\, ,
\end{equation}
which gives the first half of $d(C,T)$. The second half of \eqref{d} consists of a single term. That is because $m_{C_{dom}} = Y_{i, j}$, where $i = \length(C)= \tail(C)-\head(C) + 1$ and $j= -\tail(C) - \head(C)$. By setting $j = i-2p+1$, we evaluate $p^*=\tail(C) + 1$ and $i^*=\length(C)$ are the only values for which $u_{i^*,i^*-2p^*+1}(C_{dom}) = 1$. Therefore, 
\begin{eqnarray} 
&&\sum_{i,p}u_{i,i-2p+1}(C_{dom})v_{i,i-2p}(T) \nonumber \\ 
&&\hspace{0.2in} = v_{i^*,i^*-2p^*}(T) = \left<T_{dom}[p^*]\leq i^*\leq T[p^*]-1\right> \nonumber\\
&&\hspace{0.2in} = \left<  T_{dom}[\tail(C) + 1]\leq \length(C) \leq T[\tail(C)+1]-1 \right> \label{B} \,\, .
\end{eqnarray}
By adding \eqref{A} and \eqref{B}, we obtain $d(C,T)$. The value of $d(T,C)$ is obtained by switching $C$ and $T$, which gives a sum of $L_p^-$ in the equivalent of \eqref{A}. Putting everything together, we obtain
\begin{eqnarray*}
\gamma(C,T) = d(C,T) - d(T,C) = \sum_{p=n}^m (L_p^+ - L_p^-)  + \Bd(C,T) \,\, ,
\end{eqnarray*}
where $\Bd(C,T)$ is the boundary term given by
\begin{eqnarray*}
\Bd(C,T) &=&  \left< T_{dom}[t] < C[t+1] \leq T[t]\right>  - \left< C_{dom}[t] < T[t+1] \leq C[t]\right> \\
&&+ \left<  T_{dom}[\tail(C) + 1]\leq \length(C)< T[\tail(C)+1] \right> \\
&&- \left<  C_{dom}[\tail(T) + 1]\leq \length(T)< C[\tail(T)+1] \right>\,\, .
\end{eqnarray*}
We simplify the boundary term next. We consider two cases: when $(C,T)$ is fundamental and when $(C,T)$ is regular. 

Suppose $(C,T)$ is fundamental. Then $\tail(C) = t$, $\tail(T)\geq t$, and $C[p]=C_{dom}[p]=\infty$ for $p\geq t+1$ (see Figure~\ref{types}). Notice that,
\begin{equation}\label{eq:rel1}
C_{dom}[t] = \length(C) \leq \length(T[\head(T),t+1]) = T_{dom}[t+1] \leq T[t+1] \,\, .
\end{equation}
The following three terms from $\Bd(C,T)$ are zero:
\begin{eqnarray*}
 0 &=& \left< T_{dom}[t] < \underbrace{C[t+1]}_\text{$\infty$} \leq \underbrace{T[t]}_\text{$\neq \infty$}\right> = \left<  \underbrace{T_{dom}[t + 1]\leq \length(C)}_\text{never true by Eq. \ref{eq:rel1}} < T[t+1] \right> \\&=& \left<  \underbrace{C_{dom}[\tail(T) + 1]}_\text{$\infty$ since $\tail(T)\geq t$} \leq \length(T)< C[\tail(T)+1] \right>\,\, .
\end{eqnarray*}
Thus,
\begin{equation}
\Bd(C,T) = \left< \underbrace{C_{dom}[t] < T[t+1]}_\text{always true by Eq. \ref{eq:rel1}} \leq C[t] \right> = \left< T[t+1] \leq C[t]\right> = N_{t+1}
\end{equation}
Suppose $(C,T)$ is regular. Then $\tail(T) = t$, $\tail(C) \geq t$, and $T[p]= T_{dom}[p]=\infty$ for all $p\geq t+1$. The following two terms from $\Bd(C,T)$ are zero:
\begin{eqnarray*}
0 &=& \left< C_{dom}[t] < \underbrace{T[t+1]}_\text{$\infty$} \leq \underbrace{C[t]}_\text{$\neq \infty$} \right> \\
&=&\left<  \underbrace{T_{dom}[\tail(C) + 1]}_\text{$\infty$ since $\tail(C)\geq t$}\leq \length(C) < T[\tail(C)+1] \right> \,.
\end{eqnarray*}
Notice that
\begin{eqnarray*}
C_{dom}[t+1] &=& t+1 - \head(C) + 1 \\&<& \underbrace{t - \head(C) + 1 \leq \tail(T) - \head(T) + 1}_\text{$\head(C)>\head(T)$ (see Figure~\ref{types}) and $\tail(T)=t$} = \length(T) \,\, .
\end{eqnarray*}
Therefore, removing the condition that is always satisfied and substituting $\length(T)=T_{dom}[t]$, we obtain:
\begin{eqnarray*}
\Bd(C,T) &=& \left< T_{dom}[t] < C[t+1] \leq T[t]\right> - \left<  T_{dom}[t]< C[t+1] \right> \\
&=& - \left< T[t]<C[t+1] \right> = - \left< T[t]<C[t+1] <T[t+1]\right> = - L_{t+1}^- = L_{t+1} \,\, ,
\end{eqnarray*}
where we added the condition $C[t+1] < T[t+1]=\infty$, which is always satisfied and, therefore, does not affect the outcome. 
\end{proof}

\subsection{Exchanging boxes, compatibility conditions, and the involution $\sigma$} \label{inv}
In this section, we define the core concept of exchanging boxes of tableaux, which defines the map $\sigma$ and the three subsets $\mathcal{P}_0,\mathcal{P}_1,\mathcal{P}_{-1}$ of the set $\mathcal{B}_{k,j}^{(i)}\times \mathcal{B}_{k',j'}^{(i')}$ for any $(i,j,k)$ and $(i',j',k')$.  


\begin{defn}
Let $(C,T)$ be a pair of column tableaux. A \emph{strip} in $(C,T)$ is a slice of $(C,T)$ given by $(C[p_0,p_1], T[p_0,p_1])$ for some $p_0<p_1\in \supp(C)\cap \supp(T)$. An \emph{$L$-strip} is a strip that starts with an $L$-block and includes all the $N$-blocks that follow. 
\end{defn}

\begin{defn}
Let $(C,T)$ be a pair of column KR-tableaux. We say an $L$-strip in $(C,T)$ is \emph{column-compatible} if it is possible to exchange the boxes in the $L$-strip and obtain a pair of valid column KR-tableaux. That is, the new tableaux also have strictly increasing columns.  
\end{defn}

\begin{example} \label{ex:columncompat}
Here is an $L^+$-strip starting at $p=0$ and ending at $p=2$. It is column-compatible because it is possible to exchange $C[0,2]$ and $T[0,2]$ and obtain a new pair of column tableaux $(C',T')$. Notice that both $C'$ and $T'$ have strictly increasing columns.  
\begin{center}
\begin{tikzpicture}
\node at (0,0) {$
\ytableausetup
{mathmode, boxsize=1em}
\begin{ytableau}
\none[p]&\none&\none[C] 	&\none	&\none[T]\\
\none&\none&\none	&\none	&\none\\
\none&\none&1		&\none	&\none\\
\none[\scriptstyle 0]&\none&*(yellow)\mathbf{3}		&\none	&*(yellow)\mathbf{2}\\
\none[\scriptstyle 1]&\none&*(yellow)\mathbf{4}		&\none	&*(yellow)\mathbf{3}\\
\none[\scriptstyle 2]&\none&*(yellow)\mathbf{5}		&\none	&*(yellow)\mathbf{4}\\
\none &\none&7		&\none	&9\\
\none&\none&8		&\none	\\
\end{ytableau}$};

\draw[->] (2,0)--(4.5,0);
\node at (3.2,0)[above] {exchanging};

\node at (6,0) {$
\begin{ytableau}
\none[C'] 	&\none	&\none[T']\\
\none	&\none	&\none\\
1		&\none	&\none\\
*(yellow)\mathbf{2}		&\none	&*(yellow)\mathbf{3}\\
*(yellow)\mathbf{3}		&\none	&*(yellow)\mathbf{4}\\
*(yellow)\mathbf{4}		&\none	&*(yellow)\mathbf{5}\\
7		&\none	&9\\
8		&\none	
\end{ytableau}$};
\end{tikzpicture}
\end{center}
\end{example}

\begin{remark} \label{cross-pattern}
For a strip to be column-compatible, it must begin and end with an equality condition of a cross-pattern, which occurs at $LU$-blocks. Indeed, 

\begin{center}
\begin{tikzpicture}

\node at (-5,0) {
\begin{tikzpicture}
\node at (0.5*\X, \X*5) {$\scriptstyle{C}$};

\draw[dotted] (0,-1*\X) rectangle (\X*1,\X*0) node[fitting node] (c-1) {};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (c00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (c01) {};
\draw (0, \X*2) rectangle (\X*1, \X* 3) node[fitting node] (c03) {};
\draw[dotted] (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (c04) {};

\node at (1.5*\X, \X*5) {$\scriptstyle{T}$};
\draw[dotted] (\X*1,-1*\X) rectangle (\X*2,\X*0) node[fitting node] (t-1) {};
\draw (\X*1,\X*0) rectangle (\X*2,\X*1) node[fitting node] (t00) {};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (t01) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (t03) {};
\draw[dotted] (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (t04) {};

\node at (\X*1, -2*\X) {$\scriptstyle \text{cross-pattern}$
};

\draw[arrows={- latex'}] (c04.center) -- (t03.center);
\draw[arrows={- latex'}] (t04.center) -- (c03.center);
\draw[arrows={- latex'}] (t03.center) -- (t01.center);
\draw[arrows={- latex'}] (t01.center) -- (t00.center);
\draw[arrows={- latex'},red] (t00.center) -- (c-1.center);

\draw[arrows={- latex'}] (c03.center) -- (c01.center);
\draw[arrows={- latex'}] (c01.center) -- (c00.center);
\draw[arrows={- latex'},red] (c00.center) -- (t-1.center);

\end{tikzpicture}
};
   
\draw[arrows={latex' - latex'},thick] (-3,-.75) -- (-2,-.75);

\node at (0,-1) {
\begin{tikzpicture}
\draw[dotted] (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw[dotted] (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\draw[arrows={- latex'},thick] (01.center) -- (10.center);
\draw[arrows={- latex'},thick] (10.center) -- (00.center);
\draw[arrows={- latex'},thick,red] (11.center) -- (10.center);


\node at (1*\X,-1*\X) { $\scriptstyle L^+-\text{block}$};
\end{tikzpicture}
};

\node at (3,-1) {
\begin{tikzpicture}
\draw[dotted] (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw[dotted] (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\node at (1*\X,-1*\X) { $\scriptstyle L^- -\text{block}$};

\draw[arrows={latex'-},thick] (00.center) -- (11.center);
\draw[arrows={latex' -},thick] (10.center) -- (00.center);
\draw[arrows={- latex'},thick,red] (01.center) -- (00.center);
\end{tikzpicture}
};

\node at (6,-1) {
\begin{tikzpicture}
\draw[dotted] (0*\X,0*\X) rectangle (1*\X,1*\X) node[fitting node] (00) {};
\draw (0*\X,1*\X) rectangle (1*\X,2*\X) node[fitting node] (01) {};
\draw[dotted] (1*\X,0*\X) rectangle (2*\X,1*\X) node[fitting node] (10) {};
\draw (1*\X,1*\X) rectangle (2*\X,2*\X) node[fitting node] (11) {};

\node at (1*\X,-1*\X) { $\scriptstyle U -\text{block}$};

\draw[arrows={latex' - latex'},thick]  (00.center) -- (10.center);
\draw[arrows={- latex'},thick,red] (11.center) -- (10.center);
\draw[arrows={- latex'},thick,red] (01.center) -- (00.center);
\end{tikzpicture}
};
\end{tikzpicture}
\end{center}
An $L$-strip can either be followed by an $LU$-block or by nothing at all. In the first case, the $L$-strip is always column-compatible. In particular, this means only the very last $L$-strip is potentially not column-compatible. All others are column-compatible since they are, by definition, at the very least followed by the next $L$-block.
\end{remark}

\begin{example} \label{ex:noncolumncompat}
Here is an $L^-$-strip starting at $p=0$ and ending at $p=3$. It is not column-compatible since exchanging $C[0,3]$ and $T[0,3]$ will violate the strictly increasing columns condition post-exchange. 
\begin{center}
\begin{tikzpicture}
\node at (0,0) {$
\ytableausetup
{mathmode, boxsize=1em}
\begin{ytableau}
\none[p]&\none&\none[C] 	&\none	&\none[T]\\
\none&\none&\none	&\none	&\none\\
\none&\none&1		&\none	&\none\\
\none[\scriptstyle 0]&\none&*(yellow)\mathbf{2}		&\none	&*(yellow)\mathbf{3}\\
\none[\scriptstyle 1]&\none&*(yellow)\mathbf{3}		&\none	&*(yellow)\mathbf{4}\\
\none[\scriptstyle 2]&\none&*(yellow)\mathbf{4}		&\none	&*(yellow)\mathbf{5}\\
\none[\scriptstyle 3] &\none&*(yellow)\mathbf{5}		&\none	&*(yellow)\mathbf{6}\\
\none&\none&6	&\none	\\
\end{ytableau}$};

\draw[->] (2,0)--(4.5,0);
\node at (3.2,0)[above] {block-tableau};

\node at (6,0) {$
\begin{ytableau}
\none[B_{CT}] 	\\
\none	\\
\none		\\
\scriptscriptstyle L^-		\\
\scriptscriptstyle N^+		\\
\scriptscriptstyle N^+		\\
\scriptscriptstyle N^+		\\
\scriptscriptstyle N^+	\\
\end{ytableau}$};
\end{tikzpicture}
\end{center}
\end{example}

\begin{defn} \label{defn:compat}
Let $C=(C_{0}, C_1)$ and $T=(T_{0},T_1)$ be KR-tableaux and let $B_{i,j}$ denote the block-tableau of $(C_i,T_j)$. 

Suppose $(C_1,T_1)$ forms an $L$-strip.  We say $(C_1,T_1)$ is \emph{left-compatible} if the weakly increasing diagonals conditions in Definition \ref{defn:KRtableau}(2) are not violated when $C_1$ and $T_1$ are exchanged. Pictorially, $(C_1,T_1)$ is left-compatible if conditions $(lC)$ (left-compatibility from $C$) and $(lT)$ (left-compatibility from $T$) are satisfied:

\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptstyle{C_{0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (1.5*\X, \X*6) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};

\node at (0.5*\X+\D, \X*6) {$\scriptstyle{T_{0}}$};
\draw (0+\D,0) rectangle (\X*1+\D,\X*1) node[fitting node] (t00) {};
\draw (0+\D, \X*1) rectangle (\X*1+\D, \X*2) node[fitting node] (t01) {};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};

\node at (1.5*\X+\D, \X*6) {$\scriptstyle{T_1}$};
\draw (\X*1+\D,\X*1) rectangle (\X*2+\D,\X*2) node[fitting node] (t11) {};
\draw (\X*1+\D,\X*2) rectangle (\X*2+\D,\X*3) node[fitting node] (t12) {};
\draw (\X*1+\D,\X*3) rectangle (\X*2+\D,\X*4) node[fitting node] (t13) {};
\draw (\X*1+\D,\X*4) rectangle (\X*2+\D,\X*5) node[fitting node] (t14) {};

\draw[arrows={- latex'},dotted] (03.center) -- (t14.center);
\draw[arrows={- latex'},dotted] (02.center) -- (t13.center);
\draw[arrows={- latex'},dotted] (01.center) -- (t12.center);
\draw[arrows={- latex'},dotted] (00.center) -- (t11.center);

\draw[arrows={- latex'},dotted,red] (t03.center) -- (14.center);
\draw[arrows={- latex'},dotted,red] (t02.center) -- (13.center);
\draw[arrows={- latex'},dotted,red] (t01.center) -- (12.center);
\draw[arrows={- latex'},dotted,red] (t00.center) -- (11.center);

\draw[arrows={- latex'},thick] (4,1.5) -- (5,1.5);

\node at (0 + 3*\D+.23, \X*6) {$\scriptscriptstyle{B_{0,1}}$};
\draw (0+3*\D,0) rectangle (\X*1+3*\D,\X*1) node[fitting node] (b00) {};
\draw (0+3*\D, \X*1) rectangle (\X*1+3*\D, \X*2) node[fitting node] (b01) {};
\draw (0+3*\D, \X*2) rectangle (\X*1+3*\D, \X*3) node[fitting node] (b02) {};
\draw (0+3*\D, \X*3) rectangle (\X*1+3*\D, \X* 4) node[fitting node] (b03) {};

\node at (b00.center) {$\scriptstyle{N^+}$};
\node at (b01.center) {$\scriptstyle{N^+}$};
\node at (b02.center) {$\scriptstyle{N^+}$};
\node at (b03.center) {$\scriptstyle{N^+}$};

\node at ($(b00)+(0.25,-1)$) {$\scriptstyle{(lC)}$};

\node at (2*\X+3*\D-.23, \X*6) {$\scriptscriptstyle{B_{1,1}}$};
\draw (\X*1+3*\D,\X*1) rectangle (\X*2+3*\D,\X*2) node[fitting node] (b11) {};
\draw (\X*1+3*\D,\X*2) rectangle (\X*2+3*\D,\X*3) node[fitting node] (b12) {};
\draw (\X*1+3*\D,\X*3) rectangle (\X*2+3*\D,\X*4) node[fitting node] (b13) {};
\draw (\X*1+3*\D,\X*4) rectangle (\X*2+3*\D,\X*5) node[fitting node] (b14) {};

\node at (b11.center) {$\scriptstyle{N}$};
\node at (b12.center) {$\scriptstyle{N}$};
\node at (b13.center) {$\scriptstyle{N}$};
\node at (b14.center) {$\scriptstyle{L}$};

\node at (8.5, 1.5) {and};

\node at (0 + 5*\D+.23, \X*6) {$\scriptscriptstyle{B_{1,0}}$};
\draw (0+5*\D,0) rectangle (\X*1+5*\D,\X*1) node[fitting node] (g00) {};
\draw (0+5*\D, \X*1) rectangle (\X*1+5*\D, \X*2) node[fitting node] (g01) {};
\draw (0+5*\D, \X*2) rectangle (\X*1+5*\D, \X*3) node[fitting node] (g02) {};
\draw (0+5*\D, \X*3) rectangle (\X*1+5*\D, \X* 4) node[fitting node] (g03) {};

\node at (g00.center) {$\scriptstyle{N^-}$};
\node at (g01.center) {$\scriptstyle{N^-}$};
\node at (g02.center) {$\scriptstyle{N^-}$};
\node at (g03.center) {$\scriptstyle{N^-}$};

\node at (2*\X+5*\D-.23, \X*6) {$\scriptscriptstyle{B_{1,1}}$};
\draw (\X*1+5*\D,\X*1) rectangle (\X*2+5*\D,\X*2) node[fitting node] (g11) {};
\draw (\X*1+5*\D,\X*2) rectangle (\X*2+5*\D,\X*3) node[fitting node] (g12) {};
\draw (\X*1+5*\D,\X*3) rectangle (\X*2+5*\D,\X*4) node[fitting node] (g13) {};
\draw (\X*1+5*\D,\X*4) rectangle (\X*2+5*\D,\X*5) node[fitting node] (g14) {};

\node at (g11.center) {$\scriptstyle{N}$};
\node at (g12.center) {$\scriptstyle{N}$};
\node at (g13.center) {$\scriptstyle{N}$};
\node at (g14.center) {$\scriptstyle{L}$};

\node at ($(g00)+(0.25,-1)$) {$\scriptstyle{(lT)}$};

\end{tikzpicture}
\end{center}

Suppose $(C_0,T_0)$ forms an $L$-strip. We say $L$-strip $(C_0,T_0)$ is \emph{right-compatible} if the weakly increasing diagonals condition in Definition \ref{defn:KRtableau}(2) is not violated when the boxes in $C_0$ and $T_0$ are exchanged. Pictorially, $(C_0,T_0)$ is right-compatible if conditions $(rC)$ (right-compatibility from $C$) and $(rT)$ (right-compatibility from $T$) are satisfied:

\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptstyle{C_{0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (1.5*\X, \X*6) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};

\node at (0.5*\X+\D, \X*6) {$\scriptstyle{T_{0}}$};
\draw (0+\D,0) rectangle (\X*1+\D,\X*1) node[fitting node] (t00) {};
\draw (0+\D, \X*1) rectangle (\X*1+\D, \X*2) node[fitting node] (t01) {};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};

\node at (1.5*\X+\D, \X*6) {$\scriptstyle{T_1}$};
\draw (\X*1+\D,\X*1) rectangle (\X*2+\D,\X*2) node[fitting node] (t11) {};
\draw (\X*1+\D,\X*2) rectangle (\X*2+\D,\X*3) node[fitting node] (t12) {};
\draw (\X*1+\D,\X*3) rectangle (\X*2+\D,\X*4) node[fitting node] (t13) {};
\draw (\X*1+\D,\X*4) rectangle (\X*2+\D,\X*5) node[fitting node] (t14) {};

\draw[arrows={- latex'},dotted] (03.center) -- (t14.center);
\draw[arrows={- latex'},dotted] (02.center) -- (t13.center);
\draw[arrows={- latex'},dotted] (01.center) -- (t12.center);
\draw[arrows={- latex'},dotted] (00.center) -- (t11.center);

\draw[arrows={- latex'},dotted,red] (t03.center) -- (14.center);
\draw[arrows={- latex'},dotted,red] (t02.center) -- (13.center);
\draw[arrows={ - latex'},dotted,red] (t01.center) -- (12.center);
\draw[arrows={- latex'},dotted,red] (t00.center) -- (11.center);

\draw[arrows={- latex'},thick] (4,1.5) -- (5,1.5);

\node at (0 + 3*\D+.23, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0+3*\D,0) rectangle (\X*1+3*\D,\X*1) node[fitting node] (b00) {};
\draw (0+3*\D, \X*1) rectangle (\X*1+3*\D, \X*2) node[fitting node] (b01) {};
\draw (0+3*\D, \X*2) rectangle (\X*1+3*\D, \X*3) node[fitting node] (b02) {};
\draw (0+3*\D, \X*3) rectangle (\X*1+3*\D, \X* 4) node[fitting node] (b03) {};

\node at (b00.center) {$\scriptstyle{N}$};
\node at (b01.center) {$\scriptstyle{N}$};
\node at (b02.center) {$\scriptstyle{N}$};
\node at (b03.center) {$\scriptstyle{L}$};

\node at (2*\X+3*\D-.23, \X*6) {$\scriptscriptstyle{B_{0,1}}$};
\draw (\X*1+3*\D,\X*1) rectangle (\X*2+3*\D,\X*2) node[fitting node] (b11) {};
\draw (\X*1+3*\D,\X*2) rectangle (\X*2+3*\D,\X*3) node[fitting node] (b12) {};
\draw (\X*1+3*\D,\X*3) rectangle (\X*2+3*\D,\X*4) node[fitting node] (b13) {};
\draw (\X*1+3*\D,\X*0) rectangle (\X*2+3*\D,\X*1) node[fitting node] (b14) {};

\node at (b11.center) {$\scriptstyle{N^+}$};
\node at (b12.center) {$\scriptstyle{N^+}$};
\node at (b13.center) {$\scriptstyle{N^+}$};
\node at (b14.center) {$\scriptstyle{N^+}$};

\node at (0+3.25*\D, -1*\X) {$\scriptstyle{(rT)}$};

\node at (8.5, 1.5) {and};

\node at (0 + 5*\D+.23, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0+5*\D,0) rectangle (\X*1+5*\D,\X*1) node[fitting node] (g00) {};
\draw (0+5*\D, \X*1) rectangle (\X*1+5*\D, \X*2) node[fitting node] (g01) {};
\draw (0+5*\D, \X*2) rectangle (\X*1+5*\D, \X*3) node[fitting node] (g02) {};
\draw (0+5*\D, \X*3) rectangle (\X*1+5*\D, \X* 4) node[fitting node] (g03) {};

\node at (g00.center) {$\scriptstyle{N}$};
\node at (g01.center) {$\scriptstyle{N}$};
\node at (g02.center) {$\scriptstyle{N}$};
\node at (g03.center) {$\scriptstyle{L}$};

\node at (2*\X+5*\D-.23, \X*6) {$\scriptscriptstyle{B_{1,0}}$};
\draw (\X*1+5*\D,\X*1) rectangle (\X*2+5*\D,\X*2) node[fitting node] (g11) {};
\draw (\X*1+5*\D,\X*2) rectangle (\X*2+5*\D,\X*3) node[fitting node] (g12) {};
\draw (\X*1+5*\D,\X*3) rectangle (\X*2+5*\D,\X*4) node[fitting node] (g13) {};
\draw (\X*1+5*\D,\X*0) rectangle (\X*2+5*\D,\X*1) node[fitting node] (g14) {};

\node at (g11.center) {$\scriptstyle{N^-}$};
\node at (g12.center) {$\scriptstyle{N^-}$};
\node at (g13.center) {$\scriptstyle{N^-}$};
\node at (g14.center) {$\scriptstyle{N^-}$};

\node at (0+5.25*\D, -1*\X) {$\scriptstyle{(rC)}$};
\end{tikzpicture}
\end{center}
\end{defn}

\begin{remark} \label{automaticComp}
An $L^-$-strip always satisfies $(rT)$ and $(lC)$ and an $L^+$-strip always satisfies $(rC)$ and $(lT)$. This can be seen by composing the arrows of the $L$-strip with the arrows within the KR-tableaux. For example, 

\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptstyle{C_{0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (1.5*\X, \X*6) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};

\node at (0.5*\X+\D, \X*6) {$\scriptstyle{T_{0}}$};
\draw[dotted] (0+\D, \X*4) rectangle (\X*1+\D, \X*5) node[fitting node] (t04) {};

\draw (0+\D,0) rectangle (\X*1+\D,\X*1) node[fitting node] (t00) {};
\draw (0+\D, \X*1) rectangle (\X*1+\D, \X*2) node[fitting node] (t01) {};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};

\node at (1.5*\X+\D, \X*6) {$\scriptstyle{T_1}$};
\draw (\X*1+\D,\X*1) rectangle (\X*2+\D,\X*2) node[fitting node] (t11) {};
\draw (\X*1+\D,\X*2) rectangle (\X*2+\D,\X*3) node[fitting node] (t12) {};
\draw (\X*1+\D,\X*3) rectangle (\X*2+\D,\X*4) node[fitting node] (t13) {};
\draw (\X*1+\D,\X*4) rectangle (\X*2+\D,\X*5) node[fitting node] (t14) {};

\draw[arrows={- latex'},red] (t04.center) -- (03.center);
\draw[arrows={- latex'},red] (03.center) -- (t03.center);
\draw[arrows={- latex'},dotted,color=red] (02.center) -- (t03.center);
\draw[arrows={- latex'},dotted,color=red] (01.center) -- (t02.center);
\draw[arrows={- latex'},dotted,color=red] (00.center) -- (t01.center);

\draw[arrows={- latex},dotted,color=blue] (t03.center) -- (t14.center);
\draw[arrows={- latex'},dotted,color=blue] (t02.center) -- (t13.center);
\draw[arrows={- latex'},dotted,color=blue] (t01.center) -- (t12.center);

\draw[arrows={- latex'},blue] (t14.center) -- (t13.center);
\draw[arrows={- latex'},blue] (t13.center) -- (t12.center);

\draw[arrows={-triangle 45},thick] (4,1.5) -- (5,1.5);

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\node at (0.5*\X+3*\D, \X*6) {$\scriptstyle{C_{0}}$};
\draw (0+3*\D,0) rectangle (\X*1+3*\D,\X*1) node[fitting node] (c00) {};
\draw (0+3*\D, \X*1) rectangle (\X*1+3*\D, \X*2) node[fitting node] (c01) {};
\draw (0+3*\D, \X*2) rectangle (\X*1+3*\D, \X*3) node[fitting node] (c02) {};
\draw (0+3*\D, \X*3) rectangle (\X*1+3*\D, \X* 4) node[fitting node] (c03) {};

\node at (1.5*\X+3*\D, \X*6) {$\scriptstyle{C_1}$};
\draw (\X*1+3*\D,\X*1) rectangle (\X*2+3*\D,\X*2) node[fitting node] (c11) {};
\draw (\X*1+3*\D,\X*2) rectangle (\X*2+3*\D,\X*3) node[fitting node] (c12) {};
\draw (\X*1+3*\D,\X*3) rectangle (\X*2+3*\D,\X*4) node[fitting node] (c13) {};
\draw (\X*1+3*\D,\X*4) rectangle (\X*2+3*\D,\X*5) node[fitting node] (c14) {};

\node at (0.5*\X+4*\D, \X*6) {$\scriptstyle{T_{0}}$};
\draw[white] (0+4*\D, \X*4) rectangle (\X*1+4*\D, \X*5) node[fitting node] (tt04) {};

\draw (0+4*\D,0) rectangle (\X*1+4*\D,\X*1) node[fitting node] (tt00) {};
\draw (0+4*\D, \X*1) rectangle (\X*1+4*\D, \X*2) node[fitting node] (tt01) {};
\draw (0+4*\D, \X*2) rectangle (\X*1+4*\D, \X*3) node[fitting node] (tt02) {};
\draw (0+4*\D, \X*3) rectangle (\X*1+4*\D, \X* 4) node[fitting node] (tt03) {};

\node at (1.5*\X+4*\D, \X*6) {$\scriptstyle{T_1}$};
\draw (\X*1+4*\D,\X*1) rectangle (\X*2+4*\D,\X*2) node[fitting node] (tt11) {};
\draw (\X*1+4*\D,\X*2) rectangle (\X*2+4*\D,\X*3) node[fitting node] (tt12) {};
\draw (\X*1+4*\D,\X*3) rectangle (\X*2+4*\D,\X*4) node[fitting node] (tt13) {};
\draw (\X*1+4*\D,\X*4) rectangle (\X*2+4*\D,\X*5) node[fitting node] (tt14) {};

\draw[arrows={- latex'},dotted] (c03.center) -- (tt14.center);
\draw[arrows={- latex'},dotted] (c02.center) -- (tt13.center);
\draw[arrows={- latex'},dotted] (c01.center) -- (tt12.center);
\draw[arrows={- latex'},dotted] (c00.center) -- (tt11.center);
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\draw[arrows={-triangle 45},thick] (10,1.5) -- (11,1.5);

\node at (0 + 6*\D+.23, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0+6*\D,0) rectangle (\X*1+6*\D,\X*1) node[fitting node] (g00) {};
\draw (0+6*\D, \X*1) rectangle (\X*1+6*\D, \X*2) node[fitting node] (g01) {};
\draw (0+6*\D, \X*2) rectangle (\X*1+6*\D, \X*3) node[fitting node] (g02) {};
\draw (0+6*\D, \X*3) rectangle (\X*1+6*\D, \X* 4) node[fitting node] (g03) {};

\node at (g00.center) {$\scriptstyle{N^+}$};
\node at (g01.center) {$\scriptstyle{N^+}$};
\node at (g02.center) {$\scriptstyle{N^+}$};
\node at (g03.center) {$\scriptstyle{L^-}$};

\node at (2*\X+6*\D-.23, \X*6) {$\scriptscriptstyle{B_{0,1}}$};
\draw (\X*1+6*\D,\X*1) rectangle (\X*2+6*\D,\X*2) node[fitting node] (g11) {};
\draw (\X*1+6*\D,\X*2) rectangle (\X*2+6*\D,\X*3) node[fitting node] (g12) {};
\draw (\X*1+6*\D,\X*3) rectangle (\X*2+6*\D,\X*4) node[fitting node] (g13) {};
\draw (\X*1+6*\D,\X*0) rectangle (\X*2+6*\D,\X*1) node[fitting node] (g14) {};

\node at (g11.center) {$\scriptstyle{N^+}$};
\node at (g12.center) {$\scriptstyle{N^+}$};
\node at (g13.center) {$\scriptstyle{N^+}$};
\node at (g14.center) {$\scriptstyle{N^+}$};
\end{tikzpicture}
\end{center}

By composing the red and blue arrows, we see that the condition $(rT)$ is always satisfied. 
\end{remark}

\begin{defn}
Let $(C,T)$ be a pair of KR-tableaux and let $S=(S_1,\ldots, S_k)$, where $S_l$ is a union of one or more $L$-strips in $(C_l,T_l)$. We say $S$ is \emph{exchangeable} if it is possible to exchange the boxes in $S$ and still obtain valid KR-tableaux, i.e. the resulting new pair $(\tilde{C},\tilde{T})$ have strictly increasing columns and weakly increasing diagonals. $S$ is \emph{minimally exchangeable} if removing any nonempty subset of $L$-strips from $S$ results in a non-exchangeable sequence. 
\end{defn}

\begin{example} Consider $C=(C_1,C_2,C_3,C_4)$ and $T=(T_1,T_2,T_3)$. 

\begin{center}
\begin{tikzpicture}
\node at (0,0) {$
\ytableausetup
{mathmode, boxsize=1em}
\begin{ytableau}
\none[\scriptstyle   ]&	\none[ \scriptstyle C_1]&\none[\scriptstyle C_2]&\none[\scriptstyle C_3]&\none[\scriptstyle C_4]&\none&\none&	\none[\scriptstyle T_1]&\none[\scriptstyle T_2]&\none[\scriptstyle T_3]\\
\none[\scriptstyle -4]&\none&\none&\none&\none&\none&\none&\none &\none&\scriptstyle3	\\
\none[\scriptstyle -3]&\none&\none&\none&\scriptstyle6	   &\none&\none&\none&\scriptstyle3       &\scriptstyle4\\
\none[\scriptstyle -2]&\none&\none&\scriptstyle5	&\scriptstyle8	   &\none&\none&\scriptstyle2       &\scriptstyle4       &\scriptstyle6\\
\none[\scriptstyle -1]&\none&\scriptstyle5	     &\scriptstyle7	&\scriptstyle9 	   &\none&\none	    &\scriptstyle4       &\scriptstyle6       &\scriptstyle7\\
\none[\scriptstyle 0]&\scriptstyle5	  &\scriptstyle6	     &*(yellow)\scriptstyle\mathbf{9}	&\scriptstyle10     &\none&\none	    &\scriptstyle6       &\scriptstyle7       &*(yellow)\scriptstyle\mathbf{8}\\
\none[\scriptstyle 1]&\scriptstyle6	  &*(yellow)\scriptstyle\mathbf{9}	     &*(yellow)\scriptstyle\mathbf{10}	&\none&\none&\none    &\scriptstyle7       &*(yellow)\scriptstyle\mathbf{8}       &*(yellow)\scriptstyle\mathbf{9}\\
\none[\scriptstyle 2]&\scriptstyle8	  &*(yellow)\scriptstyle\mathbf{10}	     &\none&\none&\none&\none	    &\scriptstyle8       &*(yellow)\scriptstyle\mathbf{9}       &\none\\
\none[\scriptstyle 3]&*(yellow)\scriptstyle\mathbf{10}	  &\none&\none&\none&\none&\none&*(yellow)\scriptstyle\mathbf{9}	       &\none&\none\\
\none[\scriptstyle 4]&	\none &\none&\none&\none&\none&\none&\none&\none&\none\\
\end{ytableau}
$};

\draw[->] (3.5,0) -- (6.5,0);
\node at (5,0)[above] {exchange};

\node at (10,0) {$
\begin{ytableau}
\none[\scriptstyle   ]&	\none[ \scriptstyle \tilde{C}_1]&\none[\scriptstyle \tilde{C}_2]&\none[\scriptstyle \tilde{C}_3]&\none[\scriptstyle \tilde{C}_4]&\none&\none&	\none[\scriptstyle \tilde{T}_1]&\none[\scriptstyle \tilde{T}_2]&\none[\scriptstyle \tilde{T}_3]\\
\none[\scriptstyle -4]&\none&\none&\none&\none&\none&\none&\none &\none&\scriptstyle3	\\
\none[\scriptstyle -3]&\none&\none&\none&\scriptstyle6	   &\none&\none&\none&\scriptstyle3       &\scriptstyle4\\
\none[\scriptstyle -2]&\none&\none&\scriptstyle5	&\scriptstyle8	   &\none&\none&\scriptstyle2       &\scriptstyle4       &\scriptstyle6\\
\none[\scriptstyle -1]&\none&\scriptstyle5	     &\scriptstyle7	&\scriptstyle9 	   &\none&\none    &\scriptstyle4       &\scriptstyle6       &\scriptstyle7\\
\none[\scriptstyle 0]&\scriptstyle5	  &\scriptstyle6	     &*(yellow)\scriptstyle\mathbf{8}	&\scriptstyle10     &\none&\none   &\scriptstyle6       &\scriptstyle7       &*(yellow)\scriptstyle\mathbf{9}\\
\none[\scriptstyle 1]&\scriptstyle6	  &*(yellow)\scriptstyle\mathbf{8}	     &*(yellow)\scriptstyle\mathbf{9}&\none&\none&\none	    &\scriptstyle7       &*(yellow)\scriptstyle\mathbf{9}       &*(yellow)\scriptstyle\mathbf{10}\\
\none[\scriptstyle 2]&\scriptstyle8	  &*(yellow)\scriptstyle\mathbf{9}	     &\none&\none&\none&\none    &\scriptstyle8       &*(yellow)\scriptstyle\mathbf{10}       &\none\\
\none[\scriptstyle 3]&*(yellow)\scriptstyle\mathbf{9}	  &\none&\none&\none&\none&\none	    &*(yellow)\scriptstyle\mathbf{10}	       &\none&\none\\
\none[\scriptstyle 4]&	\none &\none&\none&\none&\none&\none    &\none&\none&\none\\
\end{ytableau}
$};
\end{tikzpicture}
\end{center}

A sequence of $L$-strips is shown in bold letters above given by $S_1 = (C_1[3],T_1[3])$, $S_2 = (C_2[1,2],T_2[1,2])$, and $S_3 = (C_3[0,1],T_3[0,1])$. Notice that each $L$-strip is column-com\-pa\-tible, $S_1$ is left-compatible, $S_3$ is right-com\-pa\-tible, and $S_1,S_2$ are not right-com\-pa\-tible. The sequence $S$ is exchangeable since it is possible to exchange the colored boxes and still obtain a pair of valid KR-tableaux. It is also minimal since removing any subset results in a non-exchangeable sequence. For example, exchanging boxes of $S_1,S_2$ without exchanging the boxes of $S_3$ will violate the weakly increasing diagonals condition since $S_2$ is not right-compatible. 
\end{example}

\begin{defn}
Given a left-compatible $L$-strip, we say \emph{it can be completed} to an exchangeable sequence if it is possible to include boxes that, when not exchanged, cause violations of the left or right compatibility conditions and achieve an exchangeable sequence. If, in the process, we end up with an irresolvable contradiction, we say the $L$-strip \emph{cannot be completed to an exchangeable sequence}. 
\end{defn}
Let us demonstrate some situations where an $L$-strip cannot be completed to an exchangeable sequence. 
\begin{example}
In both cases demonstrated below, there is an $L$-strip in $(C_1,T_1)$ (bold), which is not right-compatible. We iteratively add all the boxes that cause right or column compatibility violations. In both cases, we run out of boxes to include before resolving all the violations. 
\begin{center}
\begin{tikzpicture}
\node at (0,0) {$
\begin{ytableau}
\none[\scriptstyle   ]&	\none[ \scriptstyle C_{1}]&\none[\scriptstyle C_2]&\none[\scriptstyle C_3]&\none[\scriptstyle C_4]&\none&\none&	\none[\scriptstyle T_1]&\none[\scriptstyle T_2]&\none[\scriptstyle T_3]&\none[\scriptstyle T_4]\\
\none[\scriptstyle -4]&\none&\none&\none&\none&\none&\none&\none &\none&\none&\scriptstyle4	\\
\none[\scriptstyle -3]&\none&\none&\none&\scriptstyle4	   &\none&\none&\none&\none&2       &\scriptstyle5\\
\none[\scriptstyle -2]&\none&\none&*(yellow)\scriptstyle\mathbf{4}	&\scriptstyle6	   &\none&\none&\none&\scriptstyle2       &*(yellow)\scriptstyle\mathbf{3}       &\scriptstyle7\\
\none[\scriptstyle -1]&\none&*(yellow)\scriptstyle\mathbf{4}	     &\scriptstyle6	&\scriptstyle7 	   &\none&\none&\scriptstyle2	    &*(yellow)\scriptstyle\mathbf{3}       &\scriptstyle7       &\scriptstyle8\\
\none[\scriptstyle 0]&*(yellow)\scriptstyle\mathbf{4}	  &*(yellow)\scriptstyle\mathbf{6}	     &\scriptstyle7	&\scriptstyle8     &\none&\none&*(yellow)\scriptstyle\mathbf{3}	    &*(yellow)\scriptstyle\mathbf{4}       &\scriptstyle8       &\scriptstyle9\\
\none[\scriptstyle 1]&*(yellow)\scriptstyle\mathbf{5}	  &*(yellow)\scriptstyle\mathbf{7}	     &\scriptstyle8	&\none&\none&\none&*(yellow)\scriptstyle\mathbf{4}	    &*(yellow)\scriptstyle\mathbf{6}       &\scriptstyle9       &\scriptstyle10\\
\none[\scriptstyle 2]&\scriptstyle6  &*(yellow)\scriptstyle\mathbf{8}     &\none&\none&\none&\none&\scriptstyle6	    &*(yellow)\scriptstyle\mathbf{7}       &\scriptstyle10       &\none\\
\none[\scriptstyle 3]&\scriptstyle7	  &\none&\none&\none&\none&\none&\scriptstyle7	    &*(red)\scriptstyle\mathbf{8}	       &\none&\none\\
\none[\scriptstyle 4]&	\none &\none&\none&\none&\none&\none&\scriptstyle8	    &\none&\none&\none\\
\end{ytableau}
$};

\node at (0,-3) {$(C_2,T_2)$ is not column-compatible};

\node at (8,0) {$
\begin{ytableau}
\none[\scriptstyle   ]&	\none[ \scriptstyle C_1]&\none[\scriptstyle C_2]&\none[\scriptstyle C_3]&\none[\scriptstyle C_4]&\none&\none&	\none[\scriptstyle T_1]&\none[\scriptstyle T_2]&\none[\scriptstyle T_3]&\none[\scriptstyle T_4]\\
\none[\scriptstyle -4]&\none&\none&\none&\none&\none&\none&\none &\none&\none&\scriptstyle4	\\
\none[\scriptstyle -3]&\none&\none&\none&\scriptstyle4	   &\none&\none&\none&\none&*(red)\scriptstyle\mathbf{4}       &\scriptstyle5\\
\none[\scriptstyle -2]&\none&\none&*(yellow)\scriptstyle\mathbf{4}	&\scriptstyle6	   &\none&\none&\none&\scriptstyle3       &*(yellow)\scriptstyle\mathbf{5}       &\scriptstyle7\\
\none[\scriptstyle -1]&\none&*(yellow)\scriptstyle\mathbf{4}	     &\scriptstyle6	&\scriptstyle7 	   &\none&\none&\scriptstyle3	    &*(yellow)\scriptstyle\mathbf{5}       &\scriptstyle7       &\scriptstyle8\\
\none[\scriptstyle 0]&*(yellow)\scriptstyle\mathbf{4}	  &\scriptstyle6	     &\scriptstyle7	&\scriptstyle8     &\none&\none&*(yellow)\scriptstyle\mathbf{5}	    &\scriptstyle6       &\scriptstyle8       &\scriptstyle9\\
\none[\scriptstyle 1]&*(yellow)\scriptstyle\mathbf{5}	  &\scriptstyle7	     &\scriptstyle8	&\none&\none&\none&*(yellow)\scriptstyle\mathbf{6}	    &\scriptstyle7       &\scriptstyle9       &\scriptstyle10\\
\none[\scriptstyle 2]&	*(yellow)\scriptstyle\mathbf{6}  &\scriptstyle8     &\none&\none&\none&\none&*(yellow)\scriptstyle\mathbf{7}	    &\scriptstyle8       &\scriptstyle10       &\none\\
\none[\scriptstyle 3]&*(yellow)\scriptstyle\mathbf{7}	  &\none&\none&\none&\none&\none&*(yellow)\scriptstyle\mathbf{8}	    &\scriptstyle10	       &\none&\none\\
\none[\scriptstyle 4]&	\none &\none&\none&\none&\none&\none&\scriptstyle10	    &\none&\none&\none\\
\end{ytableau}
$};
\node at (8.5,-3) {$(C_3,T_3)$ is not column-compatible};
\end{tikzpicture}
\end{center}

\end{example}

\begin{defn}
Let $(C,T)$ be a pair of KR-tableaux such that no $L$-strip can be completed to an exchangeable sequence. Then we say $(C,T)$ \emph{has no exchangeable sequences}. 
\end{defn}

We now describe the map $\sigma$. Consider the set $\mathcal{B}_{k,j}^{(i)} \times \mathcal{B}_{k',j'}^{(i')}$ for any $(i,j,k)$ and $(i',j',k')$. We define 
\begin{eqnarray*}
\mathcal{P}_0 := \left\{ (C,T) \big| C\in \mathcal{B}_{k,j}^{(i)}, T\in \mathcal{B}_{k',j'}^{(i')}, \text{ and } (C,T) \text{ has no exchangeable sequences}\right\}
\end{eqnarray*}
Let $(C,T) \in \mathcal{B}_{k,j}^{(i)} \times \mathcal{B}_{k',j'}^{(i')} \backslash \mathcal{P}_0$. By definition, $(C,T)$ has an exchangeable sequence. Suppose $(C',T') \in \mathcal{B}_{k,j}^{(i)} \times \mathcal{B}_{k',j'}^{(i')} \backslash \mathcal{P}_0$ such that $(C',T')$ is obtained from $(C,T)$ by exchanging a minimally exchangeable sequence. Then we assign $(C,T)$ to $\mathcal{P}_1$, $(\tilde{C},\tilde{T})$ to $\mathcal{P}_{-1}$, and define $\sigma(C,T) = (\tilde{C},\tilde{T})$. The exact order is not important. All that matters is that we can partition $\mathcal{B}_{k,j}^{(i)} \times \mathcal{B}_{k',j'}^{(i')} \backslash \mathcal{P}_0$ into $2$ disjoint subsets. Since the elements in $\mathcal{P}_0$ do not have exchangeable sequences, we define $\sigma(C,T) = (C,T)$ for all $(C,T)\in \mathcal{P}_0$. 

\subsection{Proof of Theorem \ref{thm:main1}} \label{proofOfMain1}
Let $C=(C_i)$ and $T=(T_j)$ be a pair of KR-tableaux with no exchangeable sequences. We want to show $\gamma(C,T)=0$. Recall from Remark \ref{rmk:gammaAdditive} that:
\begin{eqnarray*}
\gamma(C,T) = \sum_{i,j}\gamma(C_i,T_j)\,\, ,
\end{eqnarray*} 
and by Lemma \ref{gamma}:
\begin{eqnarray*}
\gamma(C_i,T_j) = \sum_{p=h}^{t} L_p (C_i,T_j) + \left\{ \begin{array}{cc} 
N_{t+1}(C_i,T_j) &\text{ if } (C_i,T_j) \text{ is (anti-)fundamental}\\
L_{t+1}(C_i,T_j) &\text{ otherwise}
\end{array}\right.  \,\, ,
\end{eqnarray*} 
where $h=\head(C,T)$ and $t=\tail(C,T)$. 

Let us prove first the theorem in case when both $C$ and $T$ are column tableaux. Recall that column diagrams correspond to fundamental modules ($k=1$) and the fundamental modules in the fundamental cluster $\mathcal{C}$ form (anti-)fundamental pairs (see Section \ref{fundClusterDiagrams} and Definition \ref{def:pairs}).

\begin{lemma} \label{gammaContr}
Let $(C,T)$ be a pair of column KR-tableaux of type (anti-)fundamental with no exchangeable sequences, i.e. $(C,T)\in\mathcal{P}_0$. Then $\gamma(C,T)=0$. 
\end{lemma}
\begin{proof}
Suppose $(C,T)$ admits an $L$-strip. Notice that an $L$-strip in $(C,T)$ is exchangeable if and only if it is column-compatible since there are no columns to the left or to the right of both $C$ and $T$. Since $(C,T)$ has no exchangeable sequences, there can be exactly one $L$-strip. Otherwise, by Remark \ref{cross-pattern}, the $L$-strips other than the last one are all column-compatible, and, therefore, exchangeable. 

We may assume the $L$-strip in question is an $L^-$-strip. If not, we simply consider $(T,C)$ instead, where $L^+$-strips in $(C,T)$ become $L^-$-strips in $(T,C)$. Let $p^*$ be the index of the head of the $L$-strip. Then,
\begin{eqnarray*}
\gamma(C,T) = \underbrace{L_{p^*}(C,T)}_\text{-1} + N_{t+1}(C,T)\,\, ,
\end{eqnarray*}
where $t=\tail(C,T)$. Since $L_{p^*}$ is not column-compatible, it is not followed by an $LU$-block. That is, it must be followed by $N$-blocks only. By Remark \ref{rmk:obs1}(a,b), an $L^-$-block can only be followed by an $N^+$, and $N^+$-blocks cannot be followed by $N^-$-blocks. Therefore, we must have that $N_{t+1}(C,T) = +1$ and $\gamma(C,T) = 0$. 

Suppose $(C,T)$ has no $L$-strips. We can assume $(C,T)$ is a fundamental pair. If not, we simply consider $(T,C)$ instead. If the boundary term is zero, that is $N_{t+1}(C,T)=0$, then $\gamma(C,T)=0$ and the result holds. Suppose the boundary term is not zero. By Lemma \ref{rmk:obs2}(b), since $(C,T)$ is a fundamental pair, the last block is never $N^+$. Therefore, $N_{t+1}(C,T)=-1$. By Remark \ref{rmk:obs1}(a,b), an $N^-$-block can only be preceded by $L^+$ or $N^-$. However, there are no $L$-blocks in $(C,T)$. So, there must be only $N^-$-blocks in $(C,T)$. By Lemma \ref{rmk:obs2}(a), the very first block cannot be $N^-$. Contradiction. This concludes the proof. 
\end{proof}

\begin{cor}
Non-column-compatible $L$-strips in (anti-)fundamental pairs do not contribute to $\gamma$ since their contribution is always canceled out by the boundary term. 
\end{cor}
Our strategy in proving Theorem \ref{thm:main1} is to find a way to systematically cancel contributions from $L_p$ to $\gamma$ when the sequence cannot be completed to an exchangeable sequence. In the lemmas that follow, we always have $C=(C_0,C_1)$ and $T=(T_0,T_1)$ a pair of KR-tableaux and $(C_0,T_0)$ has a non-exchangeable $L^-$-strip. The goal is to find a unique non-exchangeable $L^+$-strip in $(C,T)$ to cancel out the contribution of $L^-$. Moreover, when the $L^-$ does not contribute to $\gamma$ (as in Lemma \ref{gammaContr}), we want to show that there is no such corresponding $L^+$. We make an exhaustive list of all the possibilities for $L^-$ to be non-exchangeable and give the lemma that address the situation as a reference in Table \ref{tab:cancellations}.

\begin{table}[ht]
\begin{tabular}{ |c|c|c|c|c|c|} 
\hline
$(C_0,T_0)$ type& \begin{tabular}{@{}c@{}}$L^-$ in $(C_0,T_0)$ is\\right \\com\-pa\-tible\end{tabular} &  \begin{tabular}{@{}c@{}}$L^-$ in $(C_0,T_0)$ is\\column\\com\-pa\-tible\end{tabular}&Contributes to $\gamma$&Lemma\\
 \hline\hline
any type & No & Yes &Yes& \ref{chain1}\\ 
 \hline
 (anti-)regular & Yes & No &Yes& \ref{chain2},\ref{chain3}\\ 
 \hline
 (anti-)fundamental & No & No &No& \ref{chain4}\\ 
 \hline
 (anti-)fundamental & Yes & No &No& \ref{chain5}\\ 
 \hline
 \end{tabular}
 \caption{List of non-exchangeable $L^-$-strips}
 \label{tab:cancellations}
\end{table}

\begin{example} \label{ex:chain1}
We demonstrate the most common type of cancellation in Figure~\ref{fig:cancel}. There is $L^-$-block in $(C_0,T_0)$ that is column-compatible, but not right-compatible given by $1<2<3$ (colored, left), which contributes $-1$ to $\gamma$. It is not right-compatible due to $2$ in $C_1$ since the weakly increasing diagonals condition will be violated post-exchange of the $L^-$-strip. However, there is a non-left-compatible $L^+$-block in $(C_1,T_0)$ given by $2<3<4$ (colored, right). The $L^+$-strip is not left-compatible due to $4$ in $C_0$. That is, we found a non-exchangeable $+1$ contribution to $\gamma$ to cancel out the previous $-1$. 

\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptstyle{C_{0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw[fill=yellow] (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (03.center) {$\scriptstyle 1$};
\node at (02.center) {$\scriptstyle 2$};
\node at (01.center) {$\scriptstyle 4$};
\node at (00.center) {$\scriptstyle 5$};

\node at (1.5*\X, \X*6) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw[fill=red] (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};

\node at (14.center) {$\scriptstyle 1$};
\node at (13.center) {$\scriptstyle 2$};
\node at (12.center) {$\scriptstyle 4$};
\node at (11.center) {$\scriptstyle 5$};

\node at (0.5*\X+\D, \X*6) {$\scriptstyle{T_{0}}$};
\draw[white] (0+\D, \X*4) rectangle (\X*1+\D, \X*5) node[fitting node] (t04) {};

\draw (0+\D,0) rectangle (\X*1+\D,\X*1) node[fitting node] (t00) {};
\draw (0+\D, \X*1) rectangle (\X*1+\D, \X*2) node[fitting node] (t01) {};
\draw[fill=yellow] (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};

\node at (t03.center) {$\scriptstyle 1$};
\node at (t02.center) {$\scriptstyle 3$};
\node at (t01.center) {$\scriptstyle 5$};
\node at (t00.center) {$\scriptstyle 6$};

\draw[arrows={- latex'},red] (t03.west) -- (02.east);
\draw[arrows={- latex'},red] (02.east) -- (t02.west);

\draw[arrows={-triangle 45},thick] (4,1.5) -- (5,1.5);

\node at (0.5*\X+3*\D, \X*6) {$\scriptstyle{C_{0}}$};
\draw (0+3*\D,0) rectangle (\X*1+3*\D,\X*1) node[fitting node] (c00) {};
\draw[fill=red] (0+3*\D, \X*1) rectangle (\X*1+3*\D, \X*2) node[fitting node] (c01) {};
\draw (0+3*\D, \X*2) rectangle (\X*1+3*\D, \X*3) node[fitting node] (c02) {};
\draw (0+3*\D, \X*3) rectangle (\X*1+3*\D, \X* 4) node[fitting node] (c03) {};

\node at (c03.center) {$\scriptstyle 1$};
\node at (c02.center) {$\scriptstyle 2$};
\node at (c01.center) {$\scriptstyle 4$};
\node at (c00.center) {$\scriptstyle 5$};

\node at (1.5*\X+3*\D, \X*6) {$\scriptstyle{C_1}$};
\draw (\X*1+3*\D,\X*1) rectangle (\X*2+3*\D,\X*2) node[fitting node] (c11) {};
\draw[fill=yellow] (\X*1+3*\D,\X*2) rectangle (\X*2+3*\D,\X*3) node[fitting node] (c12) {};
\draw (\X*1+3*\D,\X*3) rectangle (\X*2+3*\D,\X*4) node[fitting node] (c13) {};
\draw (\X*1+3*\D,\X*4) rectangle (\X*2+3*\D,\X*5) node[fitting node] (c14) {};

\node at (c14.center) {$\scriptstyle 1$};
\node at (c13.center) {$\scriptstyle 2$};
\node at (c12.center) {$\scriptstyle 4$};
\node at (c11.center) {$\scriptstyle 5$};

\node at (0.5*\X+4*\D, \X*6) {$\scriptstyle{T_{0}}$};
\draw[white] (0+4*\D, \X*4) rectangle (\X*1+4*\D, \X*5) node[fitting node] (tt04) {};

\draw (0+4*\D,0) rectangle (\X*1+4*\D,\X*1) node[fitting node] (tt00) {};
\draw (0+4*\D, \X*1) rectangle (\X*1+4*\D, \X*2) node[fitting node] (tt01) {};
\draw[fill=yellow] (0+4*\D, \X*2) rectangle (\X*1+4*\D, \X*3) node[fitting node] (tt02) {};
\draw (0+4*\D, \X*3) rectangle (\X*1+4*\D, \X* 4) node[fitting node] (tt03) {};

\node at (tt03.center) {$\scriptstyle 1$};
\node at (tt02.center) {$\scriptstyle 3$};
\node at (tt01.center) {$\scriptstyle 5$};
\node at (tt00.center) {$\scriptstyle 6$};

\draw[arrows={- latex'},red] (c13.east) -- (tt02.west);
\draw[arrows={- latex'},red] (tt02.west) -- (c12.east);
\end{tikzpicture}
\end{center}
\caption{Cancellation of non-right-compatible $L^-$-strips with non-left-compatible $L^+$-strips}
\label{fig:cancel}
\end{figure}

\end{example}

\begin{remark}\label{rmk:obs3}

Suppose $C=(C_0,C_1)$ and $T_0$ are KR-tableaux. Suppose there is an $LU$-block at $p$ in $(C_0,T_0)$. Then there is an $L^+N^-$-block at $p-1$ in $(C_1,T_0)$ (see Figure~\ref{fig:obs3}). Indeed, 
\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

\node at (0,0) {
\begin{tikzpicture}
\node at (0.5*\X, \X*5) {$\scriptstyle{C_0}$};

\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at ($(02) +(-.6,0)$) {$\scriptstyle p$};
\node at ($(03) +(-.6,0)$) {$\scriptstyle p-1$};

\node at (\X*1.5, \X*5) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};

\node at (0.5*\X+\D, \X*5) {$\scriptstyle{T_0}$};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};

\node at (0.5*\X+\D+1,\X*5) {$\scriptstyle{B_{0,0}}$};
\draw (0+\D+1, \X*2) rectangle (\X*1+\D+1, \X*3) node[fitting node] (b02) {};
\draw (0+\D+1, \X*3) rectangle (\X*1+\D+1, \X* 4) node[fitting node] (b03) {};

\node at (0.5*\X+\D+1.8,\X*5) {$\scriptstyle{B_{1,0}}$};
\draw (\X*1+\D+1, \X*3) rectangle (\X*1+\D+2, \X* 4) node[fitting node] (b13) {};

\node at (b02.center) {$\scriptscriptstyle L^-$};
\node at (b13.center) {$\scriptscriptstyle L^+N^-$};

\draw[arrows={ - latex'},dotted,blue] (02.center) -- (13.center);
\draw[arrows={ - latex'},blue] (t03.center) -- (02.center);
\draw[arrows={ - latex'},red,thick] (t03.center) -- (13.center);

\end{tikzpicture}
};

\node at (6,0) {
\begin{tikzpicture}
\node at (0.5*\X, \X*5) {$\scriptstyle{C_0}$};

\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at ($(02) +(-.6,0)$) {$\scriptstyle p$};
\node at ($(03) +(-.6,0)$) {$\scriptstyle p-1$};

\node at (\X*1.5, \X*5) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};

\node at (0.5*\X+\D, \X*5) {$\scriptstyle{T_0}$};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};

\node at (0.5*\X+\D+1,\X*5) {$\scriptstyle{B_{0,0}}$};
\draw (0+\D+1, \X*2) rectangle (\X*1+\D+1.1, \X*3) node[fitting node] (b02) {};
\draw (0+\D+1, \X*3) rectangle (\X*1+\D+1.1, \X* 4) node[fitting node] (b03) {};

\node at (0.5*\X+\D+1.8,\X*5) {$\scriptstyle{B_{1,0}}$};
\draw (\X*1+\D+1.1, \X*3) rectangle (\X*1+\D+2, \X* 4) node[fitting node] (b13) {};

\node at (b02.center) {$\scriptscriptstyle L^+U$};
\node at (b13.center) {$\scriptscriptstyle L^+N^-$};

\draw[arrows={ - latex'},dotted,blue] (02.center) -- (13.center);
\draw[arrows={ - latex'},blue,dotted] (t02.center) -- (02.center);
\draw[arrows={ - latex'},blue] (t03.center) -- (t02.center);
\draw[arrows={ - latex'},red,thick] (t03.center) -- (13.center);

\end{tikzpicture}
};
\end{tikzpicture}
\end{center}
\caption{Composition of arrows.}
\label{fig:obs3}
\end{figure}
by composing arrows, we find $T_0[p-1]<C_1[p-1]$, which means there cannot be an $L^-U$-block in $(C_1,T_0)$ at $p-1$. Also, there cannot be an $N^+$-block since $T_0[p-2]<T_0[p-1] < C_1[p-1]$. Thus, there can be either $L^+$ or an $N^-$ block at $p-1$ in $(C_1,T_0)$. 
\end{remark}
This next lemma is the main cancellation action demonstrated in Example \ref{ex:chain1}. 
\begin{lemma} \label{chain1}
Let $C=(C_0,C_1)$ and $T=(T_0,T_1)$ be a pair of KR-tableaux. To every right-incompatible $L^-$-strip in $(C_0,T_0)$ that contributes to $\gamma$, there exists a unique left-incompatible $L^+$-strip in $(C_1,T_0)$ that contributes to $\gamma$. All other $L^+$-strips in $(C_1,T_0)$ are left-compatible. 
\end{lemma}

\begin{proof}
Without loss of generality, we may assume $(C_0,T_0)$ is either fundamental or regular. If not, we simply consider $(T,C)$ which is of the desired type. As before, denote the block tableaux of $(C_i,T_j)$ by $B_{i,j}$. 

We will consider the case of the very last non-right-compatible $L^-$-strip separately at the end of the proof. Let $\tilde{L}^-$ be not the very last one, and let $p'$ be its index. By Lemma \ref{automaticComp}, $\tilde{L}^-$ satisfies the condition $(rT)$, i.e. $T_1$ does not pose violations. Since $\tilde{L}^-$ is not right-compatible, the condition $(rC)$ must fail. In other words, $C_1$ must pose a right-compatibility violation.  Since $\tilde{L}^-$ is not the last $L^-$, it is followed by an $L$-block or a $U$-block (whichever one comes first). Let $p''$ be the index of the $LU$-block that follows $\tilde{L}^-$. 
\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\node at (00.center) {$\scriptscriptstyle{LU}$};
\node at ($(00.center) + (-.5,0)$) {$\scriptscriptstyle p''$};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at (03.center) {$\scriptscriptstyle{\tilde{L}^-}$};
\node at ($(03.center) + (-.5,0)$) {$\scriptscriptstyle p'$}; 
\node at (02.center) {$\scriptscriptstyle{N^+}$};
\node at (01.center) {$\scriptscriptstyle{N^+}$};

\node at (1.5*\X, \X*6) {$\scriptscriptstyle{B_{1,0}}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\node at (11.center) {$\scriptscriptstyle{Q''}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\node at (13.center) {$\scriptscriptstyle{Q'}$};

\node at (\X*1+4*\D, \X*3) {\begin{minipage}{0.6\textwidth}
Denote the blocks in $(C_1,T_0)$ at indices $p'$ and $p''-1$ by $Q'$ and $Q''$ respectively. By Remark \ref{rmk:obs3}, $Q''$ is either $L^+$ or $N^-$. If $Q''=L^+$, we have a candidate. Suppose not, i.e. $Q''=N^-$. Then, by Remark \ref{rmk:obs1} it can only be preceded by $L^+$ or $N^-$.  If we don't allow any $L^+$ between $Q'$ and $Q''$, then Condition $(rC)$ is satisfied. Contradiction. Therefore, there must be at least one $L^+$-block between $Q'$ and $Q''$. \end{minipage}};
\end{tikzpicture}
\end{center}
If there is more than one $L^+$-block, let $\tilde{L}^+$ be the one with the largest index, i.e. closest to $Q''$. Then $\tilde{L}^+$ is not left-compatible due to the $LU$-block in $(C_0,T_0)$. Moreover, all other $L^+$-strips between $Q'$ and $Q''$ are left-compatible since $(lT)$ is satisfied (seen from picture) and $(lC)$ is always satisfied for $L^+$-strips.

It is not clear that the $\tilde{L}^+$ we found contributes to $\gamma$. Suppose it does not. There is exactly one situation where an $L$-block does not contribute to $\gamma$. This can happen only if $(C_1,T_0)$ is fundamental and $\tilde{L}^+$ is the last $L$-block followed by $N^-$'s. The boundary term of $\gamma$ is then $-1$, which cancels out the contribution of $\tilde{L}^+$. Then we have the following picture: 
\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*5) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\node at (00.center) {$\scriptscriptstyle{LU}$};
\node at ($(00.center) + (-.5,0)$) {$\scriptscriptstyle p''$};

\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at (03.center) {$\scriptscriptstyle{\tilde{L}^-}$};
\node at ($(03.center) + (-.5,0)$) {$\scriptscriptstyle p'$};

\node at (02.center) {$\scriptscriptstyle{N^+}$};
\node at (01.center) {$\scriptscriptstyle{N^+}$};

\draw[white] (\X*1,-2*\X) rectangle (\X*2,-1*\X) node[fitting node] (1-2) {};
\node at (1.5*\X, \X*5) {$\scriptscriptstyle{B_{1,0}}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\node at (11.center) {$\scriptscriptstyle{\tilde{L}^+}$};
\draw (\X*1,\X*0) rectangle (\X*2,\X*1) node[fitting node] (10) {};
\node at (10.center) {$\scriptscriptstyle{N^-}$};
\draw (\X*1,-1*\X) rectangle (\X*2,\X*0) node[fitting node] (1-1) {};
\node at (1-1.center) {$\scriptscriptstyle{N^-}$};
\node at (1-2.center) {$\scriptscriptstyle{\vdots}$};

\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};

\node at (\X*1+4*\D, \X*3) {\begin{minipage}{0.6\textwidth}
All $L^-$-strips in $(C_0,T_0)$ appearing at any indices $p\geq p''$ satisfy Condition $(rC)$ since $B_{1,0}$ consists of $N^-$-blocks only for all $p \geq p''$, and, therefore, are right-compatible. This means $\tilde{L}^-$ is the very last non-right-compatible $L^-$-strip, which is a contradiction. This concludes the proof for this case.  
 \end{minipage}};
\end{tikzpicture}
\end{center}
Now let $\tilde{L}^-$  be the very last $L^-$-strip that contributes to $\gamma$ and is not right-compatible. There are exactly two situations: 
\begin{enumerate}
\item The situation described above, i.e. $\tilde{L}^-$ is followed by $LU$, but the corresponding $\tilde{L}^+$ in $(C_1,T_0)$ does not contribute to $\gamma$. This happens when $(C_1,T_0)$ is fundamental. 
\item $\tilde{L}^-$ is not followed by $LU$ and $(C_0,T_0)$ is regular. 
\end{enumerate}
In situation (1), the $\tilde{L}^+$ we found did not contribute to $\gamma$ and it is the last $L^+$ in $(C_1,T_0)$. Therefore, we must look for the appropriate $L^+$ elsewhere in $(C_1,T_0)$. In situation (2), since $\tilde{L}^-$ is the absolute last $L^-$-strip that is not followed by $LU$, it is followed by $N^+$-blocks only. This means that any $L^+$ in $(C_1,T_0)$ adjacent to the strip associated to $\tilde{L}^-$ is left-compatible. Since we are looking for a non-left-compatible $L^+$ to pair with $\tilde{L}^-$, we must also look elsewhere in $(C_1,T_0)$. 

We now consider both situations. Let $\hat{L}^-$ be the very first $L^-$-strip in $(C_0,T_0)$, and let $p'$ be its index. Denote the first block in $(C_1,T_0)$ by $Q_0$ and the block at index $p'-1$ in $(C_1,T_0)$ by $Q_1$. 
\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\node at (00.center) {$\scriptscriptstyle{\hat{L}^-}$};
\node at ($(00.center) + (-.5,0)$) {$\scriptscriptstyle p'$};

\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\node at (01.center) {$\scriptscriptstyle{N^+}$};
\node at (02.center) {$\scriptscriptstyle{N^+}$};

\node at (1.5*\X, \X*6) {$\scriptscriptstyle{B_{1,0}}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\node at (11.center) {$\scriptscriptstyle{Q_1}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\node at (13.center) {$\scriptscriptstyle{Q_{0}}$};

\node at (\X*1+4*\D, \X*3) {\begin{minipage}{0.75\textwidth}
By Lemma \ref{rmk:obs2}, the first block in $B_{1,0}$, i.e. $Q_{0}$, is not $N^-$. By Remark \ref{rmk:obs3}, $Q_1$ is either $L^+$ or $N^-$. The only way to transition from non $N^-$-block to $N^-$-block is through $L^+$. Therefore, there must be at least one $L^+$-block between $Q_{0}$ and $Q_1$. Let $\hat{L}^+$ be the $L^+$ closest to $Q_1$. Then the $\hat{L}^+$-strip is not left-compatible due to $\hat{L}^-$ in $(C_0,T_0)$ and contributes to $\gamma$ since it is not the last $L$-block in $(C_1,T_0)$. All other $L^+$ above $\hat{L}^+$, if they exist, are left-compatible as seen from the picture. Notice that it is possible to have $U$-blocks above $\hat{L}^-$. Then we replace $\hat{L}^-$ with the very first $U$ and argue as before.\end{minipage}};
\end{tikzpicture}
\end{center}
\end{proof}

Let us put everything together in Figure~\ref{fig:chain1} to emphasize the fact that we have unique pairing of non-right-compatible $L^-$-strips with non-left-compatible $L^+$-strips. 
\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};

\draw (0,-1*\X) rectangle (\X*1,0*\X) node[fitting node] (0-1) {};
\node at (0-1.center) {$\scriptscriptstyle{L^-}$};

\draw (0,-2*\X) rectangle (\X*1,-1*\X) node[fitting node] (0-2) {};
\draw (0,-3*\X) rectangle (\X*1,-2*\X) node[fitting node] (0-3) {};
\node at (0-3.center) {${\color{red}\scriptscriptstyle{L^-}}$};
\node at ($(0-3)-(0.6,0)$) {$\scriptscriptstyle{\text{last}}$};
\draw (0,-4*\X) rectangle (\X*1,-3*\X) node[fitting node] (0-4) {};
\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\node at (02.center) {$\scriptscriptstyle{L^-}$};
\node at ($(02)-(0.6,0)$) {$\scriptscriptstyle{\text{first}}$};

\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};

\node at (1.5*\X, \X*6) {$\scriptscriptstyle{B_{1,0}}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*0) rectangle (\X*2,\X*1) node[fitting node] (10) {};
\draw (\X*1,-1*\X) rectangle (\X*2,\X*0) node[fitting node] (1-1) {};
\draw (\X*1,-2*\X) rectangle (\X*2,-1*\X) node[fitting node] (1-2) {};
\draw (\X*1,-3*\X) rectangle (\X*2,-2*\X) node[fitting node] (1-3) {};

\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\draw (\X*1,\X*4) rectangle (\X*2,\X*5) node[fitting node] (14) {};
\draw ($(0-3)+(-1,0.25)$) -- ($(1-3)+(0,0.25)$)
($(02)+(-1,0.25)$) -- ($(12)+(0,0.25)$)
;

\path[arrows={- latex'}]
    (02.center) edge[bend right] node [right] {} (11.west)
    (0-1.center) edge[bend right] node [right] {} (1-2.west)
    (0-3.east) edge[red,bend right=90] node [right] {} (13.east);

\node at (11.center) {$\scriptscriptstyle L^+$};
\node at (1-2.center) {$\scriptscriptstyle L^+$};
\node at (13.center) {$\color{red} \scriptscriptstyle L^+$};

\draw [decorate,decoration={brace,amplitude=10pt},xshift=-4pt,yshift=0pt]
($(0-3)+(-1,0.25)$) -- ($(02)+(-1,0.25)$) node [black,midway,xshift=-0.6cm] 
{};


\draw [decorate,decoration={brace,amplitude=10pt},xshift=-4pt,yshift=0pt]
($(1-3)+(0.25,0.25)$) -- ($(1-3)+(0.25,-1)$) node [black,midway,xshift=-0.6cm] 
{};


\node at (-4,0.25) {\begin{minipage}{0.35\textwidth} There is a unique non-left-com\-pa\-tible $L^+$ adjacent to every non-right-com\-pa\-tible $L^-$. All other $L^+$-strips in $(C_1,T_0)$ are left-com\-pa\-tible.  \end{minipage}};


\node at (4.5,-1.5) {\begin{minipage}{0.35\textwidth} All $L^+$ in $(C_1,T_0)$ in this section are either left-compatible or do not contribute to $\gamma$. \end{minipage}};

\end{tikzpicture}
\caption{Unique pairings of non-exchangeable $L^-$ and $L^+$. }
\label{fig:chain1}
\end{center}
\end{figure} 


\begin{cor}\label{chain1cor}
Let $C=(C_{-1},C_0)$ and $T=(T_{-1},T_0)$ be a pair of KR-tableaux. To every non-left-compatible $L^+$-strip in $(C_0,T_0)$ that contributes to $\gamma$, there exists a unique non-right-compatible $L^-$-strip in $(C_{-1},T_{0})$ that contributes to $\gamma$. All other $L^-$-strips in $(C_{-1},T_0)$ are right-compatible. 
\end{cor}
\begin{proof}
Left-compatibility and right-compatibility are, in fact, identical requirements with different points of references. This is evident in the underlying inequalities being the same (see Definition \ref{defn:compat}). We simply start with non-left-compatible $L^+$-strips in $(C_0,T_0)$ and assign the $L^-$-blocks in $(C_{-1},T_{0})$ that pair with the $L^+$-blocks by Lemma \ref{chain1}. 
\end{proof}

Notice that in the proof of Lemma \ref{chain1}, when considering the last $L^-$-strip with $(C_0,T_0)$ regular (case two), we did not use the fact that $L^-$ is not right-compatible. Moreover, this situation includes the case when $L^-$ is not column-compatible.  
\begin{cor}\label{chain2}
Suppose $(C_0,T_0)$ is regular and there is a non-column-compatible $L^-$-strip. If there is a column in $C$ to the right of $C_0$, call it $C_1$, then there is a unique non-left-compatible $L^+$ in $(C_1,T_0)$. \qed
\end{cor}

\begin{lemma}\label{chain3}
Suppose $(C_0,T_0)$ is regular and there is a non-column-compatible $L^-$. If there is a column in $T$ to the left of $T_0$, call it $T_{-1}$, then $L^-$ is never \emph{left}-compatible.
\end{lemma}

\begin{proof}
We want to show that Condition $(lT)$ is not satisfied. We have the following picture:

\begin{center}
\begin{tikzpicture}


\node at (1.5*\X, \X*6) {$\scriptstyle{C_0}$};
\draw (\X*1,-1*\X) rectangle (\X*2,\X*0) node[fitting node] (1-1) {};
\draw (\X*1,\X*0) rectangle (\X*2,\X*1) node[fitting node] (10) {};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};

\node at (0.5*\X+\D, \X*6) {$\scriptstyle{T_{-1}}$};
\draw[white] (0+\D,-1*\X) rectangle (\X*1+\D,\X*0) node[fitting node] (t0-1) {};
\draw (0+\D,0) rectangle (\X*1+\D,\X*1) node[fitting node] (t00) {};
\draw (0+\D, \X*1) rectangle (\X*1+\D, \X*2) node[fitting node] (t01) {};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\node at (t0-1.center) {$\scriptstyle \color{red} \infty$};

\node at (1.5*\X+\D, \X*6) {$\scriptstyle{T_0}$};
\draw[white] (\X*1+\D,\X*0) rectangle (\X*2+\D,\X*1) node[fitting node] (t10) {};
\draw (\X*1+\D,\X*1) rectangle (\X*2+\D,\X*2) node[fitting node] (t11) {};
\draw (\X*1+\D,\X*2) rectangle (\X*2+\D,\X*3) node[fitting node] (t12) {};
\draw (\X*1+\D,\X*3) rectangle (\X*2+\D,\X*4) node[fitting node] (t13) {};
\draw[dotted] (\X*1+\D,\X*4) rectangle (\X*2+\D,\X*5) node[fitting node] (t14) {};
\node at (t10.center) {$\scriptstyle \infty$};

\draw[arrows={- latex'},blue] (t14.center) -- (13.center);
\draw[arrows={- latex'},blue] (13.center) -- (t13.center);
\draw[arrows={- latex'},dotted,blue] (12.center) -- (t13.center);
\draw[arrows={- latex'},dotted,blue] (11.center) -- (t12.center);
\draw[arrows={- latex'},dotted,blue] (10.center) -- (t11.center);
\draw[arrows={- latex'},dotted,blue] (1-1.center) -- (t10.west);
\draw[arrows={- latex'},red] (10.center) -- (t0-1.west);
\node at ($(t0-1.center)+(1,0)$) {$\scriptscriptstyle\color{red} \text{not } N^-$};

\node at (0 + 2*\D, \X*6) {$\scriptscriptstyle{B_{0,-1}}$};
\draw (0+2*\D,-1*\X) rectangle (\X*1+2*\D,\X*0) node[fitting node] (g0-1) {};
\draw (0+2*\D,0) rectangle (\X*1+2*\D,\X*1) node[fitting node] (g00) {};
\draw (0+2*\D, \X*1) rectangle (\X*1+2*\D, \X*2) node[fitting node] (g01) {};
\draw (0+2*\D, \X*2) rectangle (\X*1+2*\D, \X*3) node[fitting node] (g02) {};

\node at (g0-1.center) {$\color{red} \scriptstyle{X}$};

\node at (2*\X+2*\D, \X*6) {$\scriptscriptstyle{B_{0,0}}$};
\draw (\X*1+2*\D,\X*0) rectangle (\X*2+2*\D,\X*1) node[fitting node] (g10) {};
\draw (\X*1+2*\D,\X*1) rectangle (\X*2+2*\D,\X*2) node[fitting node] (g11) {};
\draw (\X*1+2*\D,\X*2) rectangle (\X*2+2*\D,\X*3) node[fitting node] (g12) {};
\draw (\X*1+2*\D,\X*3) rectangle (\X*2+2*\D,\X*4) node[fitting node] (g13) {};

\node at (g10.center) {$\scriptstyle{N^+}$};
\node at (g11.center) {$\scriptstyle{N^+}$};
\node at (g12.center) {$\scriptstyle{N^+}$};
\node at (g13.center) {$\scriptstyle{L^-}$};

\end{tikzpicture}
\end{center}
As seen from the picture, the last block in $B_{0,-1}$ is not $N_{-}$, and, therefore, the $L^-$ is not left-compatible. 
\end{proof}

\begin{lemma} \label{fundClustShape}
Let $C=(C_i)$ and $T=(T_j)$ be KR-tableaux from $\mathcal{C}$. Suppose $(C_i,T_j)$ is regular. Then at least one of $C_{i+1}$ or $T_{j-1}$ must exist. In other words, either there is a column to the right of $C_i$ or to the left of $T_j$ or both. 
\end{lemma}

\begin{proof}
Since the central columns of $C$ and $T$ form a fundamental pair (see Section \ref{fundClusterDiagrams}), the central column of $T$ must be to the left of $T_j$ or the central column of $C$ must be to the right of $C_i$. 
\end{proof}

When looking for $L^+$ to cancel out the contributions of non-exchangeable $L^-$-blocks, we need only address left-compatible $L^-$-blocks. The reason is that non-left-compatible $L^-$-blocks would have already been paired with a non-right-compatible $L^+$-blocks previously. Lemma \ref{chain1} addresses most of the situations when we have a non-exchangeable $L^-$ that contributes to $\gamma$. Lemma \ref{fundClustShape} shows that there are exactly two other situations when $L^-$ contributes to $\gamma$ and needs a pair. Lemmas \ref{chain2} and \ref{chain3} address each of those situations respectively. This concludes the unique pairing when we have a non-exchangeable $L^-$ that contributes to $\gamma$. 

Next, we address the case when we have a non-exchangeable $L^-$ that does not contribute to $\gamma$. We must ensure that there is no corresponding $L^+$, which would create an imbalance. There are two possibilities as listed in Table \ref{tab:cancellations}.

\begin{lemma}\label{chain4}
Suppose $(C_0,T_0)$ is fundamental. Then every $L^-$-strip in $(C_0,T_0)$ is column-compatible, i.e. contributes to $\gamma$.
\end{lemma}
\begin{proof}
By Lemma \ref{rmk:obs2}(b), the last block in $B_{0,0}$ is not $N^+$, and, therefore, the boundary term of $\gamma(C_0,T_0)$ is zero. 
\end{proof}

\begin{remark} \label{rmk:obs4}
Let $C=(C_0,C_1)$ and $T=(T_0)$ be KR-tableaux. If $(C_0,T_0)$ has an $L^+UN^-$-block at $p$, then $(C_1,T_0)$ has an $N^-$-block at $p$ (see Figure~\ref{fig:obs4}).  
\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

\node at (0,0) {
\begin{tikzpicture}
\node at (0.5*\X, \X*5) {$\scriptstyle{C_0}$};

\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at ($(02) +(-.6,0)$) {$\scriptstyle p$};
\node at ($(03) +(-.6,0)$) {$\scriptstyle p-1$};

\node at (\X*1.5, \X*5) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};

\node at (0.5*\X+\D, \X*5) {$\scriptstyle{T_0}$};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};
\node at ($(t02) +(1.5,0)$) {$\scriptscriptstyle N^-\text{ in }(C_0,T_0)$};

\draw[arrows={ - latex'},dotted,blue] (02.center) -- (13.center);
\draw[arrows={ - latex'},blue] (03.center) -- (02.center);
\draw[arrows={ - latex'},blue,dotted] (t02.center) -- (03.center);
\draw[arrows={ - latex'},red,thick,dotted] (t02.center) -- (13.center);

\end{tikzpicture}
};

\node at (6,0) {
\begin{tikzpicture}
\node at (0.5*\X, \X*5) {$\scriptstyle{C_0}$};

\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at ($(02) +(-.6,0)$) {$\scriptstyle p$};
\node at ($(03) +(-.6,0)$) {$\scriptstyle p-1$};

\node at (\X*1.5, \X*5) {$\scriptstyle{C_1}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};

\node at (0.5*\X+\D, \X*5) {$\scriptstyle{T_0}$};
\draw (0+\D, \X*2) rectangle (\X*1+\D, \X*3) node[fitting node] (t02) {};
\draw (0+\D, \X*3) rectangle (\X*1+\D, \X* 4) node[fitting node] (t03) {};
\node at ($(t02) +(1.5,0)$) {$\scriptscriptstyle L^+U\text{ in }(C_0,T_0)$};

\draw[arrows={ - latex'},dotted,blue] (02.center) -- (13.center);
\draw[arrows={ - latex'},blue,dotted] (t02.center) -- (02.center);
\draw[arrows={ - latex'},red,thick,dotted] (t02.center) -- (13.center);


\end{tikzpicture}
};
\end{tikzpicture}
\end{center}
\caption{Composition of arrows.}
\label{fig:obs4}
\end{figure}
Indeed, by composing arrows, we find $T_0[p]\leq C_1[p-1]$, which means there is an $N^-$-block at $p$ in $(C_1,T_0)$.  
\end{remark}



\begin{lemma}\label{chain5}
Suppose $(C_0,T_0)$ is anti-fundamental and there is a non-right-compatible and non-column-compatible $L^-$-strip. Then there does not exist a corresponding non-exchangeable $L^+$-strip in $(C_1,T_0)$.  
\end{lemma}

\begin{proof}
Let $\tilde{L}^-$ be the $L^-$-strip in question and let $p'$ be its index. Since $L^-$ is not column-compatible, it must be the last $L$-strip in $(C_0,T_0)$ and the last block in $(C_0,T_0)$ must be $N^+$. In other words, the boundary term of $\gamma(C_0,T_0)$ is $+1$, which cancels out the contribution of $\tilde{L}^-$. Since the contribution of $\tilde{L}^-$ is already canceled out, we want to show that there is no non-exchangeable $L^+$ in $(C_1,T_0)$ that is paired with $\tilde{L}^-$. It suffices to show that all possible $L^+$'s that can be paired with $\tilde{L}^-$ are, in fact, left-compatible. Notice that there must be $C_1$ in order for $\tilde{L}^-$ to fail the condition $(rC)$.  
\begin{center}
\begin{tikzpicture}

\node at (0.5*\X, \X*5) {$\scriptscriptstyle{B_{0,0}}$};
\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\node at (00.center) {$\scriptscriptstyle{N_+}$};

\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};
\node at (01.center) {$\scriptscriptstyle{N^+}$};
\draw (0, \X*2) rectangle (\X*1, \X*3) node[fitting node] (02) {};
\node at (02.center) {$\scriptscriptstyle{N^+}$};
\draw (0, \X*3) rectangle (\X*1, \X* 4) node[fitting node] (03) {};
\node at (03.center) {$\scriptscriptstyle{\tilde{L}^-}$};
\node at ($(03.center)+(-.5,0)$) {$\scriptscriptstyle p'$};



\node at (1.5*\X, \X*5) {$\scriptscriptstyle{B_{1,0}}$};
\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\node at (11.center) {$\scriptscriptstyle{Q_2}$};
\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};
\draw (\X*1,\X*3) rectangle (\X*2,\X*4) node[fitting node] (13) {};
\node at (13.center) {$\scriptscriptstyle{Q_1}$};


\node at (\X*1+4*\D, \X*3) {\begin{minipage}{0.6\textwidth}
Let $Q_1$ and $Q_2$ be the index $p'$ and the last block in $(C_1,T_0)$. If there are any $L^+$-blocks between $Q_1$ and $Q_2$, then they are all left-compatible. Indeed, $(lC)$ is satisfied (seen from the picture) and $(lT)$ is always satisfied for $L^+$-strips. \end{minipage}};
\end{tikzpicture}
\end{center}

Let $\hat{L}^-$ be the very first $L^-$-strip in $(C_0,T_0)$. 

\begin{center}
\begin{tikzpicture}


\node at (0.5*\X, \X*4) {$\scriptscriptstyle{B_{0,0}}$};
\draw[white] (0,-2*\X) rectangle (\X*1,-1*\X) node[fitting node] (0-2) {};
\node at (0-2.center) {$\scriptscriptstyle{\vdots}$};
\draw[white] (\X*1,-1*\X) rectangle (\X*2,\X*0) node[fitting node] (1-1) {};
\node at (1-1.center) {$\scriptscriptstyle{\vdots}$};


\draw (0,0) rectangle (\X*1,\X*1) node[fitting node] (00) {};
\node at (00.center) {$\scriptscriptstyle{\hat{L}^-}$};

\draw (0,-1*\X) rectangle (\X*1,0*\X) node[fitting node] (0-1) {};
\node at (0-1.center) {$\scriptscriptstyle{N^+}$};

\draw (0, \X*1) rectangle (\X*1, \X*2) node[fitting node] (01) {};

\node at (1.5*\X, \X*4) {$\scriptscriptstyle{B_{1,0}}$};

\draw (\X*1,\X*1) rectangle (\X*2,\X*2) node[fitting node] (11) {};
\node at (10.center) {$\scriptscriptstyle{Q}$};
\draw (\X*1,\X*0) rectangle (\X*2,\X*1) node[fitting node] (10) {};

\draw (\X*1,\X*2) rectangle (\X*2,\X*3) node[fitting node] (12) {};


\node at (\X*1+4*\D, \X*1) {\begin{minipage}{0.6\textwidth}
Since $(C_0,T_0)$ is anti-fundamental, by Lemma \ref{rmk:obs2}, the first block in $B_{0,0}$ is not $N^+$. Since $\hat{L}^-$ is the first $L^-$-strip in $(C_0,T_0)$, the first block in $B_{0,0}$ is not an $L^-$-block either. So, it can be either $N^-, U$ or $L^+$. By Remark \ref{rmk:obs1}, since $N^+$ cannot follow an $L^+N^-U$-block, there are no $N^+$-blocks above $\hat{L}^-$.  By Remark \ref{rmk:obs4}, the adjacent blocks in $(C_1,T_0)$, i.e. the blocks above $Q$, are all $N^-$. In other words, there are no $L^+$-blocks above $Q$. This concludes the proof. 
\end{minipage}};
\end{tikzpicture}
\end{center}
\end{proof}


We are now ready to put everything together. 

\begin{proof}[Proof of Theorem \ref{thm:main1}]
If there are no $L$-blocks anywhere in $(C,T)$, the statement is trivially true. Let's assume there is at least one $L$-block in $(C,T)$, and without loss of generality, we may assume it is $L^-$. Otherwise we consider $(T,C)$ instead. Moreover, we can assume the $L^-$ is left-compatible. Indeed, if there is a non-left-compatible $L^-$ in $(C_i,T_j)$, there is a non-left-compatible $L^+$ in $(T_j,C_i)$. By Corollary \ref{chain1cor}, there is a non-right-compatible $L^-$ in $(T_{j-1},C_{i})$. If this $L^-$ is again non-left-compatible, we go through the same chain of arguments. We continue inductively and eventually, since the process must end when $C$ or $T$ runs out of columns, we are guaranteed to find a left-compatible $L^-$.  

Consider a left-compatible $L^-$-strip in $(C_i,T_0)$. If there are no such $L^-$-strips in $(C,T_0)$, which happens if there are no $L^-$-blocks in $(C,T_0)$, we remove $T_0$ from $T$ and consider $(C,T_1)$. The list of possibilities for $L^-$ are listed in Table \ref{tab:cancellations}.
\begin{enumerate}
\item Suppose the $L^-$-block in question contributes to $\gamma$. If it is non-right-compatible, there must be $C_{i+1}$. This is because all $L^-$-strips satisfy the condition $(rT)$. In order for the $L^-$ to be non-right-compatible, there must be $C_{i+1}$ that pose violations. By Lemma \ref{chain1}, there exists a unique non-left-compatible $L^+$ in $(C_{i+1},T_j)$. If $(C_i,T_0)$ is regular and there exists $C_{i+1}$, then Corollary \ref{chain1cor} is used. If there is no $C_{i+1}$, by Lemma \ref{fundClustShape} there must be $T_{-1}$ and by Lemma \ref{chain3}, the $L^-$ is not left-compatible, which is a contradiction. 
\item Suppose the $L^-$-block does not contribute to $\gamma$. Then $(C_i,T_0)$ is (anti-)fundamental. If $(C_i,T_0)$ is fundamental, then every $L^-$ contributes to $\gamma$ and we apply the previous analysis. If $(C_i,T_0)$ is anti-fundamental and $L^-$ is right-compatible, there is no need to pair it with anything since $L^-$ does not contribute to $\gamma$ and pose no violations with $C_{i+1}$. If $(C_i,T_0)$ is anti-fundamental and $L^-$ is not right-compatible, Lemma \ref{chain4} shows there is no corresponding non-left-compatible $L^+$'s. 
\end{enumerate}

We now remove $T_0$ from $T$ and consider $\gamma(C,T_1)$. Since $T_0$ is removed, all non-left-compatible $L^-$-strips become left-compatible. However, restrictions on $L^+$-strips are not changed since left-compatibility comes from $C$ itself. By the exact same argument, all negative non-exchangeable contributions in $\gamma(C,T_1)$ are canceled by positive non-exchangeable contributions. We continue this argument for all $T_i$. This proves all the negative non-exchangeable terms in $\gamma(C,T)$ are uniquely canceled out by non-right-compatible terms in $\gamma(C,T)$. 

Next, we consider $(T,C)$, where all left-compatible $L^+$-strips in $(C,T)$ become left-compatible $L^-$-strips in $(T,C)$ and apply the same argument. 
\end{proof}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Proof of Conditions II and III}
The previous section showed that the condition I holds. That is, the $(q,t)$-characters in the fundamental cluster $t$-commute with one another. Then, by Corollary \ref{gammaComm}, the commutation matrix $\Lambda$ of the fundamental cluster $\mathcal{C}$ is given by:
\begin{eqnarray*}
\Lambda_{i,k}^{i',k'} = 2\epsilon( \mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)}, \mathbf{Y}_{k',-k'+(i'+k'+1)_2}^{(i')} )\,\, ,
\end{eqnarray*}
where $\mathbf{Y}_{k,j(i,k)}^{(i)}$ and $\mathbf{Y}_{k',j(i',k')}^{(i')}$ are dominant monomials of KR-modules in the fundamental cluster $\mathcal{C}$.  

We now show condition II holds. 

\begin{thm} \label{main2}
Let $B$ be the infinite matrix associated to the quiver $\Gamma_B$ in Figure~\ref{fig:Tsystem} (on page~\pageref{fig:Tsystem} ). Then, 
\begin{equation*}
\Lambda B = D\,,
\end{equation*}
where $D$ is a diagonal matrix with positive entries. In other words, $(\Lambda, B)$ is a compatible pair (as in \cite{berenstein2005quantum}, Section $3$). 
\end{thm}

We will provide some definitions and lemmas first. 

\begin{defn} \label{notationCond2}
We fix the following notations for convenience:
\begin{enumerate}
\item Given $k\in \mathbb{Z}$, the $s$-number is defined as:
\begin{eqnarray*}
\left[0\right]_s&:=&0\,,\\
\left[k\right]_s &:=& \frac{ s^k - s^{-k}}{s-s^{-1}} = s^{k-1} + s^{k-3} + \cdots + s^{-k+3} + s^{-k+1}\,.
\end{eqnarray*}
\item Denote $\mathcal{F} := \mathbb{Z}[[s]][s^{-1}]$. Then, multiplication operator is well-defined in $\mathcal{F}$.
\item Given $f(s)\in \mathcal{F}$, we define $\left[f\right]_0$ to be the constant term in $f(s)$, e.g. $[s^{-1}+3+s]_0=3$. 
\end{enumerate}
\end{defn}


\begin{defn} \label{uGenFun}
Let $m$ be a monomial in $\mathbb{Z}[Y_{i,j}^{\pm 1}]$. We define the following generating series:
\begin{eqnarray*}
u_{i}(m)(s) := \sum_{j\in \mathbb{Z}} u_{i,j}(m) s^j &;& u(m)(s) := \sum_{i=1}^r e_i \otimes u_{i}(m)(s)\,\, ,
\end{eqnarray*}
where $u_{i,j}(m)$ is the exponent of $Y_{i,j}$ in $m$ as defined in Definition \ref{uv} and $e_i \in \mathbb{Z}^r$ is the vector with $1$ in the $i$th position and $0$'s everywhere else. 
\end{defn}

\begin{lemma} \label{sNum}
Let $\mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)}$ be the dominant monomial of $\chi_{k,-k+(i+k+1)_2}^{(i)}\in \mathcal{C}$. Then,
\begin{eqnarray*}
u(\mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)})(s)&=& e_i \otimes s^{-1+(i+k+1)_2}\,\,[k]_s\,.
\end{eqnarray*}
\end{lemma}
\begin{proof}
Denote $j:= (i+k+1)_2$. Modules in $\mathcal{C}$ have dominant monomials of the form:
\begin{eqnarray} \label{Yexp}
\mathbf{Y}_{k,j}^{(i)} = Y_{i, -k+j} Y_{i, -k+j+2} \cdots Y_{i, -k+ j + 2(k-1)}\,.
\end{eqnarray}
Then, using Definition \ref{uGenFun}, we directly compute as follows: 
\begin{eqnarray*}
u_i (\mathbf{Y}_{k,j}^{(i)})(s) &=& \sum_{p\in \mathbb{Z}} u_{i,p}(\mathbf{Y}_{k,j}^{(i)})s^p \\
&=&s^{-k+j} + s^{-k+j+2} + \cdots +s^{j+k-2} \\
&=& s^{j-1} \left(s^{-k+1} + s^{-k+3} + \cdots  + s^{k-1}\right)\\
&=&s^{j-1} [k]_s\,,
\end{eqnarray*}
where $u_{i,p}\left( \mathbf{Y}_{k,j}^{(i)}\right)$ is the power of $Y_{i,p}$ in $\mathbf{Y}_{k,j}^{(i)}$, which is either $1$ or $0$ as seen from \eqref{Yexp}. Notice that $u_{i',p}(\mathbf{Y}_{k,j}^{(i)})=0$ if $i' \neq i$. Therefore, $u_{i'}(\mathbf{Y}_{k,j}^{(i)})(s)=0$ and we have:
\begin{eqnarray*}
u(\mathbf{Y}_{k,j}^{(i)}) = e_i \otimes s^{j-1}[k]_s\,.
\end{eqnarray*}
\end{proof}

\begin{defn} \label{utildeGenFun}
Let $m$ be a monomial in $\mathbb{Z}[Y_{i,j}^{\pm 1}]$. We define the following generating series:
\begin{eqnarray*}
\tilde{u}_i(m)(s) = \sum_{j\in \mathbb{Z}} \tilde{u}_{i,j}(m) s^j &\text{and}& \tilde{u}(m)(s) = \sum_{i=1}^r e_i \otimes \tilde{u}_i(m)(s)\,,
\end{eqnarray*}
where $\tilde{u}_{i,j}(m)$ are the solutions of the system given in Definition \ref{uinverse}. 
\end{defn}

\begin{defn}\label{convention}
When generating functions $u$ and $\tilde{u}$ of Definitions \ref{uGenFun} and \ref{utildeGenFun} are applied to the dominant monomial of a module in the fundamental cluster $\mathcal{C}$, i.e. monomial of the form $\mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)}$, we make the following simplifying notation:
\begin{eqnarray*}
u(\mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)})(s) := u^{i,k}(s) &\text{ and }& \tilde{u}(\mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)})(s) :=  \tilde{u}^{i,k}(s)\,.
\end{eqnarray*}
\end{defn}
\begin{remark}
We emphasize that the superscript $u^{i,k}(s)$ in Definition \ref{convention} indicates the monomial $Y_{k,-k+(i+k+1)_2}^{(i)}$. In contrast, the subscript $u_{i,j}(m)$ indicates the exponent of $Y_{i,j}$ in $m$. 
\end{remark}

\begin{defn}\label{action}
Let $M\in {\rm{Mat}}_{r\times r}(\mathbb{Z})$ and $g \in \mathcal{F}$. We define an action of $M\otimes g$ on the space $\mathbb{Z}^r \times \mathcal{F}$ as follows:
\begin{eqnarray*}
\left(M\otimes g\right)\left(v\otimes f \right) = \left(Mv\right)\otimes \left(gf\right) &\text{ for any }& v\in \mathbb{Z}^r, f \in  \mathcal{F},
\end{eqnarray*}
where $Mv$ is the matrix multiplication and $gf$ is the usual multiplication. 
\end{defn}


Recall the system of equations in Definition \ref{uinverse}:
\begin{equation*} 
u_{i,j}(m) = \tilde{u}_{i,j-1}(m) + \tilde{u}_{i,j+1}(m) - \tilde{u}_{i-1,j}(m) - \tilde{u}_{i+1,j}(m)\,,
\end{equation*}
defined for any monomial $m$ and any $i\in I, j\in \mathbb{Z}$. Let $A = C-2I$, where $C$ is the Cartan matrix of $\mathfrak{sl}_{r+1}$. We rewrite this system for $m=\mathbf{Y}_{k,-k+(i+k+1)_2}^{(i)}$ as follows:
\begin{equation}
u^{i,k}(s) = (1\otimes s + 1\otimes s^{-1} + A\otimes 1) \tilde{u}^{i,k}(s)\,,
\end{equation}
where the action is as in Definition \ref{action}.
\begin{defn}
Denote the operator
\begin{equation} \label{K}
K = (1\otimes s^{-1}) ( 1\otimes 1 + A\otimes s + 1\otimes s^{2})\,.
\end{equation}
Define operator $D$ as a formal power series in $s$, expanded at $0$, given by:
\begin{eqnarray} \label{D}
D=(1\otimes 1 +  A\otimes s + 1\otimes s^2 )^{-1}(1\otimes s)\,.
\end{eqnarray}
Then we have $DK=KD=1\otimes 1$ and $Du^{i,k}(s) = \tilde{u}^{i,k}(s)$.
\end{defn}
\begin{remark}
In Definition \ref{uinverse}, we require $\tilde{u}_{i,j}(m)=0$ for $j$ sufficiently small. This condition is equivalent to expanding the formal inverse of the power series \eqref{K} at $0$, which is the choice we made in \eqref{D}.  
\end{remark}

\begin{remark}
It is easy to see that $K$ commutes with $1\otimes s^n$ for any $n$. Since $D$ is the inverse of $K$, we have:
\begin{eqnarray*}
D(1\otimes s^n) = D(1\otimes s^n)KD = DK(1\otimes s^n) D = (1\otimes s^n) D\,.
\end{eqnarray*}
That is, the operator $D$ commutes with $1\otimes s^n$ for $n\in \mathbb{Z}$. 
\end{remark}


\begin{defn} \label{myInner}
Let $v,w\in \mathbb{Z}^r$ and $f,g\in \mathcal{F}$. We define the following inner product on $\mathbb{R}^r\otimes \mathcal{F}$ as follows:
\begin{equation}
\left(v\otimes f \right) \cdot \left(w\otimes g \right) = \left< v,w\right>  \left[ f\left(s^{-1}\right)g\left(s\right) \right]_{0}\,,
\end{equation}
where $\left<v,w\right>$ is the usual inner product on $\mathbb{R}^r$ and $\left[f(s)\right]_{0}$ is the constant term in $f(s)$ as defined in Definition \ref{notationCond2}. 
\end{defn}
\begin{remark}
Notice that the inner product in Definition \ref{myInner} is symmetric. That is, 
\begin{equation*}
(v\otimes f) \cdot (w\otimes g) = (w\otimes g) \cdot (v\otimes f)\,. 
\end{equation*}
\end{remark}

\begin{defn}
Given $M\otimes h \in {\rm{Mat}}_{r\times r}(\mathbb{Z}) \times \mathcal{F}$, we define the \emph{transpose} of $M\otimes h$, denoted $(M\otimes h)^t$, by the following condition:
\begin{equation*}
\left(v\otimes f \right) \cdot M\otimes h \left(w\otimes g\right) = (M\otimes h)^t\left(v\otimes f\right) \cdot \left(w\otimes g\right)\,, 
\end{equation*}  
for any $v,w\in \mathbb{Z}^r$ and $f,g\in \mathcal{F}$. An operator $M\otimes h$ is \emph{symmetric} if $M\otimes h=(M\otimes h)^t$. 
\end{defn}
\begin{lemma} \label{transpose}
\begin{eqnarray*}
(1\otimes s)^t = 1\otimes s^{-1} ;&K^t = K; & D^t = D\,.
\end{eqnarray*}
\end{lemma}
\begin{proof}
It is easy to see that $K$ is symmetric. Since $D^{-1}=K$, $D$ is also symmetric. The remaining result is shown by direct computation:
\begin{eqnarray*}
\left(v\otimes f(s) \right) \cdot \left(1\otimes s\right) \left(w\otimes g(s) \right) &=& \left< v,w\right> \left[f(s^{-1}) (sg(s)) \right]_0 \\
&=&\left< v,w\right> \left[(s^{-1})^{-1}f(s^{-1}) g(s) \right]_0  \\&=& (1\otimes s^{-1}) (v\otimes f(s)) \cdot (w\otimes g(s))\,,
\end{eqnarray*}
for any $v,w\in \mathbb{Z}^r$ and $f,g\in \mathcal{F}$. 
\end{proof}


\begin{lemma} \label{EpsilonForm}
Let $p,p'$ be dominant monomials in $\mathcal{M}$. Then,
\begin{equation*}
\epsilon\left(p,p'\right) = (1\otimes s- 1\otimes s^{-1})Du\left(p\right)(s) \cdot u\left(p'\right)(s)\,,
\end{equation*}
where $\epsilon$ is given in Definition \ref{def:Nakajima}.
\end{lemma}

\begin{proof}
By definition, we have
\begin{eqnarray*}
u(p)(s)=\sum_i e_i \otimes \sum_{j} u_{i,j}(p)s^j &\text{ and }&\tilde{u} (p')(s) = \sum_{i'} e_{i'}\otimes \sum_{j'=1}^r \tilde{u}_{i',j'}(p') s^{j'}\,.
\end{eqnarray*}
Then,
\begin{eqnarray*}
u(p)(s) \cdot (1\otimes s) \tilde{u}(p')(s) &=& \sum_{i,i'=1}^r \left<e_i, e_{i'} \right>  \left[ \sum_{j,j'} u_{i,j}(p) s^{-j} \tilde{u}_{i',j'}(p')s^{j'+1}\right]_0 \\ &=& \sum_j u_{i,j}(p) \tilde{u}_{i, j-1}(p')\,.
\end{eqnarray*}
Therefore, using Deifnition \ref{def:Nakajima}, we have:
\begin{eqnarray*}
\epsilon(p,p')&=& - \sum_{i,j} u_{i,j}(p) \tilde{u}_{i,j-1}(p') + \sum_{j} u_{i,j}(p') \tilde{u}_{i,j-1}(p) \\
&=&- u(p)(s) \cdot (1\otimes s) \tilde{u}(p')(s)+ u(p')(s) \cdot (1\otimes s) \tilde{u}(p)(s)\\
&=&- u(p)(s) \cdot (1\otimes s) Du(p')(s)+ u(p')(s) \cdot (1\otimes s)D u^(p)(s)\\
&=&- D^t (1\otimes s)^t u(p)(s) \cdot u(p')(s) + (1\otimes s) D u(p)(s) \cdot u(p')(s) \\
&=&(1\otimes s-1\otimes s^{-1})D u(p)(s) \cdot u(p')(s)\,.
\end{eqnarray*}
\end{proof}
\begin{remark}
As a sanity check, let us verify that the expression we found for $\epsilon$ is also anti-symmetric. 
\begin{eqnarray*}
\epsilon(p',p) &=& (1\otimes s - 1\otimes s^{-1} ) D u(p')(s) \cdot u(p)(s)\\
 &=& u(p)(s) \cdot (1\otimes s - 1\otimes s^{-1} ) D u(p')(s)\\
&=& D^t (1\otimes s^{-1} - 1\otimes s) u(p)(s) \cdot u(p')(s) = - \epsilon(p,p')\,.
\end{eqnarray*}
\end{remark}


%%%%
\begin{lemma}
The following equation holds:
\begin{eqnarray*}
 u^{i,k-1}(s) + u^{i,k+1}(s) - u^{i-1,k}(s) - u^{i+1,k}(s)   =  1\otimes s^{-1+(i+k)_2} Ke_i\otimes [k]_s\,,
\end{eqnarray*}
for any $i\in I$ and $k\geq 1$. 
\end{lemma}
\begin{proof}
We have:
\begin{eqnarray*}
[k-1]_s+[k+1]_s &=& \left(s^{k-2} + \cdots + s^{-k+2} \right) + \left(s^{k} + s^{k-2} +\cdots +s^{-k+2} + s^{-k}\right) \\
&=&  \left(s^{k-2}  + \cdots + s^{-k+2} + s^{-k}\right) + \left(s^{k} + s^{k-2} +\cdots + s^{-k+2}\right) \\
&=&s^{-1} \left(s^{k-1} + \cdots + s^{-k+1} \right) + s\left(s^{k-1} +\cdots + s^{-k+1}\right) \\
&=& (s^{-1}+s) [k]_s\,.
\end{eqnarray*}
Notice that the above equation holds when $k=1$ as well:
\begin{eqnarray*}
[0]_s + [2]_s = [2]_s = s + s^{-1} = \left( s+s^{-1}\right) [1]_s\,. 
\end{eqnarray*}
By Lemma \ref{sNum}, we have $u^{i,k}(s)= e_i \otimes s^{-1+(i+k+1)_2}\,\,[k]_s$. Then,
\begin{eqnarray*}
u^{i,k-1}(s) &+& u^{i,k+1}(s) - u^{i-1,k}(s) - u^{i+1,k}(s) = \\
&=&e_i \otimes s^{-1 + (i+k)_2}[k-1]_s + e_i \otimes s^{-1 + (i+k)_2} [k+1]_s +\\
&& (-e_{i-1}-e_{i+1}) \otimes s^{-1+(i+k)_2} [k]_s\\
&=& 1\otimes s^{-1+(i+k)_2} \left( e_i \otimes \left( s+s^{-1}  \right) [k]_s + (-e_{i-1}-e_{i+1})\otimes [k]_s  \right)\\
&=& 1\otimes s^{-1+(i+k)_2} K e_i\otimes [k]_s\,.
\end{eqnarray*}
Notice that when $i=1$ or $i=r$, the above equation still holds since the matrix $A$ incorporates the boundary values. 
\end{proof}

We are now ready to prove the theorem stated at the beginning of the section. 

\begin{proof}[Proof of Theorem \ref{main2}]
We want to compute:
\begin{eqnarray}
\frac{1}{2} (\Lambda B) _{i,k}^{i',k'}&=& \frac{1}{2} \sum_{p,n} \Lambda_{i,k}^{p,n} B_{p,n}^{i',k'} \label{LambdaBValue}\\
&=&\sum_{p,n} (1\otimes s-1\otimes s^{-1}) D u^{i,k}(s) \cdot u^{p,n}(s) B_{p,n}^{i', k'} \,.\nonumber 
\end{eqnarray}
The goal is to show that the value of \eqref{LambdaBValue} is $1$ if $(i,k)= (i',k')$ and $0$ otherwise. 

Given $(i',k')$, the only non-zero terms in $B$ are given as follows:
\begin{eqnarray*}
B_{i'-1,k'}^{i',k'} = B_{i'+1,k'}^{i',k'} = (-1)^{i'+k'} &\text{ and }& B_{i',k'-1}^{i',k'} = B_{i',k'+1}^{i',k'} = (-1)^{i'+k'+1}\,,
\end{eqnarray*}
and $B_{n,p}^{i',k'} =0$ if $n\notin I$ or $p<0$ (see Figure~\ref{fig:Tsystem} on page~\pageref{fig:Tsystem}). Then,
\begin{eqnarray}
\frac{1}{2}(\Lambda B) _{i,k}^{i',k'} &=& (1\otimes s - 1\otimes s^{-1})  \nonumber \\ 
&& \hspace{0.1in} Du^{i,k}(s) \cdot (-1)^{i'+k'+1} \left( u^{i',k'-1}(s) + u^{i',k'+1}(s) -u^{i'-1,k'}(s) - u^{i'+1,k'}(s) \right) \nonumber\\ 
&=& (-1)^{i'+k'+1} (1\otimes s - 1\otimes s^{-1}) \nonumber \\
&&\hspace{0.1in} D e_i\otimes s^{-1+(i+k+1)_2} [k]_s \cdot  (1\otimes s^{-1 + (i'+k')_2} )K e_{i'}\otimes [k']_s\\
&=& (-1)^{i'+k'+1} ( 1\otimes s^{(i+k+1)_2-(i'+k')_2} )(1\otimes s - 1\otimes s^{-1}) e_i\otimes [k]_s  \cdot     e_{i'} \otimes [k']_s \nonumber \\ 
&=& (-1)^{i'+k'+1} \left< e_i,e_{i'}\right> \left[ ( s^{-1} - s)(s^{(i'+k')_2 - (i+k+1)_2} [k]_{s^{-1}} [k']_s\right]_0  \nonumber \\ 
&=& (-1)^{i+k'+1} \delta_{ii'} \left[ s^\delta \frac{1}{s-s^{-1}} \left( s^{k'-k} - s^{-k-k'} - s^{k+k'} + s^{k'-k}\right)     \right]_0 \label{CT}\,,
\end{eqnarray}
where $\delta:=(i+k')_2 - (i+k+1)_2$. Notice that $\delta$ can only be $+1,0$ or $-1$. 

It is clear that if $i\neq i'$, the value of \eqref{CT} is zero. Suppose $k\neq k'$, and without loss of generality, let's assume $k'>k$. We want to show that the constant term part of \eqref{CT} vanishes in this case. Let $b:=k'-k > 0$ and $a:=-k'-k < 0$. Notice that $a$ and $b$ have the same parity. 

If $\delta=1$, the constant term expression of \eqref{CT} is as follows:
\begin{eqnarray}\label{kneqk'}
\begin{array}{lll}s &\times& \frac{1}{s(1-s^{-2})} \left( (s^{b} + s^{-b}) - (s^{a} + s^{-a}) \right) =\\
&=&  (s^b + s^{-b}) ( 1+s^{-2}+s^{-4} +\cdots ) - (s^a+s^{-a}) (1+s^{-2} + s^{-4} +\cdots ) \end{array}
\end{eqnarray}
Notice that $-b-2n, a-2m < 0$ for any $n,m\geq 0$, and therefore, $s^{-b}s^{-2n}$ and $s^{a}s^{-2m}$ are not constants for any $n,m\geq 0$. If $a$ and $b$ are odd, it is clear that there is no constant term in \eqref{kneqk'}. If $a$ and $b$ are even, there exists some $n>0$ such that $s^{b-2n} = 1$. Since $a<0$, there must exist some $m>0$ such that $s^{-a-2m}=1$. Therefore, the constant term in \eqref{kneqk'} vanishes.  

If $\delta=-1$, we write
\begin{eqnarray*}
s^{-1} \times \frac{1}{s^{-1}(s^2-1)} \left( (s^{b} + s^{-b}) - (s^{a} + s^{-a}) \right) \,,
\end{eqnarray*}
and use the same argument. 

If $\delta = 0$, the constant term part of \eqref{CT} can be written as:
\begin{eqnarray*}
\frac{1}{(1-s^{-2})} \left( (s^{b-1} + s^{-b-1}) - (s^{a-1} + s^{-a-1}) \right)\,,
\end{eqnarray*}
and we use the same argument. Therefore, the value of \eqref{CT} is always zero if $k\neq k'$. 

Suppose $k=k'$. If $(i+k)_2=0$, we have $\delta = -1$ and \eqref{CT} can be written as:
\begin{eqnarray*}
\frac{1}{2}(\Lambda B)_{i,k}^{i,k} &=& - \left[ s^{-1} \frac{1}{-s^{-1}(1-s^2)} \left( 2 - s^{2k} - s^{-2k} \right)  \right]_0 \\
&=& \left[ \left( 2 - s^{2k} - s^{-2k} \right) ( 1 + s^{2} + s^{4} + \cdots + s^{2k} + \cdots ) \right]_0= 1\,.
\end{eqnarray*}

If $(i+k)_2=1$, we have $\delta=1$ and \eqref{CT} can be written as:
\begin{eqnarray*}
\frac{1}{2}(\Lambda B)_{i,k}^{i,k} &=& \left[ s \frac{1}{s(1-s^{-2})} \left( 2 - s^{2k}  - s^{-2k} \right)  \right]_0  \\
&=& \left[ \left( 2 - s^{2k} - s^{-2k} \right) ( 1 + s^{-2} + s^{-4} + \cdots + s^{-2k} + \cdots ) \right]_0= 1\,.
\end{eqnarray*}
This concludes the proof.
\end{proof}

We are now ready to prove the final condition.

\begin{thm} \label{main3}
The quantum mutation is given by: %Proposition 4.9 in BZ. 
\begin{eqnarray*}
T_{k,l-1}^{(i)}*T_{k,l+1}^{(i)} &=& t^{\frac{1}{2} \Lambda_{i,k,l-1}^{i,k-1,l} + \frac{1}{2}\Lambda_{i,k,l-1}^{i,k+1,l} - \frac{1}{2} \Lambda_{i,k-1,l}^{i,k+1,l} } T_{k-1,l}^{(i)}*T_{k+1,l}^{(i)}+\\
&& t^{\frac{1}{2} \Lambda_{i,k,l-1}^{i-1,k,l} + \frac{1}{2}\Lambda_{i,k,l-1}^{i+1,k,l} - \frac{1}{2} \Lambda_{i-1,k,l}^{i+1,k,l} } T_{k,l}^{(i-1)}*T_{k,l}^{(i+1)}.
\end{eqnarray*}
\end{thm}

Theorem \ref{main3} is stated in terms of $T_{k,l}^{(i)}$ variables, while Nakajima's $t$-deformed $T$-system is written in terms of $\chi_{k,j}^{(i)}$ variables, which is achieved by a change of variables as described in Remark \ref{rmk:changeOfVar}. Also recall that $\Lambda$ is expressed in terms of the $\epsilon$ function (see Corollary \ref{gammaComm}). We now restate Theorem \ref{main3} in terms of $\chi_{k,j}^{(i)}$ variables and $\epsilon$ expressions for $\Lambda$. 
\begin{thm}\label{main3chi}
The following equation holds:
\begin{eqnarray}\label{cond3}
\begin{array}{ccl}
\chi_{k,j}^{(i)}*\chi_{k,j+2}^{(i)} &=& t^{ \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k-1,j+2}^{(i)}) + \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k+1,j}^{(i)}) - \epsilon(\mathbf{Y}_{k+1,j+2}^{(i)}, \mathbf{Y}_{k-1,j}^{(i)})  } \,\, \chi_{k+1,j}^{(i)}*\chi_{k-1,j+2}^{(i)} + \\
&&t^{\epsilon( \mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+1}^{(i-1)} ) + \epsilon( \mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+1}^{(i+1)} ) -  \epsilon(\mathbf{Y}_{k,j+1}^{(i-1)}, \mathbf{Y}_{k,j+1}^{(i+1)}) } \,\, \chi_{k,j+1}^{(i-1)}*\chi_{k,j+1}^{(i+1)}\end{array}
\end{eqnarray}
\end{thm}

\begin{proof}
Recall Nakajima's $t$-deformed $T$-system (see Theorem \ref{thm:Nakajima}):
\begin{eqnarray*}
t^{-\epsilon(\mathbf{Y}_{k,j}^{(i)},\mathbf{Y}_{k,j+2}^{(i)})} \,\, \chi_{k,j}^{(i)}*\chi_{k,j+2}^{(i)}  &=& t^{- \epsilon(\mathbf{Y}_{k+1,j+2}^{(i)}, \mathbf{Y}_{k-1,j}^{(i)})  } \,\, \chi_{k+1,j}^{(i)}*\chi_{k-1,j+2}^{(i)} + \\ 
&& t^{-1- \epsilon(\mathbf{Y}_{k,j+1}^{(i-1)}, \mathbf{Y}_{k,j+1}^{(i+1)}) } \,\, \chi_{k,j+1}^{(i-1)}*\chi_{k,j+1}^{(i+1)}.
\end{eqnarray*}
We will show that the $t$-deformed $T$-system is equivalent to \eqref{cond3}. It suffices to show:
\begin{eqnarray}
\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k-1, j+2}^{(i)}) + \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k+1, j}^{(i)}) &=& \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+2}^{(i)}), \label{cond3eq1}\\
 \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+1}^{(i-1)}) + \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+1}^{(i+1)})&=&-1+\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+2}^{(i)}).\label{cond3eq2}
\end{eqnarray}
By an abuse of notation, let us denote $\mathbf{Y}_{k,j}^{(i)} := u(\mathbf{Y}_{k,j}^{(i)})(s) =  e_i \otimes ( s^j + s^{j+2} + \cdots + s^{j+2(k-1)})$ (see \eqref{KRdom} and Definition \ref{uGenFun}). 
\begin{eqnarray*}
\mathbf{Y}_{k-1,j+2}^{(i)} + \mathbf{Y}_{k+1,j}^{(i)}&=& e_i \otimes \left( (s^{j+2} + \cdots + s^{j+2+2(k-2)}) + (s^j + s^{j+2} +\cdots + s^{j+2k}) \right)\\
&=&e_i \otimes \left( (s^{j+2} + \cdots + s^{j+2k-2} + s^{j+2k}) + (s^j + s^{j+2} +\cdots + s^{j+2k-2}) \right)\\&=&\mathbf{Y}_{k, j+2}^{(i)} + \mathbf{Y}_{k,j}^{(i)}\,.
\end{eqnarray*}
Then, using Definition \ref{EpsilonForm}, we compute:
\begin{eqnarray*}
\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k-1, j+2}^{(i)}) + \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k+1, j}^{(i)}) &=& (1\otimes s - 1\otimes s^{-1})D \mathbf{Y}_{k,j}^{(i)} \cdot \left( \mathbf{Y}_{k-1,j+2}^{(i)} + \mathbf{Y}_{k+1,j}^{(i)} \right)\\
&=&(1\otimes s - 1\otimes s^{-1})D \mathbf{Y}_{k,j}^{(i)} \cdot \left( \mathbf{Y}_{k, j+2}^{(i)} + \mathbf{Y}_{k,j}^{(i)} \right)\\
&=&\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+2}^{(i)}) + \underbrace{\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j}^{(i)})}_\text{0}\,,
\end{eqnarray*}
where $\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j}^{(i)})=0$ due to anti-commutativity of $\epsilon$. This proves \eqref{cond3eq1}. 

Notice that
\begin{eqnarray*}
(1\otimes s + 1\otimes s^{-1}) \mathbf{Y}_{k,j+1}^{(i)} = \mathbf{Y}_{k,j+2}^{(i)} + \mathbf{Y}_{k,j}^{(i)}\,.
\end{eqnarray*}
Then,
\begin{eqnarray*}
\mathbf{Y}_{k,j+1}^{(i-1)} + \mathbf{Y}_{k,j+1}^{(i+1)} &=& \mathbf{Y}_{k,j+1}^{(i-1)} + \mathbf{Y}_{k,j+1}^{(i+1)} - (1\otimes s+1\otimes s^{-1})\mathbf{Y}_{k,j+1}^{(i)}  + (\mathbf{Y}_{k,j+2}^{(i)} + \mathbf{Y}_{k,j}^{(i)}) \\
&=& -(1\otimes s + 1\otimes s^{-1} +A\otimes 1) \mathbf{Y}_{k,j+1}^{(i)} +\mathbf{Y}_{k,j+2}^{(i)} + \mathbf{Y}_{k,j}^{(i)} \\
&=& -K \mathbf{Y}_{k,j+1}^{(\alpha)} + \mathbf{Y}_{k,j}^{(\alpha)} + \mathbf{Y}_{k, j+2}^{(\alpha)}\,.
\end{eqnarray*}
Next, 
\begin{eqnarray*}
\epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+1}^{(i-1)}) &+& \epsilon(\mathbf{Y}_{k,j}^{(i)},\mathbf{Y}_{k,j+1}^{(i+1)}) \\ &=& (1\otimes s - 1\otimes s^{-1})D(\mathbf{Y}_{k,j}^{(i)}) \cdot \left( \mathbf{Y}_{k,j+1}^{(i-1)} + \mathbf{Y}_{k,j+1}^{(i+1)}\right)\\
&=&(1\otimes s - 1\otimes s^{-1}) D(\mathbf{Y}_{k,j}^{(i)}) \cdot \left( -K(\mathbf{Y}_{k,j+1}^{(i)}) + \mathbf{Y}_{k,j}^{(i)} + \mathbf{Y}_{k, j+2}^{(i)}\right)\\
&=&-(1\otimes s - 1\otimes s^{-1}) \mathbf{Y}_{k,j}^{(i)} \cdot \mathbf{Y}_{k,j+1}^{(i)} + \underbrace{\epsilon(\mathbf{Y}_{k,j}^{(i)},\mathbf{Y}_{k,j}^{(i)})}_\text{0}
+ \epsilon(\mathbf{Y}_{k,j}^{(i)}, \mathbf{Y}_{k,j+2}^{(i)}) \,,
\end{eqnarray*}
where
\begin{eqnarray*}
&&(1\otimes s - 1\otimes s^{-1})\mathbf{Y}_{k,j}^{(i)} \cdot \mathbf{Y}_{k,j+1}^{(i)} \\
&&\hspace{1in}= \left[ (s^{-1}-s)( s^{-j} + \cdots +s^{-j-2k+2}) ( s^{j+1} + \cdots + s^{j+2k-1}) \right]_0\\
&&\hspace{1in}= \left[  (s^{-1}-s) s^{-j-k+1} [k]_s s^{j+k} [k]_s  \right]_0\\
&&\hspace{1in}= \left[ -s\frac{1}{s-s^{-1}} (s^{2k} - 2 + s^{-2k})  \right]_0=1
\end{eqnarray*}
This concludes the proof of \eqref{cond3eq2}. 
\end{proof}


\subsection{Explicit proof of Condition II for type $A_1$}
The commutation matrix $\Lambda$ can be computed explicitly and Theorem \ref{main2} can be verified through direct computation. We will work out the explicit description of $\Lambda$ and the direct verification of Theorem \ref{main2} for the case of type $A_1$ now. 

\begin{defn}
Given a matrix $M = (m_{i,j})_{i\in I, j\in J}$ for some index sets $I$ and $J$ (possibly infinite), we can write $M$ as a generating series $M(z_1,z_2) = \sum_{i\in I, j\in J} m_{i,j}z_1^iz_2^j$, where $z_1,z_2$ are indeterminates keeping track of the indices of the matrix.   
\end{defn}

The quiver associated to the $T$-system of type $A_1$ is as follows:

\begin{figure}[ht]
\begin{center}
\begin{tikzpicture}

	\tikzstyle{black} = [circle, minimum width=5pt, fill, inner sep=0pt]
	\tikzstyle{white} = [circle, minimum width=5pt,draw,inner sep=0pt]

\node at (0,-1) [above] {$1$};
\node at (1,-1) [above] {$2$};
\node at (2,-1) [above] {$3$};
\node at (3,-1) [above] {$4$};
\node at (4,-1) [above] {$5$};

\node[black] (10) at (0,-1) {};
\node[white] (11) at (1,-1) {};
\node[black] (12) at (2,-1) {};
\node[white] (13) at (3,-1) {};
\node[black] (14) at (4,-1) {};


\node at (5,-1) {$\cdots$};



\path[-{latex'},thick]

(10) edge (11)
(12) edge (11)
(12) edge (13)
(14) edge (13)
(6,-1) edge node[above]{$k$} (7,-1)
;

\end{tikzpicture}
\end{center}
\caption{The quiver $\Gamma_T$, ($k\in \mathbb{Z}_+$)}
\label{fig:TsystemA1}
\end{figure}

The signed adjacency matrix of the quiver $\Gamma_T$ in Figure~\ref{fig:TsystemA1} is an infinite matrix indexed by $\mathbb{Z}_+\times \mathbb{Z}_+$ given by:

\begin{equation*}
B=\left(
\begin{array}{cccccccccc}
0&1&0&0&0&0&0&0&0&\cdots\\
-1&0&-1&0&0&0&0&0&0&\cdots\\
0&1&0&1&0&0&0&0&0&\cdots\\
0&0&-1&0&-1&0&0&0&0&\cdots\\
0&0&0&1&0&1&0&0&0&\cdots\\
0&0&0&0&-1&0&-1&0&0&\cdots\\
0&0&0&0&0&1&0&1&0&\cdots\\
0&0&0&0&0&0&-1&0&-1&\cdots\\
0&0&0&0&0&0&0&1&0&\cdots\\
\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\ddots\\
\end{array}\right)
\end{equation*}

which can be written as:
\begin{eqnarray*}
B(z_1,z_2)&=&z_1z_2\frac{z_2-z_1}{1+z_1z_2}.
\end{eqnarray*}

Recall that to each vertex $k$, we associate the KR-module $W_{k, l = (k)_2-k}$ (see Equation \eqref{eq:FundClustT}), and the commutation matrix $\Lambda$ is given by:
\begin{eqnarray*}
\Lambda(k,k') = 2 \epsilon(\mathbf{Y}_{k, (k)_2-k} , \mathbf{Y}_{k', (k')_2-k'}), 
\end{eqnarray*}
where $\mathbf{Y}_{k,j} = Y_{j} Y_{j+2}\cdots Y_{j+2k-2}$ (see Equation \eqref{KRdom}), where we dropped the index $i$ in $Y_{i,j}$ since $i$ can take exactly one value, and $\epsilon$ is from Definition \ref{def:Nakajima}. More precisely, we have: 
\begin{eqnarray*}
\epsilon(\mathbf{Y}_{k,(k)_2-k}, \mathbf{Y}_{k', (k')_2-k'}) &=& - \sum_{j=0}^{k-1} \tilde{u}_{(k)_2 - k + 2j -1}( \mathbf{Y}_{k', (k')_2-k'} ) + \sum_{j = 0}^{k'-1} \tilde{u}_{(k')_2 - k' + 2j - 1}(\mathbf{Y}_{k,(k)_2-k}),
\end{eqnarray*}
where $\tilde{u}_{j}(m) \in \mathbb{R}$ ($j\in \mathbb{Z}$) is the unique solution of the system:
\begin{equation*} 
u_{j}(m) = \tilde{u}_{j-1}(m) + \tilde{u}_{j+1}(m)\,\, ,
\end{equation*}
such that $\tilde{u}_{j}(m)=0$ for $j$ sufficiently small (see Definition \ref{uinverse}). Again, we dropped the dependance on $i$ and set $u_{i,j}(m) = u_j(m)$.

\begin{example}
Consider $m=\mathbf{Y}_{1,0} = Y_{0}$. Then $\tilde{u}_{1}(m) = 1$ and $\tilde{u}_{j}(m) = 0$ for all $j\geq 1$. 
\begin{eqnarray*}
&\vdots&\\
u_4(m) = 0 &=& \tilde{u}_{5}(m) + \tilde{u}_{7}(m) = 0 + 0\\
u_2(m) = 0 &=& \tilde{u}_{1}(m) + \tilde{u}_{3}(m) = 1 + 0\\
u_0(m) = 1 &=& \tilde{u}_{-1}(m) + \tilde{u}_{1}(m) = 0 + 1\\
u_{-2}(m) = 0 &=& \tilde{u}_{-3}(m) + \tilde{u}_{-1}(m) =  0 + 0\\
&\vdots&
\end{eqnarray*}
Here, we must have $\tilde{u}_{-1}(m)=0$. Otherwise, $\tilde{u}_{-2j-1}(u)=0$ for all $j\geq 0$, which contradicts the condition $\tilde{u}_j(m)=0$ for $j$ sufficiently small. 
\end{example}


\begin{example}
When $m=\mathbf{Y}_{2,-2} = Y_{-2}Y_0$, we can compute $\tilde{u}_{-1}(m) = 1$ and $\tilde{u}_j(m) = 0$ for all $j\neq -1$. Then,
\begin{eqnarray*}
\epsilon(\mathbf{Y}_{1,0}, \mathbf{Y}_{2,-2}) &=&  - \tilde{u}_{-1}( Y_{-2}Y_0 ) + (\tilde{u}_{-3}(Y_0) + \tilde{u}_{-1}(Y_0) ) \\&=& - 1 + 0 = -1.\\
\end{eqnarray*}
\end{example}

\begin{example}
When $m=\mathbf{Y}_{4,-4} = Y_{-4}Y_{-2}Y_0Y_2$, we can compute $\tilde{u}_{-3}(m) = \tilde{u}_{1}(m) = 1$ and $\tilde{u}_j(m) = 0$ for all $j\neq -3,1$. Then,
\begin{eqnarray*}
\epsilon(\mathbf{Y}_{1,0}, \mathbf{Y}_{4,-4}) &=&  - \tilde{u}_{-1}( Y_{-4}Y_{-2}Y_0Y_2 ) + (\tilde{u}_{-5}(Y_0) + \tilde{u}_{-3}(Y_0) + \tilde{u}_{-1}(Y_0) + \tilde{u}_{1}(Y_0) ) \\
&=& - 0 + 1 = 1.\\
\end{eqnarray*}
\end{example}
Similarly, the commutation matrix $\Lambda$ can be computed and is given by:
\begin{equation*}
\Lambda=\left(
\begin{array}{cccccccccc}
0&-1&0&1&0&-1&0&1&0&\cdots\\
1&0&0&0&0&0&0&0&0&\cdots\\
0&0&0&-1&0&1&0&-1&0&\cdots\\
-1&0&1&0&0&0&0&0&0&\cdots\\
0&0&0&0&0&-1&0&1&0&\cdots\\
1&0&-1&0&1&0&0&0&0&\cdots\\
0&0&0&0&0&0&0&-1&0&\cdots\\
-1&0&1&0&-1&0&1&0&0&\cdots\\
0&0&0&0&0&0&0&0&0&\cdots\\
\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\vdots&\ddots\\
\end{array}\right)
\end{equation*}
which can be written as:
\begin{eqnarray*}
\Lambda(z_1,z_2)&=&\frac{z_1z_2(z_1-z_2)}{(1+z_1z_2)(1+z_1^2)(1+z_2^2)}.
\end{eqnarray*}
Then,
\begin{eqnarray*}
\frac{1}{2}\Lambda B (z_1,z_2) &=& {\rm{Res}}_w \left( \frac{1}{w} \Lambda(z_1,w) B(\frac{1}{w}, z_2)  \right) \\
&=&{\rm{Res}}_w \left( \frac{1}{w} \frac{z_1w(z_1-w)}{(1+wz_1)(1+z_1^2)(1+w^2)}\cdot \frac{1}{w}z_2 \frac{z_2-\frac{1}{w}}{1+\frac{z_2}{w}} \right)\\
&=& {\rm{Res}}_w \left(  \frac{1}{w} \frac{z_1z_2(z_1-w)(wz_2-1)}{(1+wz_1)(1+z_1^2)(1+w^2)(w+z_2)} \right)\\
&=&-\frac{z_1^2z_2}{(1+z_1^2)z_2}+\frac{z_1z_2(z_1+z_2)(-z_2^2-1)}{(-z_2)(1-z_1z_2)(1+z_1^2)(1+z_2^2)}\\
&=&-\frac{z_1^2}{1+z_1^2}+\frac{z_1(z_1+z_2)}{(1-z_1z_2)(1+z_1^2)}\\
&=&\frac{-z_1^2+z_1^3z_2+z_1^2+z_1z_2}{(1-z_1z_2)(1+z_1^2)}=\frac{z_1z_2}{1-z_1z_2} = I(z_1,z_2)
\end{eqnarray*}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Evolution in $k$-direction}
We showed that Nakajima's deformed $T$-system forms a quantum cluster algebra with evolution in $l$-direction in $T_{k,l}^{(i)}$ variables (equivalently in $j$-direction in $\chi_{k,j}^{(i)}$ variables). We now show that the same deformed $T$-system is not a quantum cluster algebra with evolution in $k$-direction. In particular, this shows that the quantum $T$-system is not compatible with the quantum $Q$-system considered in \cite{di2011non}. 

By re-writing Nakajima's $t$-deformed $T$-system of Theorem \ref{thm:Nakajima} so that the evolution is in $k$-direction, we obtain:
\begin{eqnarray*}
\chi_{k+1,j}^{(i)} *_\gamma \chi_{k-1,j+2}^{(i)} = \chi_{k,j}^{(i)} *_\gamma \chi_{k,j+2}^{(i)} - t^{-1} \chi_{k,j+1}^{(i-1)} *_\gamma \chi_{k,j+1}^{(i+1)} \,.
\end{eqnarray*}
Notice that the negative sign on the right-hand side is not compatible with cluster algebra interpretation, where the right-hand side expression must have exactly $2$ positive contributions. However, it is possible to renormalize $\chi_{k,j}^{(i)}$'s such that the negative sign becomes positive \cite{di2009positivity}. We call the resulting variables $\widehat{\chi}_{k,j}^{(i)}$. The ensuing $T$-system of type $A_1$ is as follows:
\begin{eqnarray} \label{A1T}
\widehat{\chi}_{k+1,j}*_\gamma \widehat{\chi}_{k-1,j+2} =  \widehat{\chi}_{k,j}*_\gamma \widehat{\chi}_{k,j+2} + t^{-1} ,
\end{eqnarray}
where we dropped the parameter $i$ since it can have only one value. 

The variables on the right-hand side of \eqref{A1T} must belong to the same cluster, and therefore, must $t$-commute if \eqref{A1T} does form a quantum mutation. We will give a simple counter example to this condition, which shows that \eqref{A1T} is not a quantum cluster algebra.  

\begin{example}
Let $\g = \mathfrak{sl}_2$. We consider $\widehat{\chi}_{1,0}$ and $\widehat{\chi}_{1,2}$, which are both on the right-hand side of \eqref{A1T}. The values of these variables are given as follows:

\begin{center}
\begin{tikzpicture}

\node at (0,0) {
\begin{tikzpicture}
\node at (-3,2) {$\widehat{\chi}_{1,0} = \chi_{q,t}(W_{1,0}^{(1)})$};
\node at (-1,2) {$=$};
\node (1') at (0,2) {$Y_{1,0}$};
\node (1) at (0,.5) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{0}}]&1
\end{ytableau}
};

\node (2') at (2,2) { $Y_{1,2}^{-1}$};

\node (2) at (2,0.5) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{0}}]&2
\end{ytableau}
};

\node at (1,2) {$+$};
\path[->,thick]
(1') edge[dotted] (1)
(2') edge[dotted] (2);
\end{tikzpicture}
};

\node at (4,.5) {;};

\node at (8,0) {
\begin{tikzpicture}
\node at (-3,2) {$\widehat{\chi}_{1,2} = \chi_{q,t}(W_{1,2}^{(1)})$};
\node at (-1,2) {$=$};
\node (1') at (0,2) {$Y_{1,2}$};
\node (1) at (0,.5) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&1
\end{ytableau}
};

\node (2') at (2,2) { $Y_{1,4}^{-1}$};

\node (2) at (2,0.5) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&2
\end{ytableau}
};

\node at (1,2) {$+$};
\path[->,thick]
(1') edge[dotted] (1)
(2') edge[dotted] (2);
\end{tikzpicture}
};

\end{tikzpicture}
\end{center}
Their twisted product can be computed using Theorem \ref{gamma}, and is given by:

\begin{center}
\begin{tikzpicture}

\node at (-3,2) {$\widehat{\chi}_{1,0} *_\gamma \widehat{\chi}_{1,2}$};
\node at (-1.5,2) {$=$};

\node (1') at (0,2) {$Y_{1,0}Y_{1,2}$};

\node (1) at (0,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&\none&1\\
\none[{\color{blue}\scriptstyle{0}}]&1\\
\end{ytableau}
};

\node (2') at (3,2) { $Y_{1,0}Y_{1,4}^{-1}$};

\node (2) at (3,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&\none&2\\
\none[{\color{blue}\scriptstyle{0}}]&1\\
\end{ytableau}
};


\node (3') at (6,2) { $t^{-1} \,Y_{1,2}^{-1}Y_{1,2}$};

\node (3) at (6,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&\none&1\\
\none[{\color{blue}\scriptstyle{0}}]&2\\
\end{ytableau}
};


\node (4') at (9,2) { $Y_{1,2}^{-1}Y_{1,4}^{-1}$};

\node (4) at (9,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&\none&2\\
\none[{\color{blue}\scriptstyle{0}}]&2\\
\end{ytableau}
};

\node at (1.5,2) {$+$};
\node at (4.5,2) {$+$};
\node at (7.5,2) {$+$};

\path[->,thick]
(1') edge[dotted] (1)
(2') edge[dotted] (2)
(3') edge[dotted] (3)
(4') edge[dotted] (4)
;

\end{tikzpicture}
\end{center}
Notice that the third term on the right-hand side forms a pair of column tableaux of regular type with non-zero boundary term in $\gamma$. On the other hand, 
\begin{center}
\begin{tikzpicture}

\node at (-3,2) {$\widehat{\chi}_{1,2} *_\gamma \widehat{\chi}_{1,0}$};
\node at (-1.5,2) {$=$};

\node (1') at (0,2) {$Y_{1,2}Y_{1,0}$};

\node (1) at (0,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&1\\
\none[{\color{blue}\scriptstyle{0}}]&\none&1\\
\end{ytableau}
};

\node (2') at (3,2) { $Y_{1,4}^{-1}Y_{1,0}$};

\node (2) at (3,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&2\\
\none[{\color{blue}\scriptstyle{0}}]&\none&1\\
\end{ytableau}
};


\node (3') at (6,2) {$t \,Y_{1,2}Y_{1,2}^{-1}$};

\node (3) at (6,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&1\\
\none[{\color{blue}\scriptstyle{0}}]&\none&2\\
\end{ytableau}
};


\node (4') at (9,2) {$Y_{1,4}^{-1}Y_{1,2}^{-1}$};

\node (4) at (9,0) {
\begin{ytableau}
\none[{\color{blue}\scriptstyle{-1}}]&2\\
\none[{\color{blue}\scriptstyle{0}}]&\none&2\\
\end{ytableau}
};

\node at (1.5,2) {$+$};
\node at (4.5,2) {$+$};
\node at (7.5,2) {$+$};

\path[->,thick]
(1') edge[dotted] (1)
(2') edge[dotted] (2)
(3') edge[dotted] (3)
(4') edge[dotted] (4)
;

\end{tikzpicture}
\end{center}

We see that $\widehat{\chi}_{1,0}$ and $\widehat{\chi}_{1,2}$ do not $t$-commute. 
\end{example}

\section{Conclusion/Discussion}
The Nakajima $(q,t)$-characters of KR-modules satisfy a deformed $T$-system, introduced in \cite{nakajima2003t}, which is a $t$-deformed discrete dynamical system with $3$ independent parameters: $i,k,j$. In this paper, we showed that this $t$-deformed $T$-system forms a quantum mutation in a quantization of the $T$-system cluster algebra only when the direction of mutation is in the $l$-parameter. 

This result pertains to type $A$ only. It is an open question whether the same holds for other types, and in particular to type $D$. There are noticeable differences between types $A$ and $D$. For example, unlike the case in type $A$, the KR-modules of type $D$ are reducible as $U_q(\g)$-modules. Also, the $(q,t)$-characters of the KR-modules are not identical to their $q$-characters. Despite these differences, the proofs of Conditions II and III are applicable to type $D$ with minimal adjustments. The hard part is the proof of Condition I, the $t$-commutativity of the fundamental cluster variables. The proof of Condition I is entirely combinatorial and requires the knowledge of the $(q,t)$-characters of all the KR-modules in the fundamental cluster. Nakajima's tableaux-sum notation exists for type $D$ as well (see \cite{nakajima2002t}). However, it is combinatorially different from the type $A$ case, and, therefore, all the combinatorial structures introduced as part of the proof of Condition I will need to be translated to the combinatorics of the type $D$. 


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\bibliographystyle{plain} 




\end{document}